SECURITIES AND EXCHANGE COMMISSION

FORM 10-Q

Quarterly report pursuant to sections 13 or 15(d)

Filing Date: 2022-05-04 | Period of Report: 2022-03-31 SEC Accession No. 0001564590-22-017813

(HTML Version on secdatabase.com)

FILER

GENTHERM Inc

CIK:903129| IRS No.: 954318554 | State of Incorp.:MI | Fiscal Year End: 1231

Type: 10-Q | Act: 34 | File No.: 000-21810 | Film No.: 22891239

SIC: 3714 Motor vehicle parts & accessories

Mailing Address 21680 HAGGERTY ROAD

Business Address 21680 HAGGERTY ROAD NORTHVILLE MI 48167-8994 NORTHVILLE MI 48167-8994 248-504-0500

UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

FORM 10-Q

(Mark One) ☑ QUARTERLY REPORT PURSUANT TO SEC For the q	CTION 13 OR 15(d) OF T uarterly period ended M OR	
T TRANSITION REPORT BURGUANT TO CO.		THE SECUDITIES EVOLVANCE ACT OF 1024
☐ TRANSITION REPORT PURSUANT TO SEC		
	ansition period from	
Col	mmission File Number: (J-21810
GENTHER	RM INCO	RPORATED
(Exact name	e of registrant as specifie	ed in its charter)
Michigan (State or other jurisdiction of incorporation or organization)		95-4318554 (I.R.S. Employer Identification No.)
21680 Haggerty Road, Northville, Mi (Address of principal executive offices)	Ī	48167 (Zip Code)
Registrant's telepho	one number, including ar	rea code: (248) 504-0500
Securities registered pursuant to Section 12(b) of the	Act:	
Title of each class Common Stock, no par value	Trading Symbol THRM	Name of each exchange on which registered Nasdaq
Indicate by check mark whether the registrant (1) has Exchange Act of 1934 during the preceding 12 month and (2) has been subject to such filing requirements for	ns (or for such shorter peri	od that the registrant was required to file such reports),
		ry Interactive Data File required to be submitted pursuant 12 months (or for such shorter period that the registrant
Indicate by check mark whether the registrant is a lar company, or an emerging growth company. See the d company," and "emerging growth company" in Rule	efinitions of "large acceler	
Large accelerated filer Non-accelerated filer Emerging growth company □		Accelerated filer □ Smaller reporting company □
If an emerging growth company, indicate by check m complying with any new or revised financial account		
Indicate by check mark whether the registrant is a she	ell company (as defined in	Rule 12b-2 of the Exchange Act). Yes \square No \boxtimes

At April 29, 2022, there were 33,131,660 issued and outstanding shares of Common Stock of the registrant.

Copyright © 2022 www.secdatabase.com . All Rights Reserved. Please Consider the Environment Before Printing This Document

TABLE OF CONTENTS

Part I. Financi	al Information	3
Item 1.	Financial Statements (Unaudited)	3
	Consolidated Condensed Balance Sheets	3
	Consolidated Condensed Statements of Income	4
	Consolidated Condensed Statements of Comprehensive Income	5
	Consolidated Condensed Statements of Cash Flows	6
	Consolidated Condensed Statements of Changes in Shareholders' Equity	7
	Notes to Unaudited Consolidated Condensed Financial Statements	8
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	20
Item 3.	Quantitative and Qualitative Disclosures About Market Risk	30
Item 4.	Controls and Procedures	31
Part II. Other	<u>Information</u>	32
Item 1.	Legal Proceedings	32
Item 1A.	Risk Factors	32
Item 2.	<u>Unregistered Sales of Equity Securities and Use of Proceeds</u>	34
Item 5.	Other Information	34
Item 6.	<u>Exhibits</u>	36
Signatures		37
	2	

PART I. FINANCIAL INFORMATION

ITEM 1. FINANCIAL STATEMENTS

GENTHERM INCORPORATED

CONSOLIDATED CONDENSED BALANCE SHEETS

(In thousands, except share data) (Unaudited)

	March 31, 2022		December 31, 2021		
ASSETS					
Current Assets:					
Cash and cash equivalents	\$	177,917	\$	190,606	
Accounts receivable, net		207,351		182,987	
Inventory:					
Raw materials		111,621		96,426	
Work in process		10,991		9,495	
Finished goods		53,556		53,556	
Inventory, net		176,168		159,477	
Other current assets		41,034		32,775	
Total current assets		602,470		565,845	
Property and equipment, net		152,288		155,270	
Goodwill		64,979		66,033	
Other intangible assets, net		34,982		37,554	
Operating lease right-of-use assets		27,445		24,387	
Deferred income tax assets		69,420		69,630	
Other non-current assets		15,529		16,624	
Total assets	\$	967,113	\$	935,343	
LIABILITIES AND SHAREHOLDERS' EQUITY	<u> </u>	707,112	<u> </u>	750,515	
Current Liabilities:					
Accounts payable	\$	156,236	\$	122,727	
Current lease liabilities	Ψ	5,920	Ψ	5,669	
Current maturities of long-term debt		2,500		2,500	
Other current liabilities		79,044		82,193	
Total current liabilities		243,700		213,089	
Long-term debt, less current maturities		36,250		36,250	
Non-current lease liabilities		19,978		19,789	
Pension benefit obligation		6,362		6,832	
Other non-current liabilities		5,036		5,577	
	¢		¢.		
Total liabilities	\$	311,326	\$	281,537	
Shareholders' equity:					
Common Stock:					
No par value; 55,000,000 shares authorized 33,127,531 and 33,008,185 issued and		117 922		110 646	
outstanding at March 31, 2022 and December 31, 2021, respectively		117,832		118,646	
Paid-in capital		5,720		5,866	
Accumulated other comprehensive loss		(45,728)		(36,922	
Accumulated earnings		577,963		566,216	
Total shareholders' equity		655,787	<u></u>	653,806	
Total liabilities and shareholders' equity	\$	967,113	\$	935,343	

CONSOLIDATED CONDENSED STATEMENTS OF INCOME

(In thousands, except per share data) (Unaudited)

Three Months Ended March 31,				
	2022	2021		
\$	267,657	\$	288,535	
	203,544		200,866	
	64,113		87,669	
	20,434		17,603	
	29,308		28,526	
	181		791	
	49,923		46,920	
	14,190		40,749	
	(569)		(1,039)	
	2,217		773	
	204		(9)	
	16,042		40,474	
	4,295		7,565	
\$	11,747	\$	32,909	
\$	0.36	\$	1.00	
\$	0.35	\$	0.99	
	33,035		32,946	
	33,377		33,390	
		2022 \$ 267,657 203,544 64,113 20,434 29,308 181 49,923 14,190 (569) 2,217 204 16,042 4,295 \$ 11,747 \$ 0.36 \$ 0.35 33,035	2022 \$ 267,657 \$ 203,544 64,113 20,434 29,308 181 49,923 14,190 (569) 2,217 204 16,042 4,295 \$ 11,747 \$ \$ 0.36 \$ \$ \$ 0.35 \$ \$ 33,035	

CONSOLIDATED CONDENSED STATEMENTS OF COMPREHENSIVE INCOME (In thousands) (Unaudited)

	T	Three Months Ended March 31,			
		2022		2021	
Net income	\$	11,747	\$	32,909	
Other comprehensive loss:					
Pension benefit obligations		29		_	
Foreign currency translation adjustments		(9,294)		(13,152)	
Unrealized gain (loss) on foreign currency derivative securities, net of tax		464		(522)	
Unrealized loss on commodity derivative securities, net of tax		(5)		_	
Other comprehensive loss, net of tax		(8,806)		(13,674)	
Comprehensive income	\$	2,941	\$	19,235	

CONSOLIDATED CONDENSED STATEMENTS OF CASH FLOWS (In thousands) (Unaudited)

	Three Months Ended March 31,			March 31,
		2022		2021
Operating Activities:				
Net income	\$	11,747	\$	32,909
Adjustments to reconcile net income to net cash provided by operating activities:				
Depreciation and amortization		9,577		9,854
Deferred income taxes		(778)		105
Non-cash stock based compensation		2,279		4,460
Loss on disposition of property and equipment		107		242
Other		256		(103)
Changes in assets and liabilities:				
Accounts receivable, net		(25,788)		(13,931)
Inventory		(18,116)		(11,546)
Other assets		(10,716)		3,243
Accounts payable		34,097		18,113
Other liabilities		(3,349)		(3,679)
Net cash (used in) provided by operating activities		(684)		39,667
Investing Activities:				
Purchases of property and equipment		(5,659)		(9,913)
Proceeds from the sale of property and equipment		52		10
Cost of technology investments		(350)		(200)
Net cash used in investing activities		(5,957)		(10,103)
Financing Activities:				
Repayments of debt		_		(130,000)
Proceeds from the exercise of Common Stock options		569		5,984
Taxes withheld and paid on employees' share-based payment awards		(4,319)		(1,532)
Acquisition contingent consideration payment		_		(68)
Net cash used in financing activities		(3,750)		(125,616)
Foreign currency effect		(2,298)		(1,338)
Net decrease in cash and cash equivalents		(12,689)		(97,390)
Cash and cash equivalents at beginning of period		190,606		268,345
Cash and cash equivalents at end of period	\$	177,917	\$	170,955
Supplemental disclosure of cash flow information:		-		-
Cash paid for taxes	\$	3,267	\$	2,555
Cash paid for interest	\$	421	\$	844
Cash para for interest	Ψ	741	Ψ	077

CONSOLIDATED CONDENSED STATEMENTS OF CHANGES IN SHAREHOLDERS' EQUITY (In thousands) (Unaudited)

					Ac	cumulated Other			
	Commo	on Stock	P	aid-in	Con	nprehensive	Ac	cumulated	
	Shares	Amount	C	Capital		Loss]	Earnings	Total
Balance at December 31, 2021	33,008	\$ 118,646	\$	5,866	\$	(36,922)	\$	566,216	\$ 653,806
Net income	_	_		_				11,747	11,747
Other comprehensive loss	_	_		_		(8,806)		_	(8,806)
Stock compensation, net	119	(814)		(146)				_	(960)
Balance at March 31, 2022	33,127	\$ 117,832	\$	5,720	\$	(45,728)	\$	577,963	\$ 655,787
					Ac	cumulated Other			

					_	Other			
	Commo	on Stock	I	Paid-in	C	omprehensive	Ac	cumulated	
	Shares	Amount	(Capital		Loss		Earnings	Total
Balance at December 31, 2020	32,921	\$ 121,073	\$	7,458	\$	(14,982)	\$	472,782	\$ 586,331
Net income	_			_		_		32,909	32,909
Other comprehensive loss	_	_		_		(13,674)		_	(13,674)
Stock compensation, net	190	8,527		(1,335)		_		_	7,192
Balance at March 31, 2021	33,111	\$ 129,600	\$	6,123	\$	(28,656)	\$	505,691	\$ 612,758

Note 1 – Overview

Gentherm Incorporated, a Michigan corporation, and its consolidated subsidiaries ("Gentherm", "we", "us", "our" or the "Company") is a global developer, manufacturer and marketer of innovative thermal management technologies for a broad range of heating, cooling and temperature control applications, primarily in the automotive and medical industries. Within the automotive industry, our products provide solutions for passenger climate comfort and convenience, battery thermal management and cell connecting systems. Within the medical industry our products provide patient temperature management solutions. Our automotive products can be found on vehicles manufactured by nearly all the major automotive original equipment manufacturers ("OEMs") operating in North America and Europe, and several major OEMs in Asia. We operate in locations aligned with our major customers' product strategies to provide locally enhanced design, integration and production capabilities. The Company is also developing a number of new technologies and products that are expected to enable improvements to existing products and to create new product applications for existing and new markets.

Basis of Presentation and Significant Accounting Policies

The unaudited consolidated condensed financial statements included herein have been prepared in accordance with accounting principles generally accepted in the United States of America ("U.S. GAAP") for interim financial information and pursuant to the rules and regulations of the U.S. Securities and Exchange Commission. Certain information and footnote disclosures normally included in financial statements prepared in accordance with U.S. GAAP have been condensed or omitted pursuant to those rules and regulations. The information furnished in the consolidated condensed financial statements include all adjustments (consisting of only normal, recurring adjustments), considered necessary to present fairly the results of operations, financial position and cash flows of the Company. These financial statements should be read in conjunction with the Company's Annual Report on Form 10-K for the year ended December 31, 2021. The operating results for interim periods are not necessarily indicative of results that may be expected for any other interim period or for the full year.

In preparing these financial statements, management was required to make estimates and assumptions that affect the reported amounts of assets, liabilities, revenues and expenses. These estimates and assumptions are based on our historical experience, the terms of existing contracts, our evaluation of trends in the industry, information provided by our customers and suppliers and information available from other outside sources, as appropriate. These estimates and assumptions are subject to an inherent degree of uncertainty. We are not presently aware of any events or circumstances that would require us to update such estimates and assumptions or revise the carrying value of our assets or liabilities. Our estimates may change, however, as new events occur and additional information is obtained. As a result, actual results may differ significantly from our estimates, and any such differences may be material to our financial statements.

Principles of Consolidation

The consolidated condensed financial statements include the accounts of the Company, its wholly owned subsidiaries and those entities in which it has a controlling financial interest. The Company evaluates its relationship with other entities for consolidation and to identify whether such entities are variable interest entities ("VIE") and to assess whether the Company is the primary beneficiary of such entities. Investments in affiliates in which Gentherm does not have control but does have the ability to exercise significant influence over operating and financial policies are accounted for under the equity method. When Gentherm does not have the ability to exercise significant influence (generally when ownership interest is less than 20%), investments in affiliates are measured at cost, less impairments, adjusted for observable price changes in orderly transactions for identical or similar investments of the same issuer.

Variable Interest Entities

The Company maintains an ownership interest in a VIE, Carrar Ltd. ("Carrar"). Carrar is a technology developer of advanced thermal management systems for the electric mobility market. The Company determined that Carrar is a VIE; however, the Company does not have a controlling financial interest or have the power to direct the activities that most significantly affect the economic performance of the investment. Therefore, the Company has concluded that it is not the primary beneficiary. Gentherm's investment in Carrar is measured at cost, less impairments, adjusted for observable price changes in orderly transactions for identical or similar investments of the same issuer. The Carrar investment was \$5,200 as of March 31, 2022 and December 31, 2021, and is recorded in Other non-current assets in the consolidated condensed balance sheet.

Revenue Recognition

The Company has no material contract assets or contract liabilities as of March 31, 2022.

The Company recognizes an asset for the incremental costs of obtaining a contract with a customer if the benefits of those costs are expected to be realized for a period greater than one year. Total capitalized costs to obtain a contract were \$1,884 and \$1,946 as of March 31, 2022 and December 31, 2021, respectively. These amounts are recorded in Other non-current assets and are being amortized into Product revenues over the expected production life of the applicable program.

Segment Reporting

The Company has two reportable segments: Automotive and Medical.

The Automotive reporting segment is comprised of the results from our global automotive businesses, including the design, development, manufacturing and sales of automotive climate comfort systems, automotive cable systems, battery performance solutions, and automotive electronic and software systems.

The Medical reporting segment is comprised of the results from the patient temperature management business in the medical industry. Patient temperature management includes temperature management systems across multiple product categories addressing the needs of hyper-hypothermia therapy in intensive care, normothermia in surgical procedures and additional warming/cooling therapies utilized in acute care, ambulatory, clinics and home health.

Note 2 – New Accounting Pronouncements

Reference Rate Reform

In March 2020, the FASB issued ASU 2020-04, "Reference Rate Reform (Topic 848): Facilitation of Effects of Reference Rate Reform on Financial Reporting". ASU 2020-04 provides practical expedients and exceptions for applying GAAP to contracts, hedging relationships, and other transactions affected by reference rate reform if certain criteria are met. The expedients and exceptions provided by the amendments in this update apply only to contracts, hedging relationships, and other transactions that reference the London Interbank Offered Rate ("LIBOR") or another reference rate expected to be discontinued as a result of reference rate reform. These amendments are not applicable to contract modifications made and hedging relationships entered into or evaluated after December 31, 2022. In January 2021, the FASB subsequently issued ASU 2021-01, "Reference Rate Reform (Topic 848): Scope" to clarify that certain optional expedients and exceptions in Topic 848 for contract modifications and hedge accounting apply to derivatives that are affected by the discounting transition. ASU 2020-04 and ASU 2021-01 are effective as of March 12, 2020 through December 31, 2022 and may be applied retrospectively to contract modifications and hedging relationships from the beginning of an interim period that includes or is subsequent to March 12, 2020, or on a prospective basis to new modifications from any date within an interim period that includes or is subsequent to the date of the issuance of a final update, up to the date that financial statements are available to be issued. The adoption of this standard is not expected to have a material impact on the Company's financial statements.

Note 3 – Restructuring

Manufacturing Footprint Rationalization

In September 2019, the Company committed to a restructuring plan ("Plan") to improve the Company's manufacturing productivity and rationalize its footprint. Under this Plan, the Company is relocating and consolidating certain automotive electronics manufacturing plants in North America and China. During 2021, the Company completed the closures and relocation of its automotive electronics manufacturing operations from Burlington, Canada to Celaya, Mexico and from Longgang, Shenzhen, China to Bantian, Shenzhen, China. As of March 31, 2022, the electronics manufacturing in Acuña, Mexico continues to transition to Celaya, Mexico.

During the three months ended March 31, 2022, the Company recognized restructuring expense of \$50 for employee separation costs and \$101 for other costs. During the three months ended March 31, 2021, the Company recognized restructuring expense of \$206 for employee separation costs and \$259 for other costs.

The Company has recorded approximately \$10,256 of restructuring expenses since the inception of this program and as of March 31, 2022, \$538 remains accrued. Actions under the Plan are expected to be substantially completed by the end of 2022 and future expenses are expected to be less than \$1,000.

Other Restructuring Activities

As part of the Company's continued efforts to optimize its cost structure, the Company has undertaken several discrete restructuring actions. During the three months ended March 31, 2022, the Company recognized \$30 for other costs. During the three months ended March 31, 2021, the Company recognized \$326 for employee separation costs. These restructuring expenses were primarily associated with restructuring actions focused on the rotation of our manufacturing footprint to best cost locations and the reduction of global overhead costs.

Restructuring Expenses By Reporting Segment

The following table summarizes restructuring expense for the three months ended March 31, 2022 and 2021 by reporting segment:

	 Three Months Ended March 31,					
	2022		2021			
Automotive	\$ 181	\$	791			
Medical	_		_			
Corporate	_		_			
Total	\$ 181	\$	791			

Restructuring Liability

Restructuring liabilities are classified as Other current liabilities in the consolidated condensed balance sheets. The following table summarizes restructuring liability for the three months ended March 31, 2022:

	ployee tion Costs	Other Related Costs	Total
Balance at December 31, 2021	\$ 1,494	\$ —	\$ 1,494
Additions, charged to restructuring expenses		131	131
Cash payments	(544)	(131)	(675)
Non-cash utilization	_	_	_
Change in estimate	50	_	50
Currency translation	 (18)		(18)
Balance at March 31, 2022	\$ 982	<u> </u>	\$ 982

Note 4 – Details of Certain Balance Sheet Components

	Mar	rch 31, 2022	December 31, 2021		
Other current assets:					
Notes receivable	\$	15,958	\$	13,033	
Income tax and other tax receivable		12,501		10,681	
Billable tooling		4,993		3,778	
Prepaid expenses		4,751		3,407	
Other		2,831		1,876	
Total other current assets	\$	41,034	\$	32,775	
Other current liabilities:					
Liabilities from discounts and rebates	\$	26,632	\$	27,343	
Accrued employee liabilities		21,244		28,818	
Income tax and other taxes payable		19,529		17,068	
Restructuring		982		1,494	
Accrued warranty		1,772		1,916	
Other		8,885		5,554	
Total other current liabilities	\$	79,044	\$	82,193	

Note 5 – Goodwill and Other Intangibles

Goodwill

Changes in the carrying amount of goodwill, by reportable segment, for the three months ended March 31, 2022 was as follows:

	Au	tomotive	Medical		Total
'Balance as of December 31, 2021	\$	37,329	\$ 28,704	\$	66,033
Exchange rate impact		(762)	(292)		(1,054)
Balance as of March 31, 2022	\$	36,567	\$ 28,412	\$	64,979

Other Intangible Assets

Other intangible assets and accumulated amortization balances as of March 31, 2022 and December 31, 2021 were as follows:

	March 31, 2022						December 31, 2021					
	Gross Carrying Value		Accumulated Amortization		Net Carrying Value		Gross Carrying Value		Accumulated Amortization			Net Carrying Value
Definite-lived:						_						
Customer relationships	\$	87,854	\$	(63,445)	\$	24,409	\$	90,448	\$	(64,105)	\$	26,343
Technology		28,706		(24,288)		4,418		29,464		(24,487)		4,977
Product development costs		19,661		(19,183)		478		20,329		(19,772)		557
Software development		1,007		_		1,007		1,007		_		1,007
Indefinite-lived:												
Tradenames		4,670		_		4,670		4,670		_		4,670
Total	\$	141,898	\$	(106,916)	\$	34,982	\$	145,918	\$	(108,364)	\$	37,554

In addition to annual impairment testing, which is performed in the fourth quarter of each fiscal year, the Company continuously monitors for events and circumstances that could negatively impact the key assumptions used in determining fair value and therefore require interim impairment testing, including long-term revenue growth projections, profitability, discount rates, recent market valuations from transactions by comparable companies, volatility in the Company's market capitalization, and general industry, market and macroeconomic conditions. We are not presently aware of any events or circumstances that would require us to revise the carrying value of our assets or liabilities as of March 31, 2022.

Note 6 - Debt

The following table summarizes the Company's debt as of March 31, 2022 and December 31, 2021:

	March 31,	2022	December 31, 2021				
	Interest Rate	Principal Balance	Interest Rate	Principal Balance			
Amended Credit Agreement:							
U.S. Revolving Note (U.S. Dollar denominations)	1.71%	\$ 35,000	1.35%	\$ 35,000			
U.S. Revolving Note (Euro denominations)	_	_	_	_			
DEG Vietnam Loan	5.21%	3,750	5.21%	3,750			
Total debt		38,750		38,750			
Current maturities		(2,500)		(2,500)			
Long-term debt, less current maturities		\$ 36,250		\$ 36,250			

Credit Agreement

On June 27, 2019, the Company entered into an Amended and Restated Credit Agreement (the "Credit Agreement") with a consortium of lenders and Bank of America, N.A. as administrative agent, which includes a revolving credit note ("U.S. Revolving Note"). The Credit Agreement has a maximum borrowing capacity of \$475,000 and matures on June 27, 2024. The Credit Agreement also provides \$15,000 availability for the issuance of letters of credit and a maximum of \$40,000 for swing line borrowing. Any amount of the facility utilized for letters of credit or swing line loans outstanding will reduce the amount available under the Credit Agreement. The Company had no outstanding letters of credit issued under the Credit Agreement as of March 31, 2022 and December 31, 2021.

The U.S. borrowers and guarantors participating in the Credit Agreement also entered into a related amended and restated pledge and security agreement. The amended and restated pledge and security agreement grants a security interest to the lenders in substantially all of the personal property of the Company and its U.S. subsidiaries designated as borrowers to secure their respective obligations under the Credit Agreement, including the stock and membership interests of specified subsidiaries (limited to 66% of the stock in the case of certain non-U.S. subsidiaries). In addition to the security obligations, all obligations under the Credit Agreement are unconditionally guaranteed by certain of the Company's subsidiaries. The Credit Agreement restricts, among other things, the amount of dividend payments the Company can make to shareholders.

The Credit Agreement contains covenants, that, among other things, (i) prohibit or limit the ability of the borrowers and any material subsidiary to incur additional indebtedness, create liens, pay dividends, make certain types of investments (including acquisitions), enter into certain types of transactions with affiliates, prepay other indebtedness, sell assets, merge with other companies or enter into certain other transactions outside the ordinary course of business, and (ii) require that Gentherm maintain a minimum Consolidated Interest Coverage Ratio and Consolidated Leverage Ratio (based on consolidated EBITDA for the applicable trailing 12-month period as defined in the Credit Agreement) as of the end of any fiscal quarter. The Credit Agreement also contains customary events of default. As of March 31, 2022, the Company was in compliance with the terms of the Credit Agreement.

Under the Credit Agreement, U.S. Dollar denominated loans bear interest at either a base rate ("Base Rate Loans") or Eurocurrency rate ("Eurocurrency Rate Loans"), plus a margin ("Applicable Rate"). The rate for Base Rate Loans is equal to the highest of the Federal Funds Rate (0.33% at March 31, 2022) plus 0.50%, Bank of America's prime rate (3.50% at March 31, 2022), or the Eurocurrency rate plus 1.00%. The rate for Eurocurrency Rate Loans denominated in U.S. Dollars is equal to the London Interbank Offered Rate (0.45% at March 31, 2022). All loans denominated in a currency other than the U.S. Dollar must be Eurocurrency Rate Loans. Interest is payable at least quarterly.

The Applicable Rate varies based on the Consolidated Leverage Ratio reported by the Company. As long as the Company is not in default of the terms and conditions of the Credit Agreement, the lowest and highest possible Applicable Rate is 1.25% and 2.25%, respectively, for Eurocurrency Rate Loans and 0.25% and 1.25%, respectively, for Base Rate Loans.

As of March 31, 2022, \$35,000 was outstanding under the Credit Agreement. Borrowing availability is subject to, among other things, the Company's compliance with the minimum Consolidated Interest Coverage Ratio and Consolidated Leverage Ratio as of the end of any fiscal quarter. Based upon consolidated EBITDA for the trailing twelve months calculated for purposes of the Consolidated Leverage Ratio, \$424,521 remained available as of March 31, 2022 for additional borrowings under the Credit Agreement subject to specified conditions that Gentherm currently satisfies.

DEG Vietnam Loan

The Company also has a fixed interest rate loan with the German Investment Corporation ("DEG"), a subsidiary of KfW Banking Group, a Germany government-owned development bank. The fixed interest rate senior loan agreement with DEG was used to finance the construction and set up of the Vietnam production facility ("DEG Vietnam Loan"). The DEG Vietnam Loan is subject to semi-annual principal payments that began November 2017 and will end May 2023. Under the terms of the DEG Vietnam Loan, the Company must maintain a minimum Enhanced Equity Ratio, as defined by the DEG Vietnam Loan agreement, based on the financial statements of Gentherm's wholly owned subsidiary, Gentherm Vietnam Co. Ltd. As of March 31, 2022, the Company was in compliance with the terms of the DEG Vietnam Loan.

The scheduled principal maturities of our debt as of March 31, 2022 were as follows:

	DEG Vietnam Note			U.S. Revolving Note	 Total
2022	\$	2,500	\$	_	\$ 2,500
2023		1,250			1,250
2024				35,000	35,000
Total	\$	3,750	\$	35,000	\$ 38,750

Note 7 – Commitments and Contingencies

Legal and other contingencies

The Company may be subject to various legal actions and claims in the ordinary course of its business, including those arising out of breach of contracts, intellectual property rights, environmental matters, regulatory matters and employment-related matters. The Company establishes accruals for matters which it believes that losses are probable and can be reasonably estimated. Although it is not possible to predict with certainty the outcome of these matters, the Company is of the opinion that the ultimate resolution of these matters will not have a material adverse effect on its consolidated results of operations or financial position.

Product Liability and Warranty Matters

In the event that the Company's products fail to perform as expected or result in alleged bodily injury or property damage, our products may subject us to warranty claims and product liability. If any of our products are or are alleged to be defective, we may be required to participate in a recall or other corrective action involving such products. The Company maintains liability insurance coverage at levels based on commercial norms and historical claims experience. The Company can provide no assurances that it will not experience material claims in the future or that it will not incur significant costs to defend such claims.

The Company accrues warranty obligations for products sold based on management estimates of future failure rates and current claim cost experience, with support from the sales, engineering, quality and legal functions. Using historical information available to the Company, including claims already filed by customers, the warranty accrual is adjusted quarterly to reflect management's best estimate of future claims.

The following is a reconciliation of the changes in accrued warranty costs:

	Three Months Ended March 31,						
		2022		2021			
Balance at the beginning of the period	\$	1,916	\$	2,391			
Warranty claims paid		(334)		(352)			
Warranty expense for products shipped during the current period		354		519			
Adjustments to warranty estimates from prior periods		(152)		271			
Adjustments due to currency translation		(12)		(31)			
Balance at the end of the period	\$	1,772	\$	2,798			

Other matters

Purchase commitments for materials, supplies, services and capital expenditures, as part of the normal course of business, are generally consistent from year to year. In addition, due to supply shortages of semiconductors, the Company has entered into agreements with various suppliers to reserve the right to purchase certain semiconductor chips over periods of 12-24 months, with volume commitments determined based on our anticipated production requirements. As of March 31, 2022, the Company's total commitments for these semiconductor chip agreements was \$38,919. Such agreements provide the Company with priority access to semiconductor chips as they become available, however, these agreements do not guarantee that our suppliers will meet the timing and quantities requested by Gentherm. All other purchase commitments as of March 31, 2022 were immaterial.

Note 8 - Earnings Per Share

Basic earnings per share are computed by dividing net income by the weighted average number of shares of Common Stock outstanding during the period. The Company's diluted earnings per share give effect to all potential shares of Common Stock outstanding during a period that do not have an anti-dilutive impact to the calculation. In computing the diluted earnings per share, the treasury stock method is used in determining the number of shares assumed to be issued from the exercise of Common Stock equivalents.

The following table illustrates earnings per share and the weighted average shares outstanding used in calculating basic and diluted earnings per share:

	Three Months Ended March 31,						
		2022		2021			
Net income	\$	11,747	\$	32,909			
Basic weighted average shares of Common Stock outstanding		33,034,872		32,945,970			
Dilutive effect of stock options, restricted stock awards and restricted stock units		341,892		443,621			
Diluted weighted average shares of Common Stock outstanding		33,376,764		33,389,591			
Basic earnings per share	\$	0.36	\$	1.00			
Diluted earnings per share	\$	0.35	\$	0.99			

There were no shares excluded from the Company's diluted earnings per share for the three months ended March 31, 2022 and 2021 on the basis that their inclusion would have an anti-dilutive impact on the calculation.

Note 9 - Financial Instruments

Derivative Financial Instruments

The Company is exposed to various market risks including, but not limited to, changes in foreign currency exchange rates, changes in interest rates and price fluctuations of certain material commodities such as copper. Market risks for changes in interest rates relate primarily to its debt obligations under the Credit Agreement. Foreign currency exchange risks are attributable to sales to foreign customers and purchases from foreign suppliers not denominated in a location's functional currency, foreign plant operations, intercompany indebtedness, intercompany investments and include exposures to the Euro, Mexican Peso, Canadian Dollar, Hungarian Forint, North Macedonian Denar, Ukrainian Hryvnia, Japanese Yen, Chinese Renminbi, Korean Won and Vietnamese Dong.

The Company regularly enters into derivative contracts with the objective of managing its financial and operational exposure arising from these risks by offsetting gains and losses on the underlying exposures with gains and losses on the financial instruments used to hedge them. The maximum length of time over which the Company hedges its exposure to foreign currency exchange risks and price fluctuations in material commodities is fifteen months. The Company had foreign currency derivative contracts with a notional value of \$21,180 and \$13,974 and copper commodity swap contracts with a notional value of \$0 and \$309 outstanding as of March 31, 2022 and December 31, 2021, respectively. The principal currency hedged by the Company was the Mexican Peso.

The Company does not enter into derivative financial instruments for speculative or trading purposes. The Company's hedging relationships are formally documented at the inception of the hedge, and hedges must be highly effective in offsetting changes to future cash flows on hedged transactions both at the inception of a hedge and on an ongoing basis to be designated for hedge accounting treatment. For derivative contracts which can be classified as a cash flow hedge, the effective portion of the change in the fair value of the derivative is recorded to Accumulated other comprehensive loss in the consolidated condensed balance sheets. When the underlying hedge transaction is realized, the gain or loss included in Accumulated other comprehensive loss is recorded in earnings in the consolidated condensed statements of income on the same line as the gain or loss on the hedged item attributable to the hedged risk. The Company records the ineffective portion of foreign currency and copper commodity hedging instruments, if any, to cost of sales, in the consolidated condensed statements of income. Cash flows associated with derivatives are reported in net cash (used in) provided by operating activities in the Company's consolidated condensed statements of cash flows.

The Company uses an income approach to value derivative instruments, analyzing quoted market prices to calculate the forward values and then discounting such forward values to the present value using benchmark rates at commonly quoted intervals for the instrument's full term.

Information related to the recurring fair value measurement of derivative instruments in our consolidated condensed balance sheet as of March 31, 2022 is as follows:

			Asset Deriva	atives		Liability Deri						
	Hedge Designation			Fair Value							air ılue	Asset/ pilities)
	Cash flow		Other current			Other current			_			
Foreign currency derivatives	hedge	Level 2	assets	\$	904	liabilities	\$	_	\$ 904			
	Cash flow		Other current			Other current						
Commodity hedges	hedge	Level 2	assets	\$	_	liabilities	\$	_	\$ 			

Information related to the recurring fair value measurement of derivative instruments in our consolidated condensed balance sheet as of December 31, 2021 is as follows:

			Asset Deriva	atives		Liability Deri			
	Hedge Designation	Fair Value Hierarchy	Balance Sheet Location	Fair Value		Balance Sheet Location		air alue	Asset/ bilities)
	Cash flow		Other current			Other current			
Foreign currency derivatives	hedge	Level 2	assets	\$	294	liabilities	\$	_	\$ 294
	Cash flow		Other current			Other current			
Commodity hedges	hedge	Level 2	assets	\$	6	liabilities	\$	_	\$ 6

Information relating to the effect of derivative instruments on our consolidated condensed statements of income and the consolidated condensed statements of comprehensive income is as follows:

		Three Months Ended March 31,								
	Location	2	2022		2021					
Foreign currency derivatives	Cost of sales – income	\$	153	\$	478					
	Other comprehensive income (loss)		610		(667)					
Total foreign currency derivatives		\$	763	\$	(189)					
Commodity derivatives	Cost of sales – income	\$	19	\$	_					
	Other comprehensive loss		(6)							
Total commodity derivatives		\$	13	\$						

The Company did not incur any hedge ineffectiveness during the three months ended March 31, 2022 and 2021.

Accounts Receivable Factoring

The Company is party to a receivable factoring agreement with HSBC Bank USA, National Association. Under the receivable factoring agreement, we can sell receivables for certain of our North America account debtors up to \$41,300, on a revolving basis, subject to outstanding balances and concentration limits. As of March 31, 2022, there were no outstanding receivables transferred under the receivable factoring agreement and our availability under the receivables factoring agreement was \$19,385.

Note 10 - Fair Value Measurements

Fair value is defined as the exchange price that would be received to sell an asset or paid to transfer a liability (an exit price) in the principal or most advantageous market for the asset or liability in an orderly transaction between market participants on the measurement date. Fair value measurements are based on one or more of the following three valuation techniques:

Market: This approach uses prices and other relevant information generated by market transactions involving identical or comparable assets or liabilities.

Income: This approach uses valuation techniques to convert future amounts to a single present value amount based on current market expectations.

Cost: This approach is based on the amount that would be required to replace the service capacity of an asset (replacement cost).

The Company uses the following fair value hierarchy to measure fair value into three broad levels, which are described below:

- Level 1: Quoted prices (unadjusted) in active markets that are accessible at the measurement date for assets or liabilities. The fair value hierarchy gives the highest priority to Level 1 inputs.
- Level 2: Inputs, other than quoted market prices included in Level 1, that are observable either directly or indirectly for the asset or liability.
- Level 3: Unobservable inputs that are used when little or no market data is available. The fair value hierarchy gives the lowest priority to Level 3 inputs.

Items Measured at Fair Value on a Recurring Basis

Except for derivative instruments (see Note 9) and pension plan assets, the Company had no material financial assets and liabilities that were carried at fair value at March 31, 2022 and December 31, 2021. In determining fair value, the Company utilizes valuation techniques that maximize the use of observable inputs and minimize the use of unobservable inputs to the extent possible and also considers counterparty credit risk in its assessment of fair value.

Items Measured at Fair Value on a Nonrecurring Basis

The Company measures certain assets and liabilities at fair value on a non-recurring basis. As these nonrecurring fair value measurements are generally determined using unobservable inputs, these fair value measurements are classified within Level 3 of the fair value hierarchy. As of March 31, 2022 and December 31, 2021, there were no significant assets or liabilities measured at fair value on a non-recurring basis.

Items Not Carried at Fair Value

The Company uses an income valuation technique to measure the fair values of its debt instruments by converting amounts of future cash flows to a single present value amount using rates based on current market expectations (Level 2 inputs). As of March 31, 2022, and December 31, 2021, the carrying values of the indebtedness under the Company's Credit Agreement were not materially different than the estimated fair values because the interest rates on variable rate debt approximated rates currently available to the Company (see Note 6). Discount rates used to measure the fair value of Gentherm's DEG Vietnam Loan are based on quoted swap rates. As of March 31, 2022, the carrying value of the DEG Vietnam Loan was \$3,750 as compared to an estimated fair value of \$3,796. As of December 31, 2021, the carrying value of the DEG Vietnam Loan was \$3,750 as compared to an estimated fair value of \$3,778.

Note 11 – Equity

In December 2020, the Board of Directors of Gentherm Incorporated ("Board of Directors") authorized a stock repurchase program (the "2020 Stock Repurchase Program") to commence upon expiration of the prior stock repurchase program on December 15, 2020. Under the 2020 Stock Repurchase Program, the Company is authorized to repurchase up to \$150,000 of its issued and outstanding common stock over a three-year period, expiring December 15, 2023.

Repurchases may be made, from time to time, in amounts and at prices the Company deems appropriate, subject to market conditions, applicable legal requirements, debt covenants and other considerations. Any such repurchases may be executed using open market purchases, privately negotiated agreements or other transactions. Repurchases may be funded from cash on hand, available borrowings or proceeds from potential debt or other capital markets sources. The Company did not make any repurchases under the 2020 Stock Repurchase Program during the three months ended March 31, 2022, or March 31, 2021. The 2020 Stock Repurchase Program had \$130,000 repurchase authorization remaining as of March 31, 2022.

Note 12 - Reclassifications Out of Accumulated Other Comprehensive Loss

Reclassification adjustments and other activities impacting Accumulated other comprehensive loss during the three months ended March 31, 2022 and 2021 were as follows:

	Defined Benefit Pension Plans	(Ti	Foreign Currency canslation ljustments	(Foreign Currency Hedge erivatives	Commod Hedge Derivati	e		Total
Balance at December 31, 2021	\$ (2,893)	\$	(34,188)	\$	154	\$	5	\$	(36,922)
Other comprehensive (loss) income before									
reclassifications	_		(9,154)		763		13		(8,378)
Income tax effect of other comprehensive loss before									
reclassifications	_		(140)		(179)		(3)		(322)
Amounts reclassified from accumulated other									
comprehensive loss into net income	35				(153) a	ı	(19)	a	(137)
Income taxes reclassified into net income	(6)				33		4		31
Net current period other comprehensive income (loss)	29		(9,294)		464		(5)		(8,806)
Balance at March 31, 2022	\$ (2,864)	\$	(43,482)	\$	618	\$		\$	(45,728)

⁽a) The amounts reclassified from accumulated other comprehensive loss were included in cost of sales.

	Defined Benefit Pension Plans	7	Foreign Currency Translation djustments	C	Foreign Currency Hedge erivatives		Total
Balance at December 31, 2020	\$ (3,451)	\$	(12,637)	\$	1,106	\$	(14,982)
Other comprehensive loss before reclassifications	_		(12,872)		(189)		(13,061)
Income tax effect of other comprehensive loss before							
reclassifications	_		(280)		41		(239)
Amounts reclassified from accumulated other comprehensive							
loss into net income	_		_		(478) a	ı	(478)
Income taxes reclassified into net income	 _		<u> </u>		104		104
Net current period other comprehensive loss	_		(13,152)		(522)		(13,674)
Balance at March 31, 2021	\$ (3,451)	\$	(25,789)	\$	584	\$	(28,656)

⁽a) The amounts reclassified from accumulated other comprehensive loss were included in cost of sales.

The Company expects all of the existing gains and losses related to foreign currency and commodity hedge derivatives reported in Accumulated other comprehensive loss as of March 31, 2022 to be reclassified into earnings during the next twelve months. See Note 9 for additional information about derivative financial instruments and the effects from reclassification to net income.

Note 13 – Income Taxes

At the end of each interim period, the Company makes its best estimate of the annual expected effective income tax rate and applies that rate to its ordinary year-to-date earnings or loss. The income tax provision or benefit related to unusual or infrequent items, if applicable, that will be separately reported or reported net of their related tax effects are individually computed and recognized in the interim period in which those items occur. In addition, the effect of changes in enacted tax laws or rates, tax status, judgment on the realizability of a beginning-of-the-year deferred tax asset in future years or income tax contingencies is recognized in the interim period in which the change occurs.

The computation of the annual expected effective income tax rate at each interim period requires certain estimates and assumptions including, but not limited to, the expected pre-tax income (or loss) for the year, projections of the proportion of income (and/or loss) earned and taxed in respective jurisdictions, permanent and temporary differences, and the likelihood of the realizability of deferred tax assets generated in the current year. Jurisdictions with a projected loss for the year for which no tax benefit can be recognized due to a valuation allowance are excluded from the estimated annual effective tax rate. The impact of such an exclusion could result in a higher or lower effective tax rate during a particular quarter, based upon the composition and timing of actual earnings compared to annual projections. The estimates used to compute the provision or benefit for income taxes may change as new events occur, additional information is obtained or as our tax environment changes. To the extent that the expected annual effective income tax rate changes, the effect of the change on prior interim periods is included in the income tax provision in the period in which the change in estimate occurs.

A summary of the provision for income taxes and the corresponding effective tax rate for the three months ended March 31, 2022 and 2021, is shown below:

		Three Months Ended March 31,						
		2022		2021				
Income tax expense	\$	4,295	\$	7,565				
Earnings before income tax	\$	16,042	\$	40,474				
Effective tax rate	_	26.8%		18.7%				

Income tax expense was \$4,295 for the three months ended March 31, 2022, on earnings before income tax of \$16,042, representing an effective tax rate of 26.8%. The effective tax rate differed from the U.S. Federal statutory rate of 21.0% primarily due to the impact of income taxes on foreign earnings taxed at rates varying from the U.S. statutory rate, the unfavorable impact of the global intangible low-tax income ("GILTI"), and the quarterly accrual for uncertain tax positions, partially offset by the impact of certain favorable tax effects on equity vesting.

Income tax expense was \$7,565 for the three months ended March 31, 2021, on earnings before income tax of \$40,474, representing an effective tax rate of 18.7%. The effective tax rate differed from the U.S. Federal statutory rate of 21.0% primarily due to certain intercompany transactions, partially offset by unfavorable impact of the global intangible low-tax income ("GILTI").

Note 14 – Segment Reporting

Segment information is used by management for making strategic operating decisions for the Company. Management evaluates the performance of the Company's segments based primarily on operating income or loss.

The Company's reportable segments are as follows:

- *Automotive* this segment represents the design, development, manufacturing and sales of automotive climate comfort systems, automotive cable systems, battery performance solutions, and automotive electronic and software systems.
- Medical this segment represents the results from our patient temperature management business within the medical industry.

The *Corporate* category includes unallocated costs related to our corporate headquarter activities, including selling, general and administrative costs and acquisition transaction costs, which do not meet the requirements for being classified as an operating segment.

The tables below present segment information about the reported product revenues, depreciation and amortization and operating income (loss) of the Company for three months ended March 31, 2022 and 2021.

Three Months Ended March 31,	Automotive		Medical		Medical Corporate		Total	
2022:				_				
Product revenues	\$	257,864	\$	9,793	\$	_	\$	267,657
Depreciation and amortization		8,667		604		306	\$	9,577
Operating income (loss)		31,275		(851)		(16,234)	\$	14,190
2021:								
Product revenues	\$	279,370	\$	9,165	\$		\$	288,535
Depreciation and amortization		9,053		584		217	\$	9,854
Operating income (loss)		53,116		(531)		(11,836)	\$	40,749
	18							

Automotive and Medical segment product revenues by product category for the three months ended March 31, 2022 and 2021 were as follows:

	 Three Months Ended March 31,						
	2022		2021				
Climate Control Seat	\$ 102,734	\$	109,173				
Seat Heaters	68,896		76,721				
Steering Wheel Heaters	28,736		28,864				
Automotive Cables	22,045		24,281				
Battery Performance Solutions	17,613		17,760				
Electronics	10,828		15,105				
Other Automotive	7,012		7,466				
Subtotal Automotive segment	257,864		279,370				
Medical segment	9,793		9,165				
Total Company	\$ 267,657	\$	288,535				

Total product revenues information by geographic area for the three months ended March 31, 2022 and 2021 is as follows (based on shipment destination):

	 Three Months Ended March 31,						
	 2022		2021				
United States	\$ 104,122	\$	115,287				
China	38,353		33,515				
South Korea	21,175		24,640				
Germany	19,786		19,180				
Romania	12,755		15,085				
Japan	11,816		17,920				
Other	 59,650		62,908				
Total Non-U.S.	163,535		173,248				
Total Company	\$ 267,657	\$	288,535				

Note 15 – Subsequent Events

On May 4, 2022, the Company entered into a definitive purchase agreement to acquire the automotive business of Alfmeier Präzision SE ("Alfmeier"), the global leader in automotive lumbar and massage comfort solutions and a leading provider of advanced valve systems technology, integrated electronics and software.

Under the terms of the agreement, the Company will acquire all shares in Alfmeier for €177,500, net of cash and debt, subject to customary adjustments related to Alfmeier's net working capital as of the closing and may be increased by up to €4,000 at or post-closing upon resolution of certain Alfmeier tax matters. The transaction will be funded through a combination of the Company's existing cash balances and revolving credit facility. The transaction is expected to close during the third quarter of 2022 subject to the receipt of regulatory approvals and the satisfaction or waiver of other customary closing conditions.

Also on May 4, 2022, the Company entered into a put and call option agreement for the real property in Germany on which Alfmeier's headquarters is located. Pursuant to this agreement, the seller will have a put to sell the headquarters and the Company will have an option to purchase the headquarters post-closing of the acquisition of Alfmeier.

ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

Forward-Looking Statements

This Quarterly Report on Form 10-Q (this "Report") contains forward-looking statements within the meaning of the "safe harbor" provisions of the Private Securities Litigation Reform Act of 1995. These forward-looking statements represent our goals, beliefs, plans and expectations about our prospects for the future and other future events, such as the impact of the COVID-19 pandemic on our financial statements, liquidity, and business as well as the global economy, global supply chain and automotive and medical industries, the impact of the conflict in Ukraine on our operations, the timing of closing of the acquisition of the Alfmeier transaction described herein (the "Acquisition"), the expected sources of funds to be used for as consideration to be paid in the Acquisition, the nature of the closing conditions for the Acquisition, the expected synergies and growth prospects following the closing of the Acquisition, the significant supply disruptions and shifts in the labor market currently faced by the automotive and medical industries, our ability and our customers' ability to maintain production levels, the amount of borrowing availability under the Credit Agreement and the sufficiency of our cash balances and cash generated from operating, investing and financing activities for our future liquidity and capital resource needs and our ability to finance sufficient working capital. Reference is made in particular to forward-looking statements included in this "Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations". Such statements may be identified by the use of forward-looking terminology such as "may", "will", "expect", "believe", "estimate", "anticipate", "intend", "continue", or similar terms, variations of such terms or the negative of such terms. The forward-looking statements included in this Report are made as of the date hereof or as of the date specified herein and are based on management's current expectations and beliefs. Such statements are subject to a number of assumptions, risks, uncertainties and other factors, which are set forth in "Item 1A. Risk Factors" and elsewhere in our Annual Report on Form 10-K for the year ended December 31, 2021, Part II "Item 1A. Risk Factors" in this Report and subsequent reports filed with the Securities and Exchange Commission, and which could cause actual results to differ materially from that described in the forward-looking statements. In addition, except for the Acquisition, such forward-looking statements do not include the potential impact of any other business combinations, acquisitions, divestitures, strategic investments and other significant transactions that may be completed after the date hereof, each of which may present material risks to the Company's business and financial results. Except as required by law, we expressly disclaim any obligation or undertaking to update any forward-looking statements to reflect any change in our expectations with regard thereto or any change in events, conditions or circumstances on which any such statement is based.

The following discussion and analysis should be read in conjunction with, and is qualified in its entirety by, our consolidated condensed financial statements and related notes thereto included elsewhere in this Report and our consolidated financial statements and related notes included in our Annual Report on Form 10-K for the year ended December 31, 2021.

Overview

Gentherm Incorporated is a global developer, manufacturer and marketer of innovative thermal management technologies for a broad range of heating and cooling and temperature control applications in the automotive and medical industries. Within the automotive industry, our products provide solutions for passenger climate comfort and convenience, battery thermal management and cell connecting systems. Within the medical industry our products provide patient temperature management solutions. Our automotive products can be found on vehicles manufactured by nearly all the major OEMs operating in North America and Europe, and several major OEMs in Asia. We operate in locations aligned with our major customers' product strategies to provide locally enhanced design, integration and production capabilities. The Company is also developing a number of new technologies and products that are expected to enable improvements to existing products and to create new product applications for existing and new markets.

Our sales are driven by the number of vehicles produced by the OEMs, which is ultimately dependent on consumer demand for automotive vehicles, our product content per vehicle, and other factors that may limit or otherwise impact production by us, our supply chain and our customers. Historically, new vehicle demand and product content (i.e. vehicle features) have been driven by macroeconomic and other factors, such as interest rates, automotive manufacturer and dealer sales incentives, fuel prices, consumer confidence, employment levels, income growth trends and government and tax incentives. Vehicle content has also been driven by trends in consumer preferences, such as preferences for smart devices and features, personalized user experience, and comfort, health and wellness. Economic volatility or weakness, as well as geopolitical factors, in North America, Europe or Asia, have had and could result in a significant reduction in automotive sales and production by our customers, which have and would have an adverse effect on our business, results of operations and financial condition. In 2020 and 2021, and continuing into 2022, the automotive industry has experienced fluctuating demand and production disruptions related to supply chain challenges, facility closures, labor shortages, work stoppages and inflationary pressures, as a result of the COVID-19 pandemic and associated macroeconomic conditions, as described below. We believe our diversified OEM customer base and geographic revenue base, along with our flexible cost structure, have well positioned us to withstand the impact of industry downturns, including the ongoing impact of the COVID-19 pandemic and associated

economic conditions, and benefit from industry upturns in the ordinary course. However, shifts in the mix of global automotive production to higher cost regions or to vehicles with less of our product content as well as continuing production challenges and inflationary pressures could adversely impact our profitability. In addition, we may be adversely impacted by volatility, weakness or accelerated growth in markets for hybrid or electric vehicles specifically. We believe our products offer certain advantages for hybrid and electric vehicles, including improved energy efficiency, and position us well to withstand changes in the volume mix between vehicles driven by internal combustion engines and hybrid and other electric vehicles. We also believe that products we are developing, such as ClimateSense®, position us well to address trends in consumer preferences such as personalized user experience, comfort, health and wellness.

Recent Trends

General Economic Conditions

The COVID-19 pandemic that began around December 2019 introduced significant volatility to the global economy, disrupted supply chains and had a widespread adverse effect on the global automotive industry in the first half of 2020, with various direct and indirect adverse impacts continuing throughout 2021 and into 2022.

Beginning in February 2020 and continuing into June 2020, substantially all of the Company's major OEM and Tier 1 customers temporarily ceased or significantly reduced production as a result of restrictions that were requested or mandated by governmental authorities. As a result, substantially all of our manufacturing facilities either temporarily suspended production or experienced significant reductions in volumes during this period. By the end of the second quarter of 2020, the Company had reopened all of its manufacturing facilities, in line with industry demand, and in accordance with local government requirements. Although global automotive industry production has improved relative to the first half of 2020, production remains below recent historic levels.

The lingering impacts of COVID-19 throughout 2021 and into 2022 have impeded global supply chains, resulted in longer lead times and delays in procuring component parts and raw materials, and resulted in inflationary cost increases in certain raw materials, labor and transportation. These broad-based inflationary impacts have negatively impacted the Company's financial condition, results of operations and cash flows throughout 2021 and into 2022. We expect these inflationary impacts to continue for the foreseeable future.

Supply shortages of semiconductor chips and other components have resulted in decreases in global automotive vehicle production and significant volatility in customer vehicle production schedules. The Company's semiconductor suppliers, along with most automotive component supply companies that use semiconductors, including Gentherm, have been unable to fully meet the vehicle production demands of the OEMs due to events which are outside the Company's control, including but not limited to, the COVID-19 pandemic, the global semiconductor shortage, fires at suppliers' facilities, significant weather events impacting semiconductor supplier facilities in the southern United States, and other extraordinary events. The Company was able to mitigate the impacts of supply chain disruptions in order to satisfy customer orders during the first three quarters of 2021; however, during the fourth quarter of 2021 and continuing into 2022 we have experienced and may continue to experience direct adverse impacts of ongoing shortages of semiconductors. Our ability to meet customer orders without significant delay and/or expense for 2022 and beyond remains subject to significant uncertainty.

In response to the global supply chain instability and inflationary cost increases the Company has taken several actions to minimize any potential and actual adverse impacts by working closely with its suppliers and customers to closely monitor the availability of semiconductor microchips and other component parts and raw materials, customer vehicle production schedules and any other supply chain inefficiencies that may arise. We expect global supply chain instability will continue to have an adverse impact on our business and financial performance for the foreseeable future, and such adverse impact may be material. The consequences of the pandemic, global supply chain instability and inflationary cost increases and their adverse impact to the global economy continue to evolve. Accordingly, the significance of the future adverse impact on our business and financial statements remains subject to significant uncertainty as of the date of this filing.

In addition to the direct and indirect impacts of COVID-19, the United States and global markets are experiencing volatility and disruption following the escalation of geopolitical tensions and the start of the military conflict between Russia and Ukraine. In February 2022, Russia launched a full-scale military invasion of Ukraine. As a result of the conflict, the United States, United Kingdom, European Union and other countries have levied economic sanctions and bans on Russia and Russia has responded with its

own retaliatory measures. These measures have impacted the availability and price of certain raw materials and could have a lasting impact on regional and global economies.

Our facility in Vynohradiv is on the far western corner of Ukraine near the Hungary border. In 2021, products manufactured at our Ukraine facility represented approximately 11% of the Company's total revenue, including automotive cables, seat heaters and steering wheel heaters. At this time, our Ukraine facility is operating at normal levels and we have begun executing contingency plans and, in coordination with certain customers, specific equipment and production relocations leveraging our flexible global manufacturing footprint. Our response to the escalating situation is based on a severity level contingency response plan that has been developed with certain customers. As the situation in Ukraine is very fluid, we continue to monitor its effects on our business and we continue to work closely with our customers to adjust our contingency response as necessary. See Part II, Item. 1A., "Risk Factors" in this Report for additional information on risks and uncertainties related to the Ukraine conflict.

Alfmeier Acquisition

On May 4, 2022, the Company entered into a definitive purchase agreement to acquire the automotive business of Alfmeier Präzision SE ("Alfmeier"), the global leader in automotive lumbar and massage comfort solutions and a leading provider of advanced valve systems technology, integrated electronics and software.

Under the terms of the agreement, the Company will acquire all shares in Alfmeier for €177.5 million, net of cash and debt, subject to customary adjustments related to Alfmeier's net working capital as of the closing and may be increased by up to €4.0 million at or post-closing upon resolution of certain Alfmeier tax matters. The transaction will be funded through a combination of the Company's existing cash balances and revolving credit facility. The transaction is expected to close during the third quarter of 2022 subject to the receipt of regulatory approvals and the satisfaction or waiver of other customary closing conditions.

Also on May 4, 2022, the Company entered into a put and call option agreement for the real property in Germany on which Alfmeier's headquarters is located. Pursuant to this agreement, the seller will have a put to sell the headquarters and the Company will have an option to purchase the headquarters post-closing of the acquisition of Alfmeier. See Part II, Item 5, "Other Information" in this Report for more information about the acquisition of Alfmeier.

Light Vehicle Production Volumes

Our sales are driven by the number of vehicles produced by the automotive manufacturers, which is ultimately dependent on consumer demand for automotive vehicles, and our content per vehicle, and other factors that may limit or otherwise impact production by us, our supply chain and our customers. According to the forecasting firm IHS Markit (April 2022 release), global light vehicle production in the first quarter of 2022 in the Company's key markets of North America, Europe, China, Japan and Korea, as compared to the first quarter of 2021, are shown below (in millions of units):

	Three M	Three Months Ended March 31,						
	2022	2021	% Change					
North America	3.6	3.6	0.0%					
Europe	3.9	4.7	(17.0)%					
Greater China	6.2	5.9	5.1%					
Japan / South Korea	2.7	3.1	(12.9)%					
Total light vehicle production volume in key markets	16.4	17.3	(5.3)%					

The IHS Markit (April 2022 release) forecasted light vehicle production volume in the Company's key markets for full year 2022 to increase to 67.2 million units, a 4.0% increase from full year 2021 light vehicle production volumes. Forecasted light vehicle production volumes are a component of the data we use in forecasting future business. However, these forecasts generally are updated monthly, and future forecasts may be significantly different from period to period due to changes in macroeconomic conditions or matters specific to the automotive industry, such as the fluctuations that occurred in 2020 and remain ongoing due to the COVID-19 pandemic. Further, due to differences in regional product mix at our manufacturing facilities, as well as material production schedules from our customers for our products on specific vehicle programs, our future forecasted results do not directly correlate with the global and/or regional light vehicle production forecasts of IHS Markit or other third-party sources.

New Business Awards

We believe that innovation is an important element to gaining market acceptance of our products and strengthening our market position. During the first quarter of 2022, we secured an estimated \$170 million of automotive new business awards. Automotive new

business awards represent the aggregate projected lifetime revenue of new awards provided by our customers to Gentherm in the applicable period, with the value based on the price and volume projections received from each customer as of the award date. Although automotive new business awards are not firm customer orders, we believe that new business awards are an indicator of future revenue. New business awards are not projections of revenue or future business as of March 31, 2022, the date of this Report or any other date. Customer projections regularly change over time and we do not update our calculation of any new business award after the date initially communicated. Automotive new business awards in the first quarter 2022 also do not reflect, in particular, the impact of the COVID-19 pandemic and related macroeconomic challenges on future business. Revenues resulting from automotive new business awards also are subject to additional risks and uncertainties that are included in this Report or incorporated by reference in "Forward-Looking Statements" above.

Stock Repurchase Program

On December 11, 2020, the Board of Directors authorized a new stock repurchase program (the "2020 Stock Repurchase Program") to commence upon expiration of the prior stock repurchase program on December 15, 2020. Under the 2020 Stock Repurchase Program, the Company is authorized to repurchase up to \$150 million of its issued and outstanding Common Stock over a three-year period, expiring December 15, 2023. Repurchases under the 2020 Stock Repurchase Program may be made, from time to time, in amounts and at prices the Company deems appropriate, subject to market conditions, applicable legal requirements, debt covenants and other considerations. During the three months ended March 31, 2022, we did not make any repurchases under the 2020 Stock Repurchase Program and have a remaining repurchase authorization of \$130.0 million as of March 31, 2022.

Reportable Segments

The Company has two reportable segments for financial reporting purposes: Automotive and Medical.

See Note 14, "Segment Reporting", to the consolidated condensed financial statements included in this Report for a description of our reportable segments as well as their proportional contribution to the Company's reported product revenues and operating income. The financial information used by our chief operating decision maker to assess operating performance and allocate resources is based on these reportable segments.

Consolidated Results of Operations

The results of operations for the three months ended March 31, 2022 and 2021, in thousands, were as follows:

2022			2021		vorable / favorable)
\$	267,657	\$	288,535	\$	(20,878)
	203,544		200,866		(2,678)
	64,113		87,669		(23,556)
	20,434		17,603		(2,831)
	29,308		28,526		(782)
	181		791		610
	49,923		46,920		(3,003)
	14,190		40,749		(26,559)
	(569)		(1,039)		470
	2,217		773		1,444
	204		(9)		213
	16,042		40,474		(24,432)
	4,295		7,565		3,270
\$	11,747	\$	32,909	\$	(21,162)
		\$ 267,657 203,544 64,113 20,434 29,308 181 49,923 14,190 (569) 2,217 204 16,042 4,295	\$ 267,657 \$ 203,544 64,113 20,434 29,308 181 49,923 14,190 (569) 2,217 204 16,042 4,295	\$ 267,657 \$ 288,535 203,544 200,866 64,113 87,669 20,434 17,603 29,308 28,526 181 791 49,923 46,920 14,190 40,749 (569) (1,039) 2,217 773 204 (9) 16,042 40,474 4,295 7,565	\$ 267,657 \$ 288,535 \$ 203,544 200,866 64,113 87,669 20,434 17,603 29,308 28,526 181 791 49,923 46,920 14,190 40,749 (569) (1,039) 2,217 773 204 (9) 16,042 40,474 4,295 7,565

Product revenues by product category, in thousands, for the three months ended March 31, 2022 and 2021, were as follows:

	Three Months Ended March 31,								
	 2022		2021	% Change					
Climate Control Seat	\$ 102,734	\$	109,173	(5.9)%					
Seat Heaters	68,896		76,721	(10.2)%					
Steering Wheel Heaters	28,736		28,864	(0.4)%					
Automotive Cables	22,045		24,281	(9.2)%					
Battery Performance Solutions	17,613		17,760	(0.8)%					
Electronics	10,828		15,105	(28.3)%					
Other Automotive	7,012		7,466	(6.1)%					
Subtotal Automotive segment	 257,864		279,370	(7.7)%					
Medical segment	9,793		9,165	6.9%					
Total Company	\$ 267,657	\$	288,535	(7.2)%					

Product Revenues

Below is a summary of our product revenues, in thousands, for the three months ended March 31, 2022 and 2021:

	Three N	Ionths Ende	ed March 31,	Variance Due To:						
			Favorable /	Automotive		Pricing/				
	2022	2021	(Unfavorable)	Volume	FX	Other	Total			
Product revenues	\$267,657	\$288,535	\$ (20,878)	\$ (12,970)	\$ (6,601)	\$ (1,307)	\$ (20,878)			

Product revenues for the three months ended March 31, 2022 decreased 7.2% as compared to the three months ended March 31, 2021. The decrease in product revenues is primarily related to decreased volumes in our Automotive segment and unfavorable foreign currency impacts, primarily related to the Euro, Chinese Renminbi and Korean Won. Product revenues decreased across all Automotive segment product lines, including decreases of \$7.8 million, \$6.4 million, \$4.3 million, and \$2.2 million in Seat Heaters, Climate Control Seat, Electronics and Automotive Cables, respectively, primarily due to a decrease in light vehicle production across relevant markets. The decrease in product revenues included in Variance Due To Pricing/Other above is primarily attributable to decreases in customer pricing in our Automotive segment offset by an increase in product revenue in our Medical segment.

Cost of Sales

Below is a summary of our cost of sales and gross margin, in thousands, for the three months ended March 31, 2022 and 2021:

	Three N	Three Months Ended March 31,				Variance Due To:								
	2022	2021		Favorable / Automotive (Unfavorable) Volume		· · · · · · · · · · · · · · · · · · ·		oerational rformance	FX	Other	Total			
Cost of sales	\$203,544	\$200,866	\$	(2,678)	\$	8,348	\$	(12,008)	\$ 3,917	\$(2,935)	\$ (2,678)			
Gross margin	\$ 64,113	\$ 87,669	\$	(23,556)	\$	(4,622)	\$	(12,008)	\$(2,684)	\$(4,242)	\$(23,556)			
Gross margin - Percentage of														
product revenues	24.0%	30.4%	Ó											

Cost of sales for the three months ended March 31, 2022 increased 1.3% as compared to the three months ended March 31, 2021. The increase in cost of sales within the Variance Due To Operational Performance is primarily related to inflation associated with higher freight costs and material costs. This increase was offset by decreased volumes in our Automotive segment and favorable foreign currency impacts primarily attributable to the Euro and Chinese Renminbi. The increase in cost of sales included in Variance Due To Other above is due to the following items:

- \$2.6 million of increase due to wage inflation;
- \$0.7 million of increase attributable to higher volumes in the Medical segment; and
- \$0.6 million of decrease due to lower stock compensation expense.

Net Research and Development Expenses

Below is a summary of our net research and development expenses, in thousands, for the three months ended March 31, 2022 and 2021:

		Three Months Ended March 31,								
		2022		2021	Favorable / (Unfavorable)					
Research and development expenses	\$	23,903	\$	22,426	\$	(1,477)				
Reimbursed research and development expenses		(3,469)		(4,823)		(1,354)				
Net research and development expenses	\$	20,434	\$	17,603	\$	(2,831)				
Percentage of product revenues	_	7.6%		6.1%						

Net research and development expenses for the three months ended March 31, 2022 increased 16.1% as compared to the three months ended March 31, 2021. The increase in net research and development expenses is primarily related to increased project-related spending and lower reimbursements for costs to design, develop and purchase tooling pursuant to customer contracts.

Selling, General and Administrative Expenses

Below is a summary of our selling, general and administrative expenses, in thousands, for the three months ended March 31, 2022 and 2021:

	 Three Months Ended Marc							
	2022		2021	Favorable / (Unfavorable)				
Selling, general and administrative expenses	\$ 29,308	\$	28,526	\$	(782)			
Percentage of product revenues	10.9%)	9.9%					

Selling, general and administrative expenses for the three months ended March 31, 2022 increased 2.7% as compared to the three months ended March 31, 2021. The increase in selling, general and administrative expenses is primarily related to acquisition related costs partially offset by lower stock compensation expense.

Restructuring Expenses

Below is a summary of our restructuring expenses, in thousands, for the three months ended March 31, 2022 and 2021:

	_		Three	ch 31,			
		20	022	2021		Favorable / (Unfavorable)	
s	9	\$	181	\$ 791	\$	610	

Restructuring expenses primarily relate to the Manufacturing Footprint Rationalization restructuring program and other discrete restructuring actions focused on the rotation of our manufacturing footprint to lower cost locations and the reduction of global overhead expenses.

During the three months ended March 31, 2022, the Company recognized expenses of \$0.1 million for employee separation costs and \$0.1 million for other costs.

During the three months ended March 31, 2021, the Company recognized expenses of \$0.5 million for employee separation costs and \$0.3 million for other costs.

See Note 3, "Restructuring" of the consolidated condensed financial statements included in this Report for additional information.

Interest Expense

Below is a summary of our interest expense, in thousands, for the three months ended March 31, 2022 and 2021:

	 Three	ch 31,	h 31,	
			Favo	rable /
	2022	2021	(Unfav	vorable)
Interest expense, net	\$ (569)	\$ (1,039)	\$	470

Interest expense, net for the three months ended March 31, 2022 decreased 45.2% as compared to the three months ended March 31, 2021. The decrease is due to a lower balance on our revolving credit agreement during the three months ended March 31, 2022, as compared to the three months ended March 31, 2021.

Foreign Currency Gain

Below is a summary of our foreign currency gain, in thousands, for the three months ended March 31, 2022 and 2021:

	 Three	Month	ıs Ended Mar	ch 31,	
	2022		2021		vorable / favorable)
Foreign currency gain	\$ 2,217	\$	773	\$	1,444

Foreign currency gain for the three months ended March 31, 2022 included net realized foreign currency loss of \$0.1 million and net unrealized foreign currency gain of \$2.3 million.

Foreign currency gain for the three months ended March 31, 2021 included net realized foreign currency gain of \$0.5 million and net unrealized foreign currency gain of \$0.3 million.

Other Income (Loss)

Below is a summary of our other income (loss), in thousands, for the three months ended March 31, 2022 and 2021:

	_		Three	Month	s Ended Mar	ch 31,	
	_	2022			2021		vorable / avorable)
Other income (loss)	9	\$	204	\$	(9)	\$	213

The increase in Other income (loss) is due to an increase in miscellaneous income.

Income Tax Expense

Below is a summary of our income tax expense, in thousands, for the three months ended March 31, 2022 and 2021:

	_		Three	Mont	hs Ended Mar	ch 31,	
	_						vorable /
	_	2022			2021	(Uni	favorable)
Income tax expense	9	\$	4,295	\$	7,565	\$	3,270

Income tax expense was \$4.3 million for the three months ended March 31, 2022, on earnings before income tax of \$16.0 million, representing an effective tax rate of 26.8%. The effective tax rate differed from the U.S. Federal statutory rate of 21.0% primarily due to the impact of income taxes on foreign earnings taxed at rates varying from the U.S. statutory rate, the unfavorable impact of the global intangible low-tax income ("GILTI"), and the quarterly accrual for uncertain tax positions partially offset by the impact of certain favorable tax effects on equity vesting.

Income tax expense was \$7.6 million for the three months ended March 31, 2021, on earnings before income tax of \$40.5 million, representing an effective tax rate of 18.7%. The effective tax rate differed from the U.S. Federal statutory rate of 21% primarily due to certain intercompany transactions, partially offset by unfavorable impact of the global intangible low-tax income ("GILTI").

Liquidity and Capital Resources

Overview

Our primary sources of liquidity and capital resources are cash flows from operations and borrowings available under our Credit Agreement. Our cash requirements consist principally of working capital, capital expenditures, research and development, operating lease payments, income tax payments and general corporate purposes. We generally reinvest available cash flows from operations into our business, while opportunistically utilizing our authorized stock repurchase program. Further, we continuously evaluate acquisition and investment opportunities that will enhance our business strategies.

As of March 31, 2022, the Company had \$177.9 million of cash and cash equivalents, \$424.5 million of availability under our Credit Agreement and \$19.4 million of availability under our North America receivables factoring arrangement. We also continue to maintain access to the capital markets and may issue debt or equity securities, which may provide an additional source of liquidity. There can be no assurance equity or debt financing will be available to us when we need it or, if available, the terms will be satisfactory to us and not dilutive to our then-current shareholders.

We continue to expect to be able to move funds between different countries to manage our global liquidity needs without material adverse tax implications, subject to current monetary policies and the terms of the Credit Agreement. We utilize a combination of strategies, including dividends, cash pooling arrangements, intercompany loan repayments and other distributions and advances to provide the funds necessary to meet our global liquidity needs. There are no significant restrictions on the ability of our subsidiaries to pay dividends or make other distributions to Gentherm Incorporated. As of March 31, 2022, the Company's cash and cash equivalents held by our non-U.S. subsidiaries totaled approximately \$157.2 million. If additional non-U.S. cash was needed for our U.S. operations, we may be required to accrue and pay withholding if we were to distribute such funds from non-U.S. subsidiaries to the U.S.; however, based on our current liquidity needs and strategies, we do not anticipate a need to accrue and pay such additional amounts.

We currently believe that our cash and cash equivalents and borrowings available under our Credit Agreement and the North America receivables factoring arrangement will be adequate to meet anticipated cash requirements for at least the next twelve months and the foreseeable future.

Cash and Cash Flows

The following table represents our cash and cash equivalents, in thousands:

		Three Months Ended March 31,						
	2022			2021				
Cash and cash equivalents at beginning of period	\$	190,606	\$	268,345				
Net Cash (used in) provided by operating activities		(684)		39,667				
Net Cash used in investing activities		(5,957)		(10,103)				
Net Cash used in financing activities		(3,750)		(125,616)				
Foreign currency effect on cash and cash equivalents		(2,298)		(1,338)				
Cash and cash equivalents at end of period	\$	177,917	\$	170,955				

Cash Flows From Operating Activities

Cash used in operating activities totaled \$0.7 million during the three months ended March 31, 2022 primarily reflecting net income of \$11.7 million, decreased by non-cash adjustments of \$0.8 million for deferred income taxes and \$23.9 million related to changes in assets and liabilities, partially offset by \$12.3 million for non-cash charges for depreciation, amortization, non-cash stock based compensation, pension plan adjustments and loss on disposition of property and equipment.

Cash Flows From Investing Activities

Cash used in investing activities was \$6.0 million during the three months ended March 31, 2022, reflecting purchases of property and equipment of \$5.6 million and payments related to an equity method investment for \$0.4 million.

Cash Flows From Financing Activities

Cash used in financing activities was \$3.8 million during the three months ended March 31, 2022, reflecting taxes withheld and paid on employees' share based payment awards totaling \$4.3 million, partially offset by proceeds from the exercise of Common Stock options totaling \$0.5 million.

Debt

The following table summarizes the Company's debt, in thousands, as of March 31, 2022 and 2021:

_	March 31, 2022			March 3	1	
	Interest Rate		Principal Balance	Interest Rate		Principal Balance
Amended Credit Agreement:	_			_		_
U.S. Revolving Note (U.S. Dollar denominations)	1.71%	\$	35,000	1.61%	\$	41,500
U.S. Revolving Note (Euro denominations)	_			1.50%		14,069
DEG Vietnam Loan	5.21%		3,750	5.21%		6,250
Total debt			38,750			61,819
Current maturities			(2,500)			(2,500)
Long-term debt, less current maturities		\$	36,250		\$	59,319

Credit Agreement

Gentherm, together with certain of its subsidiaries, maintain a revolving credit note ("U.S. Revolving Note") under its Amended and Restated Credit Agreement (the "Credit Agreement") with a consortium of lenders and Bank of America, N.A. as administrative agent. The Credit Agreement has a maximum borrowing capacity of \$475 million and matures on June 27, 2024. The Credit Agreement contains covenants, that, among other things, (i) prohibit or limit the ability of the borrowers and any material subsidiary to incur additional indebtedness, create liens, pay dividends, make certain types of investments (including acquisitions), enter into certain types of transactions with affiliates, prepay other indebtedness, sell assets, merge with other companies or enter into certain

other transactions outside the ordinary course of business, and (ii) require that Gentherm maintain a minimum Consolidated Interest Coverage Ratio and Consolidated Leverage Ratio (based on consolidated EBITDA for the applicable trailing 12-month period as defined in the Credit Agreement) as of the end of any fiscal quarter.

DEG Vietnam Loan

The Company also has a fixed interest rate loan with the German Investment Corporation ("DEG"), a subsidiary of KfW Banking Group, a Germany government-owned development bank. The fixed interest rate senior loan agreement with DEG was used to finance the construction and set up of the Vietnam production facility ("DEG Vietnam Loan"). The DEG Vietnam Loan is subject to semi-annual principal payments that began November, 2017 and will end May, 2023.

Other Sources of Liquidity

The Company is party to a receivable factoring agreement with HSBC Bank USA, National Association. Under the receivable factoring agreement, we can sell receivables for certain of our North America account debtors up to \$41.3 million, on a revolving basis, subject to outstanding balances and concentration limits. As of March 31, 2022, there were no outstanding receivables transferred under the receivable factoring agreement and our availability under the receivables factoring agreement was \$19.4 million.

Material Cash Requirements

Significant liquidity will be required during the second half of 2022 to close the Alfmeier Acquisition. Under terms of the definitive purchase agreement, the purchase price is epsilon 177.5 million, net of cash and debt, subject to customary adjustments related to Alfmeier's net working capital as of the closing and may be increased by up to epsilon 4.0 million at or post-closing upon resolution of certain Alfmeier tax matters. We believe cash on hand, cash generated from operations, and the borrowing capacity available under our revolving credit facility will be sufficient to support this transaction.

The Company continues to enter into agreements with suppliers to reserve the right to purchase certain semiconductor chips over periods of 12-24 months. As of March 31, 2022, the Company's total commitments for these semiconductor chip agreements was \$38.9 million. See Note 7, "Commitments and Contingencies" of the consolidated condensed financial statements included in this Report for additional information.

Except as described above, there have been no material changes in our cash requirements since December 31, 2021, the end of fiscal year 2021. See Part II, Item 7 "Management's Discussion and Analysis of Financial Condition and Results of Operations" in our Annual Report on Form 10-K for the year ended December 31, 2021 for additional information regarding our material cash requirements.

Effects of Inflation

The automotive component supply industry has historically been subject to inflationary pressures with respect to materials and labor. In 2021 and continuing in 2022, macroeconomic effects of the COVID-19 pandemic have resulted in inflationary cost increases in certain materials, labor and transportation. These inflationary cost increases are expected to continue into the foreseeable future as demand remains elevated and supply remains constrained. Although the Company has developed and implemented strategies to mitigate the impact of higher material component costs and transportation costs, these strategies, together with commercial negotiations with Gentherm's customers and suppliers may not fully offset our future cost increases. Such inflationary cost increase may increase the cash required to fund our operations by a material amount.

Critical Accounting Policies

The discussion and analysis of our financial condition and results of operations are based upon our consolidated condensed financial statements, which have been prepared in accordance with accounting principles generally accepted in the United States of America. For discussion of our significant accounting policies, see Note 2, "Summary of Significant Accounting Policies," to the consolidated financial statements included in our Annual Report on Form 10-K for the year ended December 31, 2021. There have been no significant changes in our critical accounting policies or critical accounting estimates during the three months ended March 31, 2022. We are not presently aware of any events or circumstances that would require us to update our estimates, assumptions or revise the carrying value of our assets or liabilities. Our estimates may change, however, as new events occur and additional information is obtained. As a result, actual results may differ significantly from our estimates, and any such differences may be

material to our financial statements. For information on the impact of recently issued accounting pronouncements, see Note 2, "New Accounting Pronouncements" in the consolidated condensed financial statements included in this Report.

Recent Accounting Pronouncements

For information on the impact of recently issued accounting pronouncements, see Note 2, "New Accounting Pronouncements" in the consolidated condensed financial statements included in this Report.

ITEM 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

Our exposure to market risk for changes in interest rates relates primarily to our debt obligations and foreign currency contracts. We have in the past, and may in the future, place our investments in bank certificates of deposits, debt instruments of the U.S. government, and in high-quality corporate issuers.

We are exposed to market risk from changes in foreign currency exchange rates, short-term interest rates and price fluctuations of certain material commodities such as copper. Market risks for changes in interest rates relate primarily to our debt obligations under our Credit Agreement. Foreign currency exchange risks are attributable to sales to foreign customers and purchases from foreign suppliers not denominated in a location's functional currency, foreign plant operations, intercompany indebtedness, intercompany investments and include exposures to the Euro, Mexican Peso, Canadian Dollar, Hungarian Forint, North Macedonian Denar, Ukrainian Hryvnia, Japanese Yen, Chinese Renminbi, Korean Won and Vietnamese Dong.

The Company regularly enters into derivative contracts with the objective of managing its financial and operational exposure arising from these risks by offsetting gains and losses on the underlying exposures with gains and losses on the financial instruments used to hedge them. The maximum length of time over which we hedge our exposure to foreign currency exchange risks is fifteen months. We had foreign currency derivative contracts with a notional value of \$21.2 million and \$14.0 million outstanding at March 31, 2022 and December 31, 2021 and copper commodity swap contracts with a notional value of \$0 million and \$0.3 million outstanding as of March 31, 2022 and December 31, 2021, respectively. The potential loss in fair value for such financial instruments from a hypothetical 10% adverse change in quoted currency exchange rates would be \$2.0 million and \$1.3 million as of March 31, 2022 and December 31, 2021, respectively. The potential gain in fair value from a hypothetical 10% change in quoted currency exchange rates would be \$2.5 million and \$1.6 million as of March 31, 2022 and December 31, 2021, respectively. The impact of a 10% change in rates on fair value differs from a 10% change in the net fair value asset due to the existence of hedges. The model assumes a parallel shift in currency exchange rates; however, currency exchange rates rarely move in the same direction. The assumption that currency exchange rates change in a parallel fashion may overstate the impact of changing currency exchange rates on assets and liabilities denominated in currencies other than the U.S. dollar.

We do not enter into derivative financial instruments for speculative or trading purposes. Our hedging relationships are formally documented at the inception of the hedge, and hedges must be highly effective in offsetting changes to future cash flows on hedged transactions both at the inception of a hedge and on an ongoing basis to be designated for hedge accounting treatment. For derivative contracts which can be classified as a cash flow hedge, the effective portion of the change in the fair value of the derivative is recorded to Accumulated other comprehensive loss in the consolidated condensed balance sheets. When the underlying hedge transaction is realized, the gain or loss included in accumulated other comprehensive loss is recorded in earnings in the consolidated condensed statements of income on the same line as the gain or loss on the hedged item attributable to the hedged risk. We record the ineffective portion of foreign currency and copper commodity hedging instruments, if any, to cost of sales in the consolidated condensed statements of income. Though we continuously monitor the hedging program, derivative positions and hedging strategies, foreign currency forward exchange agreements have not always been designated as hedging instruments for accounting purposes.

The Company uses an income approach to value derivative instruments, analyzing quoted market prices to calculate the forward values and then discounts such forward values to the present value using benchmark rates at commonly quoted intervals for the instrument's full term.

Information related to the fair values of all derivative instruments in our consolidated condensed balance sheet as of March 31, 2022 is set forth in Note 9, "Financial Instruments" in the consolidated condensed financial statements included in this Report.

Interest Rate Sensitivity

The table presents principal cash flows and related weighted average interest rates by expected maturity dates for each of the Company's debt obligations. The information is presented in U.S. dollar equivalents, which is the Company's reporting currency. The instruments actual cash flows are denominated in the currency indicated in parentheses.

	 Expected Maturity Date										
	2022 2023			2024		Total		air Value			
<u>Liabilities</u>									_		
Long-Term Debt:											
Variable rate (\$USD)	\$ _	\$	_	\$	35,000	\$	35,000	\$	35,000		
Variable interest rate as of March 31, 2022					1.71%)	1.71%				
Fixed rate (\$USD)	\$ 2,500	\$	1,250	\$	_	\$	3,750	\$	3,796		
Fixed interest rate	5.21%)	5.21%				5.21%				

Based on the amounts outstanding as of March 31, 2022, a hypothetical 100 basis point change (increase or decrease) in interest rates would impact annual interest expense by \$0.4 million.

Exchange Rate Sensitivity

The table below provides information about the Company's foreign currency forward exchange rate agreements that are sensitive to changes in foreign currency exchange rates. The table presents the notional amounts and weighted average exchange rates by expected (contractual) maturity dates for each type of foreign currency forward exchange agreement. These notional amounts generally are used to calculate the contractual payments to be exchanged under the contract.

		Expected	Matu	ate		
Anticipated Transactions and Related Derivatives	2022		Total		F	air Value
USD Functional Currency						
Forward Exchange Agreements:						
(Receive \$MXN / Pay \$USD)						
Total contract amount	\$	21,180	\$	21,180	\$	904
Average contract rate		21.25		21.25		

ITEM 4. CONTROLS AND PROCEDURES

(a) Evaluation of Disclosure Controls and Procedures

Management of the Company, under the supervision and with the participation of the Chief Executive Officer and the Chief Financial Officer, carried out an evaluation of the effectiveness of the design and operation of the Company's disclosure controls and procedures as of March 31, 2022. As defined in Rules 13a-15(e) and 15d-15(e) under the Securities Exchange Act of 1934 (as amended, the "Exchange Act"), disclosure controls and procedures are controls and procedures designed to provide reasonable assurance that information required to be disclosed in reports filed or submitted under the Exchange Act is recorded, processed, summarized and reported on a timely basis, and that such information is accumulated and communicated to management, including the Chief Executive Officer and Chief Financial Officer, as appropriate, to allow timely decisions regarding required disclosure. Based upon this evaluation, the Chief Executive Officer and Chief Financial Officer have concluded that the Company's disclosure controls and procedures were effective as of March 31, 2022.

(b) Changes in Internal Control over Financial Reporting

There were no changes in the Company's internal control over financial reporting during the three months ended March 31, 2022 that have materially affected, or are reasonably likely to materially affect, the Company's internal control over financial reporting.

PART II OTHER INFORMATION

ITEM 1. LEGAL PROCEEDINGS

We are subject to litigation from time to time in the ordinary course of business, however there is no material pending litigation to which we are a party and no material legal proceeding was terminated, settled or otherwise resolved during the three months ended March 31, 2022.

ITEM 1A. RISK FACTORS

Except as set forth below, the Company's risk factors have not materially changed from those previously disclosed in Part 1, Item 1A "Risk Factors" in our Annual Report on Form 10-K for the year ended December 31, 2021.

Our operations within Ukraine subject us to risks that may harm our operations and financial results.

The United States and global markets are experiencing volatility and disruption following the escalation of geopolitical tensions and the start of the military conflict between Russia and Ukraine. In February 2022, Russia launched a full-scale military invasion of Ukraine. As a result of the conflict, the United States, United Kingdom, European Union and other countries have levied economic sanctions and bans on Russia and Russia has responded with its own retaliatory measures. These measures have impacted the availability and price of certain raw materials and could have a lasting impact on regional and global economies. Although the length and impact of the ongoing military conflict is highly unpredictable, the conflict in Ukraine has led to and could lead to further market disruptions, including significant volatility in commodity prices, credit and capital markets, as well as supply chain interruptions.

Our facility in Vynohradiv is on the far western corner within the Transcarpathia region of Ukraine near the Hungary border. In 2021, products manufactured at our Ukraine facility represented approximately 11% of the Company's total revenue, including automotive cables, seat heaters and steering wheel heaters. At this time, our Ukraine facility is operating at normal levels and we have begun executing contingency plans and, in coordination with certain customers, specific equipment and production relocations leveraging our flexible global manufacturing footprint. Our response to the escalating situation is based on a severity level contingency response plan that has been developed with certain customers. As the situation in Ukraine is very fluid, we continue to monitor its effects on our business and we continue to work closely with our customers to adjust our contingency response as necessary.

We cannot be certain that the current conflict will not affect our facility and local employees in the future, including due to electrical outages, a loss of control over the assets due to destruction or theft, or periodic battles with foreign forces closer to our facility. We have incurred, and will likely continue to incur costs to support our employees and relocate equipment and production. Our management team has and will likely continue to spend significant time and attention on supporting our team members in Ukraine and mitigating the impacts of the Ukraine conflict and related events. Most of our products manufactured in Ukraine are shipped across the border from Ukraine to Hungary for further delivery to our customers. If that border crossing were to be closed or restricted for any reason, we may experience a significant disruption to our operations. The current conflict may lead to further disruption of our supply chains and operations, increased costs of labor, materials and components, result in impairment of tangible assets and implementation of restructuring activities, and may impair our ability to complete financial or banking transactions.

Any of the foregoing factors could have a material adverse effect on our business, financial condition, results of operations, cash flows and stock price. The extent and duration of the military action, sanctions, and resulting market and/or supply disruptions are impossible to predict, but could be substantial. Any such disruptions may also magnify the impact of other risks described in Part 1, Item 1A, "Risk Factors" in our Annual Report on Form 10-K for the year ended December 31, 2021.

Our proposed Acquisition of Alfmeier is subject to conditions, as well as other uncertainties, and there can be no assurances as to whether or when it may be completed. Failure to complete the proposed Acquisition could adversely affect our business.

Our proposed Acquisition of Alfmeier is expected to close during the third quarter of 2022; however, closing is subject to certain conditions that may not be satisfied or waived, including that a governmental entity may prohibit, delay or refuse to grant required antitrust approval. Events may occur that could give rise to termination of the Acquisition agreement, including due to factors outside of Gentherm's control. There may also be delays in closing the proposed Acquisition which could result in additional consultation and advisory fees necessary to finalize the Acquisition. Any of the foregoing, or other risks arising in connection with the

failure of or delay in completing the Acquisition, including the diversion of management's attention from pursuing other opportunities, could have a material adverse effect on our business, financial condition and results of operations.

The announcement and pendency of the proposed Acquisition of Alfmeier could adversely affect our business, financial condition and results of operations and could adversely affect Alfmeier's ability to retain and hire key personnel and maintain relationships with customers, suppliers and other third parties for the post-transaction benefit of Gentherm.

The announcement and pendency of the potential Acquisition of Alfmeier could cause disruptions and create uncertainty surrounding our business and affect our relationships with our customers, suppliers and employees. As a result of the Acquisition, some customers, suppliers or strategic partners may terminate their business relationship with us or Alfmeier. Potential customers, suppliers or strategic partners may delay entering into, or decide not to enter into, a business relationship with us or Alfmeier because of the Acquisition. If customer or supplier relationships or strategic alliances of us or Alfmeier are adversely affected by the Acquisition, the post-transaction benefit to Gentherm of the Acquisition of Alfmeier and our business, financial condition and results of operations following the Acquisition could be adversely affected.

In addition, we have diverted, and will continue to divert, significant management resources to complete the Acquisition and may incur unexpected costs during the pendency of the proposed Acquisition, which could adversely impact our ability to manage existing operations or pursue alternative strategic transactions, which could adversely affect our business, financial condition and results of operations.

Our efforts to integrate Alfmeier's business may not be successful and could adversely affect our production, financial condition and results of operations.

The success of the Company's acquisitions is dependent, in part, on its ability to realize the expected benefits from the integration of the acquired businesses and there are risks inherent in the achievement of expected financial results, growth prospects and cost synergies for the Acquisition and the timing thereof. To realize these anticipated benefits, acquired companies must be successfully integrated, which is subject to the Company's ability to consolidate operations, corporate cultures and systems and to eliminate redundancies and costs. If the Company is unsuccessful in combining companies, including Alfmeier, the anticipated benefits of our acquisitions, including the Acquisition of Alfmeier, may not be realized fully or at all or may take longer to realize than expected. Further, there is potential for unknown or inestimable liabilities relating to the acquired business. In addition, the actual integration may result in additional and unforeseen expenses, which could reduce the anticipated benefits of the acquisition.

The integration of acquired businesses is a complex, costly and time-consuming process that requires significant management attention and resources. It is possible that the integration process could result in the loss of key employees, the disruption of the Company's operations, the inability to maintain or increase its competitive presence, inconsistencies in standards, controls, procedures and policies, difficulties in achieving anticipated cost savings, synergies, business opportunities and growth prospects from the acquisition, the diversion of management's attention to integration matters and/or difficulties in the assimilation of employees and corporate cultures. Any or all of these factors and the Company's increased debt leverage following the closing of the Acquisition or other acquisitions could have an adverse effect on the Company. In addition, many of these factors are outside of the Company's control, and any one of these factors could result in increased costs, decreases in the amount of expected revenues and additional diversion of management's time and energy, which could materially adversely impact its business, financial condition and results of operations.

ITEM 2. UNREGISTERED SALES OF EQUITY SECURITIES AND USE OF PROCEEDS

Issuer Purchases of Equity Securities During First Quarter 2022

				(c)	(d)
				Total Number	Approximate
				of Shares	Dollar Value of
				Repurchased as	Shares That
	(a)			Part of Publicly	May Yet Be
	Total Number	(1	b)	Announced	Purchased
	of Shares	Averag	ge Price	Plans or	Under the Plans
Period	of Shares Purchased	-	ge Price er Share	Plans or Programs	Under the Plans or Programs (1)
Period January 1, 2022 to January 31, 2022		-	,		
		Paid Pe	,		or Programs (1)

(1) On December 11, 2020, the Board of Directors authorized a new stock repurchase program (the "2020 Stock Repurchase Program") to commence upon expiration of the prior stock repurchase program on December 15, 2020. Under the 2020 Stock Repurchase Program, the Company is authorized to repurchase up to \$150.0 million of its issued and outstanding common stock over a three-year period, expiring December 15, 2023. The authorization of this stock repurchase program does not require that the Company repurchase any specific dollar value or number of shares and may be modified, extended or terminated by the Company's Board of Directors at any time.

ITEM 5. OTHER INFORMATION

On May 4, 2022, Gentherm GmbH (the "Purchaser"), a limited liability company incorporated under the laws of Germany and a wholly owned subsidiary of Gentherm Incorporated (the "Company"), entered into a Share Purchase and Transfer Agreement (the "Acquisition Agreement"), by and among Gebhardt Holding GmbH, a limited liability company incorporated under the laws of Germany ("Gebhardt Holding"), ELBER GmbH, a limited liability company incorporated under the laws of Germany ("ELBER" and, together with Gebhardt Holding, the "Sellers"), and Andreas Gebhardt, Markus Gebhardt and Dr. Johann Vielberth (such three persons, the "Seller Guarantors"), pursuant to which the Purchaser has agreed to acquire all of the outstanding shares in Alfmeier Präzision SE, a European stock corporation incorporated under the laws of Germany and headquartered in Treuchtlingen, Germany. Alfmeier Präzision SE, together with its direct and indirect subsidiaries (collectively, "Alfmeier"), operate a worldwide automotive seating comfort and fluid systems business that generated €232 million in revenue for 2021 (the "Business"). Following the Acquisition (as defined below), Alfmeier will be an indirect wholly-owned subsidiary of the Company.

The Purchaser has agreed to acquire the Business through the purchase of all shares in Alfmeier pursuant to the terms and conditions of the Acquisition Agreement (the "Acquisition"), for a purchase price of \in 177.5 million (approximately \$187.6 million at current foreign exchange rates), net of cash and debt, subject to customary adjustments related to Alfmeier's net working capital as of the closing and may be increased by up to \in 4.0 million (approximately \$4.2 million at current foreign exchange rates) at or post-closing upon resolution of certain Alfmeier tax matters. The Acquisition will be funded through a combination of the Company's existing cash balances and revolving credit facility.

Prior to the closing of the Acquisition, Alfmeier and certain of its direct subsidiaries will complete a series of carve-out transactions such that Alfmeier's industrial and non-automotive business is not a part of the Business to be acquired by Purchaser.

Also on May 4, 2022, the Purchaser and Gebhardt Holding entered into the Put and Call Option Agreement COSMIQ, pursuant to which Gebhardt Holding will have a put to sell, and the Purchaser will have an option to purchase, all of the shares in COSMIQ Industrieverwaltungs- und Vermietungs-GmbH, a limited liability company incorporated under the laws of Germany ("COSMIQ") and the owner of the real property in Germany on which Alfmeier's headquarter is located.

The Acquisition Agreement is governed under German law and contains customary warranties and covenants by each party that are subject, in certain cases, to specified exceptions and qualifications contained in the Acquisition Agreement. The Acquisition Agreement also contains indemnification provisions under which the Sellers have agreed to indemnify the Purchaser against certain liabilities. The Sellers have agreed that Alfmeier, during the period from the signing of the Acquisition Agreement to the closing of the Acquisition, will conduct business in the ordinary course consistent with past practice and will not undertake certain measures or activities as set forth in the Purchase Agreement, unless consented to by the Purchaser under applicable law. The Purchaser has

obtained warranty and indemnification insurance as a potential initial source of recovery against the warranties and tax indemnity but in all cases subject to the terms and conditions of the insurance policy, including certain exclusions from coverage. The Seller Guarantors have agreed to act as joint and several debtors for certain of the Sellers' obligations under and in connection with the Acquisition Agreement.

The closing of the Acquisition is subject to customary closing conditions, merger control approval in Germany and North Macedonia, German Foreign Investment Control Clearance, and national security approval in the U.K. Subject to the receipt of regulatory approvals and satisfaction or waiver of the other closing conditions, the Acquisition is expected to close during the third quarter of 2022. If the Acquisition is not consummated prior to December 31, 2022, then the Purchaser, on one hand, and the Sellers, on the other hand, may withdraw from the Acquisition and terminate the Acquisition Agreement.

The foregoing description of the Acquisition Agreement and the Acquisition do not purport to be complete, and are qualified in their entirety by reference to the full text of the Acquisition Agreement, which is filed as Exhibit 2.1 to this Quarterly Report on Form 10-Q and incorporated herein by reference.

The Acquisition Agreement has been included to provide information regarding the terms thereof. The Acquisition Agreement contains warranties, indemnities and covenants that are the product of negotiations among the parties thereto and that the parties made to, and solely for the benefit of, each other as of specified dates. The assertions in those warranties, indemnities and covenants are subject to qualifications and limitations agreed to by the respective parties and are also qualified by confidential exhibits and schedules delivered in connection with the Acquisition Agreement. The warranties, indemnities and covenants were made for the purposes of allocating contractual risk between the parties to the Acquisition Agreement and should not be relied upon as disclosure of factual information relating to the Company, the Purchaser or Alfmeier.

ITEM 6. EXHIBITS

Exhibits to this Report are as follows:

				Incorporated	l by Reference	
Exhibit Number	Exhibit Description	Filed /Furnished Herewith	Form	Period Ending	Exhibit / Appendix Number	Filing Date
2.1*	Share Purchase and Transfer Agreement, dated May 4, 2022, by and among Gebhardt Holding GmbH, ELBER GmbH, Gentherm GmbH, and Andreas Gebhardt, Markus Gebhardt and Dr. Johann Vielberth.	X				
3.1	Second Amended and Restated Articles of Incorporation of Gentherm Incorporated		8-K		3.2	3/5/18
3.2	Amended and Restated Bylaws of Gentherm Incorporated		8-K		3.1	5/26/16
31.1	Section 302 Certification – CEO	X				
31.2	Section 302 Certification – CFO	X				
32.1**	Section 906 Certification – CEO	X				
32.2**	Section 906 Certification – CFO	X				
101.INS	Inline XBRL Instance Document – The instance document does not appear in the Interactive Data File because its XBRL tags are embedded within the Inline XBRL document.	X				
101.SCH	Inline XBRL Taxonomy Extension Schema Document.	X				
101.CAL	Inline XBRL Taxonomy Extension Calculation Linkbase Document.	X				
101.DEF	Inline XBRL Taxonomy Extension Definition Linkbase Document.	X				
101.LAB	Inline XBRL Taxonomy Extension Label Linkbase Document.	X				
101.PRE	Inline XBRL Taxonomy Extension Presentation Linkbase Document.	X				
104	Cover Page Interactive Data File (embedded within the Inline XBRL document contained in Exhibit 101)	X				

^{*} Schedules and exhibits to this agreement have been omitted pursuant to Item 601(a)(5) of Regulation S-K. The Company agrees to furnish any omitted schedules or exhibits upon the request of the SEC.

^{**} Documents are furnished not filed.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

Gentherm Incorporated

/s/ PHILLIP EYLER

Phillip Eyler
President and Chief Executive Officer
(Principal Executive Officer)

Date: May 4, 2022

/s/ MATTEO ANVERSA

Matteo Anversa

Executive Vice President, Chief Financial Officer and Treasurer (Principal Financial Officer)

Date: May 4, 2022

37

Project Astute Share Purchase and Transfer Agreement

dated

3 May 2022

TABLE OF CONTENTS

Preamble		5
I.	INTERPRETATION; DEFINITIONS AND EXHIBITS	6
1.	Rules of Interpretation	6
2.	Definitions and Exhibits	8
II.	CURRENT STATUS	8
3.	Current Status of the Company	8
4.	Financing	8
ш.	SALE AND TRANSFER	10
5.	Sale and Transfer of the Shares	10
6.	Approvals	10
IV.	PURCHASE PRICE; PAYMENTS; PURCHASER'S FINANCING	11
7.	Purchase Price	11
8.	Determination of Company Purchase Price / Effective Date Adjustment Statement	14
9.	Calculation and Due Date of Adjustment Amount	17
10.	General Rules for Payments; Bank Accounts	18
11.	Escrow	18
12.	Purchaser's Financing	20
13.	VAT	21
V.	SEPARATION FROM THE SELLER GROUP	21
14.	Pre-Closing Carve-out	21
VI.	TERMINATION OF EXTERNAL FINANCING AGREEMENTS	25
15.	Repayment of External Financing	25
VII.	FURTHER PRE-CLOSING COVENANTS	29
16.	Obtaining of Governmental Clearances/Approvals	29
17.	Sellers' Conduct of Business Covenants	31
18.	Cooperation by Sellers prior to Closing; Information Rights of Purchaser	38
VIII.	CLOSING	38
19.	Closing Conditions	38
20.	Closing	40

IX.	WARRANTIES	42
21.	Warranties of Sellers	42
22.	Warranties of Purchaser	43
Χ.	PURCHASER'S REMEDIES; CLAIM PROCEDURES; SELLERS' REMEDIES	44
23.	Purchaser's Remedies	44
24.	Claim Procedures	46
XI.	TAXES	48
25.	Taxes	48
XII.	INDEMNITIES	53
26.	Indemnities	53
XIII.	LIMITATIONS	55
27.	Limitation of Claims	55
28.	Intentional or Fraudulent Behavior; Limitations	57
XIV.	POST-CLOSING COVENANTS	58
29.	Business Records; Post-Closing Assistance; Required Financial Statements	58
30.	Non-Compete; Non-Solicitation	60
31.	Non-Disparagement	61
XV.	MISCELLANEOUS	62
32.	Withdrawal from this Agreement	62
33.	Liability of Sellers	62
34.	Obligations of Seller Guarantors	63
35.	Confidentiality	64
36.	Notices	65
37.	Public Announcement	68
38.	SAES Getters, Reimbursement of Duty Drawback	69
39.	Costs and Expenses; Transfer Taxes	69
40.	General Provisions	70
		3/78

ANNEX	72	
1.	Definitions	72
2.	List of Definitions	77
ANNEX	X 2.2 (LIST OF EXHIBITS)	80
		4/78

Share Purchase and Transfer Agreement

betweer

- (1) **Gebhardt Holding GmbH**, a limited liability company (*Gesellschaft mit beschränkter Haftung*) incorporated under the laws of Germany, having its registered seat in Treuchtlingen, registered with the commercial register of the local court of Ansbach under no. HRB 3460, with business address at Industriestraße 5, 91757 Treuchtlingen ("**Gebhardt Holding**");
- (2) **ELBER GmbH**, a limited liability company (*Gesellschaft mit beschränkter Haftung*) incorporated under the laws of Germany, having its registered seat in Regensburg, registered with the commercial register of the local court of Regensburg under no. HRB 12769, with business address at Im Gewerbepark C 25, 93059 Regensburg ("**ELBER**");
- (3) **Gentherm GmbH**, a limited liability company (*Gesellschaft mit beschränkter Haftung*) incorporated under the laws of Germany, having its registered seat in Odelzhausen, registered with the commercial register of the local court of Munich under no. HRB 208876, with business address at Rudolf-Diesel-Straße 12, 85235 Odelzhausen ("**Purchaser**");
- (4) Andreas Gebhardt, born 1 July 1966, with business address at Industriestraße 5, 91757 Treuchtlingen ("Andreas Gebhardt");
- (5) Markus Gebhardt, born 1 July 1966, with business address at Industriestraße 5, 91757 Treuchtlingen ("Markus Gebhardt"); and
- (6) **Dr. Johann Vielberth**, born 1 April 1932, with business address at Im Gewerbepark C 25, 93059 Regensburg ("**Johann Vielberth**").

Gebhardt Holding and ELBER are each individually also referred to as a "Seller" and collectively as the "Sellers". Each of the Sellers and Purchaser are individually also referred to as a "Party" and collectively as the "Parties". Andreas Gebhardt, Markus Gebhardt and Johann Vielberth are each individually also referred to as a "Seller Guarantor" and collectively as the "Seller Guarantors".

Preamble

(A) Sellers are the sole shareholders of Alfmeier Präzision SE, a European stock corporation (*Societas Europaea*) incorporated under the laws of Germany, having its registered seat in Treuchtlingen, registered with the commercial register of the local court of Ansbach under HRB 6249 and with business address at Industriestraße 5, 91757 Treuchtlingen (the "Company"). The Company and its Subsidiaries (collectively the "Alfmeier Target Group" or "Target Group" and each an "Alfmeier Target Group Company" or "Target Group Company") operate a worldwide seating comfort and fluid systems business, in particular by developing and manufacturing precision components such as valves, actuators, pumps, bladders and controllers for the

automotive industry (the business as conducted by the Alfmeier Target Group on the Signing Date, the "Business").

- (B) Purchaser is a German subsidiary of Gentherm Incorporated ("Parent"), which is a developer and marketer of thermal management technologies for heating, cooling and temperature control devices for a variety of industries including the automotive and medical industries, and based in the United States and operating worldwide.
- (C) (i) On 29 April 2022, the Company and its direct subsidiary AW Holding GmbH, a limited liability company (Gesellschaft mit beschränkter Haftung) incorporated under the laws of Germany, having its registered seat in Treuchtlingen, registered with the commercial register of the local court of Ansbach under HRB 5488 and with business address at Industriestraße 5, 91757 Treuchtlingen ("AW Holding"), have entered into an asset purchase agreement, under which certain assets pertaining to the so-called "Sentire Business" have been transferred to AW Holding, and (ii) on 22 March 2022, the Company and its direct subsidiary SMA Holding GmbH, a limited liability company (Gesellschaft mit beschränkter Haftung) incorporated under the laws of Germany, having its registered seat in Treuchtlingen, registered with the commercial register of the local court of Ansbach under HRB 5220 and with business address at Industriestraße 5, 91757 Treuchtlingen ("SMA Holding"), have entered into a share purchase and transfer agreement, under which all shares in AW Holding have been transferred to SMA Holding (roll of deeds no. 468/2022 S of the notary public Dr. Heike Stiebitz in Gunzenhausen) (the transactions described in this Preamble (C) collectively the "Pre-Signing Carve-out"). Copies of the Pre-Signing Carve-out agreements are attached hereto as Exhibit P-(C) for information purposes only.
- (D) Sellers now intend to sell and transfer all shares in the Company to Purchaser and Purchaser intends to acquire all shares in the Company, in each case in accordance with the terms and conditions of this Agreement (the "Transaction"). Sellers Guarantors intend to act as joint and several debtors only with the relevant Seller controlled by them for certain of the Sellers' obligations under and in connection with this Agreement and as further specified in this Agreement.

NOW, THEREFORE, the Parties, the Company and the Seller Guarantors hereby agree as follows (the "Agreement"):

I. Interpretation; Definitions and Exhibits

1. Rules of Interpretation

Unless expressly provided otherwise herein, for purposes of this Agreement, the following rules of interpretation shall apply:

- 1.1 The headings and sub-headings in this Agreement are inserted for convenience only and shall not affect the interpretation or construction of this Agreement.
- 1.2 Unless specified otherwise, any reference in this Agreement to any "Section" is to the corresponding section of this Agreement and any reference in this Agreement to any "Annex" or "Exhibit" is to the corresponding annex or exhibit of this Agreement.
- 1.3 The Annexes and Exhibits shall be construed as an integral part of this Agreement. Any reference to this Agreement includes this Agreement and the Annexes and Exhibits as a whole.
- 1.4 If any person or entity shall, pursuant to this Agreement, use "**commercially reasonable efforts**", such person/entity shall be required to make a diligent, reasonable and good faith effort to accomplish the applicable objective. Such obligation, however, does not require an expenditure of material funds or the incurrence of a material liability, nor does it require to make any concession or acts in a manner that would be contrary to normal commercial practices under the given circumstances in order to accomplish the objective. The fact that the objective is or is not actually accomplished is no indication that the person/entity did, or did not, in fact use its reasonable commercial efforts in attempting to accomplish the objective.
- 1.5 If this Agreement, by using the term "procure", sets forth that a Party shall arrange for any action, omission, status, result, or other objective, this shall mean that such Party shall be responsible and liable for any failure of such objective irrespective of fault (*verschuldensunabhängige Sicherstellungs- und Einstandspflicht*). If this Agreement provides that Sellers shall procure that the Company takes or omits certain actions, this shall mean that Sellers are, to the extent permitted under German stock corporation law, obliged to exercise their influence on the Company to cause the Company to take or omit the respective action. Vis-à-vis Purchaser, Sellers shall remain liable according to the principle set forth in the first sentence of this Section 1.5.
- 1.6 Words such as "hereof", "herein" or "hereunder" refer (unless otherwise required by the context) to this Agreement as a whole and not to a specific provision of this Agreement.
- 1.7 The word "including", "in particular" or any variation shall be deemed to be followed by the words "without limitation" and shall not be construed to limit a general statement to the specific exemplary items or matters following it.
- 1.8 Any reference in this Agreement to a gender shall include all genders. Words (including definitions) imparting the singular number only shall include the plural and *vice versa*.
- 1.9 Where a German term has been inserted in italics, it alone (and not the English term to which it relates) shall be authoritative for the purpose of the interpretation of the relevant English term in this Agreement. The relevant English term shall be interpreted the same way throughout the Agreement, unless the context requires otherwise.

- 1.10 Wherever in this Agreement the term "material" or "materiality" or any variation thereof appears, no monetary threshold set forth in any provision of this Agreement shall be considered in connection with the interpretation thereof unless specifically and explicitly tied thereto.
- 1.11 Any monetary thresholds denominated in EUR shall include the equivalent in any foreign currency calculated based on the Exchange Rate as of the Signing Date.
- 1.12 Except where otherwise specifically set forth herein, references to a specific time herein are references to local time in Munich, Germany.

2. Definitions and Exhibits

- 2.1 Annex 2.1 sets out, for purposes of this Agreement, (i) certain defined terms and (ii) a list of all defined terms used in this Agreement.
- 2.2 <u>Annex 2.2</u> sets forth a list of the Exhibits to this Agreement.

II. Current Status

3. Current Status of the Company

- 3.1 The Company currently has an aggregate registered share capital (*Grundkapital*) of EUR 4,418,646.29 (in words: four million four hundred eighteen thousand six hundred forty-six Euro and twenty-nine Cents), divided into 1,728,082 (in words: one million seven hundred twenty-eight thousand eighty-two) registered shares (*Namensaktien*) without nominal value (*Nennbetrag*). Gebhardt Holding holds 1,048,600 (in words: one million forty-eight thousand six hundred) shares in the Company with the consecutive numbers 1 to 977,200 and 1,048,601 to 1,120,000 (the "Gebhardt Shares"), whereas ELBER holds 679,482 (in words: six hundred seventy-nine thousand four hundred eighty-two) shares in the Company with the consecutive numbers 977,201 to 1,048,600 and 1,120,001 to 1,728,082 (the "ELBER Shares" and collectively with the Gebhardt Shares, the "Company Shares" or the "Shares").
- 3.2 The Company holds, as of the date hereof, directly or indirectly, as the case may be, all or majority shares and interests in the legal entities as set out in Exhibit 3.2 (the "Subsidiaries") as well as all shares in SMA Holding and in RKT (as defined below).

4. Financing

4.1 The Company as borrower and RKT Rodinger Kunststoff-Technik GmbH ("**RKT**"), KITE Electronics GmbH and k3 Works GmbH as joint and several debtors (together with the Company the "**Obligors**") are party to an originally EUR 30,000,000 syndicated loan agreement originally dated 1 August 2017 with Commerzbank Aktiengesellschaft, Unicredit Bank AG and Bank für Tirol und Vorarlberg Aktiengesellschaft, Zweigniederlassung as lenders (the "**Lenders**"),

Commerzbank Aktiengesellschaft and Unicredit Bank AG as arrangers and bookrunners and Commerzbank Aktiengesellschaft as facility agent (the "Facility Agent") and security agent (the "Security Agent") (as subsequently amended and restated from time to time, including the 2022 Amendment (as defined below), the "Facilities Agreement" and the facilities made available thereunder the "Facilities"). On 14 April 2022, the parties to the Facilities Agreement have entered into an amendment and restatement agreement regarding, *inter alia*, (i) the extension of the term of the Facilities, (ii) adjustments to the financial covenants to be complied with, (iii) the reduction of one of the existing term facilities, (iv) the implementation of a new term facility on the basis of the previous accordion facility for the purposes of refinancing the 2017 Promissory Notes, (v) implementation of a new accordion facility, and (vi) implementation of an increase option for the revolving facility, a copy of which is attached as Exhibit 4.1 (the "2022 Amendment").

- 4.2 The Obligors and Alfmeier CZ have granted security interests over their assets respectively a guarantee in respect of the liabilities and obligations of the Obligors under the Facilities as set forth in Exhibit 4.2 (collectively, the "Financing Collateral").
- 4.3 The Company is borrower under (i) a EUR 7,000,000 promissory note dated 14 July 2017, due on 30 June 2022, (ii) a EUR 4,000,000 promissory note dated 14 July 2017, due on 14 July 2022, (iii) a EUR 3,000,000 promissory note dated 14 July 2017, due on 14 July 2022 (the promissory notes listed in (i) through (iii) together the "2017 Promissory Notes") and (iv) a EUR 10,000,000 promissory note dated 22 August 2019, due on 31 July 2024, (the "2019 Promissory Notes") each with Joh. Berenberg, Gossler & Co. KG as original lender and paying agent (the "Paying Agent").
- The Company as receiving entity and RKT as transferring entity are parties to the profit and loss transfer agreement (*Ergebnisabführungsvertrag*) dated 17 October 2016, a copy of which is attached as <u>Exhibit 4.4</u> (the "**PLTA**").
- 4.5 The Company as cash pool leader and, *inter alia*, RKT as cash pool participant are parties to a cash pool agreement dated 26 November 2012, a copy of which is attached as <u>Exhibit 4.5</u> (the "Cash Pool").
- 4.6 One or more of the Target Group Companies as borrower or debtor has entered into the various loan and financing agreements in the United States, the Czech Republic and China as set forth in Part I of Exhibit 4.6-A (as subsequently amended and restated from time to time, the "Other Reported Debt Agreements" and the Debt thereunder the "Other Reported Debt"). Certain of the Target Group Companies have granted security interests over their respective assets in respect of its liabilities and obligations under the Other Reported Debt as set forth in Part I of Exhibit 4.6-B (collectively, the "Other Reported Debt Collateral").

4.7 The Company has granted certain security interest, guarantees and sureties (*Bürgschaften*) over its assets in respect of liabilities from Actuator Solutions GmbH and RKT as set forth in <u>Exhibit 4.7</u> (collectively, the "ASG/RKT Collateral").

III. Sale and Transfer

5. Sale and Transfer of the Shares

- 5.1 Subject to the terms and conditions set forth herein, (i) Gebhardt Holding hereby sells to Purchaser, and Purchaser hereby purchases from Gebhardt Holding, the Gebhardt Shares, and (ii) ELBER hereby sells to Purchaser, and Purchaser hereby purchases from ELBER, the ELBER Shares.
- The sale and purchase of the Shares shall include all ancillary rights (*Nebenrechte*) attached to the Shares, including all claims, rights and obligations pertaining thereto as of the last calendar day, 24:00 hours (end of day) of the calendar month in which Closing has occurred (the "**Effective Date**"), including the right to any undistributed profits. In the event that the Closing Date is the last calendar day of a quarter (*i.e.* 30 June 2022, 30 September 2022 or 31 December 2022), the Effective Date shall be 00:00 hours (beginning of the day) of the day following the end of the calendar month in which Closing has occurred (*i.e.* 1 July 2022, 1 October 2022 or 1 January 2023).
- 5.3 Gebhardt Holding hereby transfers (*tritt ab*) the Gebhardt Shares to Purchaser who accepts such transfer (*Abtretung*). ELBER hereby transfers (*tritt ab*) the ELBER Shares to Purchaser who accepts such transfer (*Abtretung*).
- Each of the aforementioned transfers is subject to the condition precedent pursuant to Section 158 (1) BGB (aufschiebende Bedingung) of the fulfilment or due waiver of all of the Closing Actions set forth in Section 20.2.

6. Approvals

- Pursuant to section 5.3 of the articles of association of the Company, the transfer of the Company Shares requires the approval of the Company, represented by the board of directors (*Verwaltungsrat*), and the approval of the general meeting. By resolution dated 2 May 2022, the general meeting of the Company has approved the sale and transfer of the Company Shares in accordance with this Agreement. A copy of the minutes of the general meeting is attached hereto as Exhibit 6.1-A. A copy of the approval of the board of directors is attached hereto as Exhibit 6.1-B.
- By shareholder resolution dated 28 April 2022, a copy of which is attached hereto as <u>Exhibit 6.2</u>, the shareholders' meeting of Gebhardt Holding has approved the sale and transfer of the Gebhardt Shares in accordance with this Agreement.

- By shareholder resolution dated 27 April 2022, a copy of which is attached hereto as Exhibit 6.3, the shareholders' meeting of ELBER has approved the sale and transfer of the ELBER Shares in accordance with this Agreement.
- Andreas Gebhardt and Markus Gebhardt each hereby confirm that they are legally divorced and not married and that they have not agreed to any further approval requirements restricting the entering into the Transaction by Gebhardt Holding. Dr. Johann Vielberth hereby confirms that he is not married in the statutory marital status of accrued gains (nicht im gesetzlichen Güterstand der Zugewinngemeinschaft verheiratet) and that he has not agreed to any further approval requirements restricting the entering into the Transaction by ELBER.

IV. Purchase Price; Payments; Purchaser's Financing

7. Purchase Price

7.1 Company Purchase Price

- 7.1.1 The purchase price to be paid as consideration of the sale and transfer of the Shares (the "Company Purchase Price" or the "Purchase Price") shall be equal to
 - a) a fixed amount of EUR 177,500,000 (in words: Euro one hundred seventy-seven million five hundred thousand);
 - b) <u>plus</u> the Carve-out Purchase Price;
 - c) plus the Effective Date Cash Company;
 - d) <u>minus</u> the Effective Date Financial Debt Company;
 - e) <u>plus</u> the amount by which the Effective Date Working Capital Company exceeds the Target Working Capital Company or minus the amount by which the Effective Date Working Capital Company falls short of the Target Working Capital Company;

A sample calculation of the Company Purchase Price based on the Reference Pro-Forma Balance Sheet and the applicable adjustments set forth in <u>Exhibit 7.1.2</u> is attached hereto as <u>Exhibit 7.1.1</u>.

7.1.2 For the purpose of this Agreement

- a) "Effective Date Cash Company" means the consolidated amount of the line items of the Alfmeier Target Group Companies set forth in Exhibit 7.1.2 under the heading "Effective Date Cash Company" in Euro as per the Effective Date, as determined on the basis of the Final Effective Date Adjustment Statement.
- b) "Effective Date Financial Debt Company" means the consolidated amount of the line items of the Alfmeier Target Group Companies set forth in Exhibit 7.1.2 under the heading

- "Effective Date Financial Debt Company" in Euro as per the Effective Date, as determined on the basis of the Final Effective Date Adjustment Statement.
- c) "Effective Date Working Capital Company" means the consolidated amount of the line items of the Alfmeier Target Group Companies set forth in Exhibit 7.1.2 under the heading "Effective Date Working Capital" in Euro as per the Effective Date, as determined on the basis of the Final Effective Date Adjustment Statement.
- d) The "**Target Working Capital Company**" shall be an amount of EUR 46,832,689 (in words: forty-six million eight hundred thirty-two thousand six hundred eighty-nine Euro).

7.2 Preliminary Purchase Price Company

- 7.2.1 As preliminary consideration for the sale and transfer of the Company Shares, Purchaser shall, on the Scheduled Closing Date, pay a total amount equal to EUR 111,249,000 (in words: Euro one hundred eleven million two hundred forty-nine thousand) (the "**Preliminary Purchase Price Company**").
- 7.2.2 A part of the Preliminary Purchase Price Company in an amount of EUR 11,844,000 (in words: Euro eleven million eight hundred forty-four thousand) (the "Carve-out Purchase Price") will not be settled by way of cash payment, but by way of assumption with debt-discharging effect of the purchase price claims of the Company against Sellers under the RKT SPA and the SMA Holding SPA (each as defined below) by Purchaser, in case of the RKT SPA as assumed by way of assumption agreement with debt-discharging effect by Sellers from SMA Holding, to be entered into on or before the Scheduled Closing Date substantially in the form attached hereto as Exhibit 7.2.2-A (the "Assumption Agreements Carve-out"). To effect the assumption, Sellers and Purchaser shall, on the Scheduled Closing Date, enter into an assumption agreement (Schuldübernahmevereinbarung) substantially in the form as attached hereto as Exhibit 7.2.2-B (the "Assumption Agreement") and notify the Company of the conclusion of the Assumption Agreement. From the Closing Date, Purchaser shall indemnify Sellers from any claims raised by or in the name of the Company as well as by any third party (such as an insolvency administrator) on its behalf in connection with the Assumption Agreement.
- 7.2.3 A part of the Preliminary Purchase Price Company in an amount of EUR 1,500,000 (in words: Euro one million five hundred thousand) shall not be paid out on the Scheduled Closing Date but shall be held back by Purchaser (the "Hold Back Amount"). The Hold Back Amount becomes payable by Purchaser to Sellers upon the expiry of six months after Closing (the "Hold Back Period"), provided however, that any claims of Thomas Konetznik and/or Stephan Schulak in connection with (i) the Transaction, (ii) any change of control clause in each of their agreements with the Company, (iii) any payment claims after the termination of the respective employment agreement (except for ordinary payment claims arising out of the respective employment or service relationship, as the case may be), and/or (iv) any severance payment on the basis of the termination agreement entered into in accordance with Section 17.2.1e) (the "Employment Claims") shall primarily be satisfied from the Hold Back Amounts and shall reduce the obligation

of Purchaser to pay the Hold Back Amount when due. Should Stephan Schulak fully and unconditionally waive any and all of his Employment Claims, in particular by entering into the termination agreement as set out in Section 17.2.1c), parts of the Hold Back Amount in the amount of EUR 1,000,000 (in words: Euro one million) less any severance payment to Stephan Schulak becomes payable before the expiry of the Hold Back Period to Sellers within ten (10) Business Days after the waiver. For the avoidance of doubt: any claim for severance payment shall not be considered as Effective Date Financial Debt Company. Should Thomas Konetznik fully and unconditionally waive any and all of his Employment Claims, in particular by entering into a retention agreement with the Company and/or the Purchaser, parts of the Hold Back Amount in the amount of EUR 500,000 (in words: Euro five hundred thousand) becomes payable before the expiry of the Hold Back Period to Sellers within ten (10) Business Days after the waiver, however, in each case of this and the preceding sentence not before the Closing Date.

- 7.2.4 The remaining part of the Preliminary Purchase Price Company, being the Preliminary Purchase Price Company minus the Carve-out Purchase Price and minus the Hold Back Amount, if any, shall be paid by the Purchaser on the Scheduled Closing Date as follows:
 - a) an amount of EUR 67,505,893 (in words: Euro sixty-seven million five hundred and five thousand eight hundred and ninety-three) (the "**Preliminary Gebhardt Amount**") shall be paid to the account for Gebhardt Holding set forth on Exhibit 10.2;
 - b) an amount of EUR 43,743,107 (in words: Euro forty-three million seven hundred forty-three thousand one hundred and seven) (the "**Preliminary ELBER Amount**") shall be paid to the account for ELBER set forth on Exhibit 10.2;
 - c) an amount of EUR 1,775,000 (in words: Euro one million seven hundred seventy-five thousand) (the "Escrow Amount Company") shall be paid to the Escrow Account; and
 - d) an amount of EUR 1,500,000 (in words: Euro one million five hundred thousand) (the "Escrow Amount Tax" and together with the Escrow Amount Company the "Escrow Amounts") shall be paid to the Escrow Account.

7.3 Mexico VAT Reinstatement Certification and Remediation

7.3.1 Sellers shall use all commercially reasonable efforts to effect (i) that AF&R Mexico is provided with a certificate fully reinstating its former VAT status for tax purposes (the "VAT Reinstatement") until the date that AF&R Mexico receives a definitive determination from the Mexican Taxing Authorities regarding the reinstatement of its former VAT status for Tax purposes (the "VAT Determination Date") and (ii) the self-correction and other remediation measures with the respect to AF&R Mexico's VAT/IMMEX/PROSEC compliance identified by the Company's VAT consultant engaged as of the Signing Date, including, as necessary, effecting amendments to AF&R Mexico's export pedimentos and the implementation of the recommending compliance policies and procedures.

- 7.3.2 Subject to a full and unconditional implementation of the VAT Reinstatement by 11 October 2022, Purchaser shall pay an additional amount of EUR 4,000,000 (in words: Euro four million) to Sellers (in the relevant amount the "VAT Reinstatement Compensation"), split between Sellers in accordance with Section 9.1. In case of a full and unconditional implementation from 12 October 2022 until and including 11 July 2024, the VAT Reinstatement Compensation shall amount to EUR 3,000,000 (in words: Euro three million), split between Sellers in accordance with Section 9.1. For the avoidance of doubt, the VAT Reinstatement Compensation for any VAT Reinstatement from 12 July 2024 onwards, shall be EUR 0 (in words: Euro zero).
- 7.3.3 The VAT Reinstatement Compensation, if any, shall become due on the 10th Business Day after the VAT Reinstatement has taken place in full, but in any case not before the Closing Date.
- 8. Determination of Company Purchase Price / Effective Date Adjustment Statement

8.1 Preparation of the Effective Date Adjustment Statement

- 8.1.1 After the Effective Date, Purchaser shall set up (i) a consolidated balance sheet of the Alfmeier Target Group, in each case according to IFRS and as of the Effective Date, showing all balance sheet items required for the determination of the Effective Date Cash Company, the Effective Date Financial Debt Company and the Effective Date Working Capital Company (the "Effective Date Balance Sheet") and (ii) a statement setting forth the Effective Date Cash Company, the Effective Date Financial Debt Company and the Effective Date Working Capital Company based on the Effective Date Balance Sheet and the other adjustments described in Exhibit 8.3.2 for the determination of the Effective Date Cash Company, the Effective Date Financial Debt Company and the Effective Date Working Capital Company (together with the Effective Date Balance Sheet, collectively, the "Effective Date Adjustment Statement").
- 8.1.2 The Effective Date Balance Sheet shall be prepared on a going concern basis in the same format as the Consolidated Financial Statements (the "Reference Pro-Forma Balance Sheet"), applying the applicable specific procedures that would be adopted at a financial year-end, including detailed analyses of accruals, cut-off procedures, performing account reconciliations and adjusting general ledger balances accordingly. The Effective Date Adjustment Statement (including the Effective Date Balance Sheet) shall be prepared in the English language as if the respective business years were to end for accounting and tax purposes on the Effective Date,
 - a) applying the adjustments, accounting policies, principles, procedures, categorizations and consolidation principles set forth in Exhibit 8.1.2a) and the specific adjustments set forth in Exhibit 7.1.2b), which shall in any event prevail over lit. b) and lit. c) below;
 - b) in accordance with the principles, policies, and procedures applied in the preparation of the Consolidated Financial Statements as of 31 December 2021, which shall in any event prevail over lit. c) below; and
 - c) in accordance with IFRS.

For the avoidance of doubt, in the event of conflict, lit. a) will prevail over lit. b) and lit. c), and lit. b) will prevail over lit. c). Amounts in foreign currency shall be converted into EUR with the applicable Exchange Rate as of the Effective Date.

- 8.1.3 For the avoidance of doubt, any loss compensation amounts in connection with a loss transfer claim (*Verlustausgleichsanspruch*) of RKT under the PLTA shall be treated as financial debt as of the Effective Date and such amounts may not be challenged by the Sellers under Sections 8.2 and 8.3. For the avoidance of doubt, since the Company waives any profit transfer claim (*Gewinnabführungsanspruch*) in connection with the PLTA under the RKT SPA, any such profit transfer claim shall not be treated as cash as of the Effective Date.
- 8.1.4 Purchaser shall deliver the Effective Date Adjustment Statement together with a convenience calculation of the Company Purchase Price and the Adjustment Amount derived therefrom to Sellers within ninety (90) calendar days after the Effective Date.

8.2 Review by Sellers

- 8.2.1 To the extent Sellers believe that the Effective Date Adjustment Statement has not been prepared in accordance with this Agreement, Sellers may, collectively and by notice made by Gebhardt Holding, raise objections to the Effective Date Adjustment Statement within thirty (30) calendar days after receipt of the Effective Date Adjustment Statement by providing Purchaser with a statement of objections specifying the relevant positions, the reasons for each objection and the amounts in dispute in detail.
- 8.2.2 If and to the extent Sellers do not raise any objections in accordance with Section 8.2.1, Sellers shall be deemed to have agreed to the positions set forth in the Effective Date Adjustment Statement which shall become final and binding on such positions upon the expiry of the time period set forth in Section 8.2.1.

8.3 Dispute Resolution

- 8.3.1 If and to the extent Sellers have raised objections to the Effective Date Adjustment Statement in accordance with Section 8.2, the Parties shall discuss the disputed items in order to reach an agreement. To the extent the Parties reach such agreement, its results shall be incorporated into the Effective Date Adjustment Statement, which in this regard shall be binding on the Parties for the purpose of calculating the Company Purchase Price.
- 8.3.2 To the extent the Parties are unable to reach such an agreement within thirty (30) calendar days after expiry of the period set forth in Section 8.2.1 on any disputed items, Purchaser, on one hand, and Sellers, on the other hand, may, within five (5) Business Days, request that the items still in dispute between the Parties shall be decided by either KPMG or Deloitte (as may be decided by Purchaser in its sole discretion) which shall act as a neutral expert. If KPMG and Deloitte refuse to act as a neutral expert, the Parties shall agree (in the form of Section 36.1) on an alternative independent accounting firm of international standing to act as a neutral expert. If the Parties

cannot agree on an alternative accounting firm within five (5) Business Days following the later of KPMG and Deloitte's refusal, an independent accounting firm of international standing shall be selected upon written request of Purchaser, on one hand, and Sellers, on the other hand, by the Institute of Chartered Accountants (*Institut der Wirtschaftsprüfer e.V.*) in Düsseldorf, Germany (KPMG, Deloitte or the accounting firm finally determined as a neutral expert pursuant to this Section 8.3.2 to be referred to as the "Neutral Expert"). If and to the extent the Parties have not reached an agreement pursuant to Section 8.3.1 and neither Purchaser nor either Seller requests that the matter in dispute be decided by the Neutral Expert in accordance with and within the time limit set forth in sentence 1, Section 8.2.2 shall apply *mutatis mutandis*.

8.3.3 Sellers and Purchaser shall jointly engage the Neutral Expert to decide the items in dispute in accordance with the provisions of this Agreement.

8.3.4 The Neutral Expert shall

- a) act as an expert (*Schiedsgutachter*), not as an arbitrator, and shall decide by way of a written expert's opinion (*Schiedsgutachten*) substantiating the decision;
- b) decide only on the items in dispute between the Parties;
- c) apply the accounting and valuation standards and principles referred to in Section 8.1.2, shall interpret this Agreement only to the extent necessary in order to decide the items in dispute, and not decide upon legal issues (unless such legal issues specifically pertain to the applicable accounting and valuation standards and principles);
- d) give each Party adequate opportunity to present their point of view in writing, shall share such arguments with the respective other Party, and shall hold at least one (1) hearing in the presence of the Parties and their professional advisors and accountants;
- e) not pass a decision beyond or outside the positions taken by the Parties; and
- f) decide upon the allocation of its costs and expenses between the Parties by applying the principles of Sections 91 *et seq.* ZPO.

8.3.5 The Parties shall

- a) instruct the Neutral Expert to use its best efforts to deliver its written expert's opinion to them as soon as reasonably practicable, but no later than six (6) weeks after the items in dispute have been referred to, and the Parties have properly instructed, the Neutral Expert; and
- b) provide any information and other support which the Neutral Expert deems required in connection with the preparation of its written expert's opinion.
- 8.3.6 The decision of the Neutral Expert on the disputed items shall be final and binding on the Parties for the purpose of determining the Company Purchase Price, as the case may be.

8.4 Final Effective Date Adjustment Statement

The Effective Date Adjustment Statement shall be final and binding on the Parties for the purpose of determining the Company Purchase Price,

- a) in accordance with Section 8.2.2, if and to the extent Sellers have not raised objections to the Effective Date Adjustment Statement within the time period set forth in Section 8.2.1;
- b) if and to the extent Sellers and Purchaser have reached an agreement concerning the disputed items within the time period set forth in Section 8.3.2 or at any time thereafter;
- c) in accordance with Section 8.3.2 in connection with Section 8.2.2, if and to the extent the Parties have not reached an agreement and neither Purchaser nor either Seller requests within the time limit set forth in Section 8.3.2 that the matter in dispute be decided by the Neutral Expert; or
- d) in accordance with Section 8.3.6, if and to the extent the Neutral Expert has decided about the unresolved disputed items;

(the so-determined final and binding Effective Date Adjustment Statement are referred to herein as "Final Effective Date Adjustment Statement").

8.5 Costs and Expenses

- a) Purchaser shall bear the costs and expenses of the preparation of the Effective Date Adjustment Statement.
- b) Sellers shall bear the costs and expenses of their own review of, and preparation of objections to, the Effective Date Adjustment Statement.
- c) Each Party shall bear its own costs incurred in connection with the dispute resolution procedure set forth in Section 8.3.

9. Calculation and Due Date of Adjustment Amount

- 9.1 If, on the basis of the Final Effective Date Adjustment Statement, the Company Purchase Price deviates from the Preliminary Purchase Price Company,
 - a) Purchaser shall pay (i) to Gebhardt Holding an amount equal to 60.68 percent and (ii) to ELBER an amount equal to 39.32 percent of the difference by which the Company Purchase Price exceeds the Preliminary Purchase Price Company;
 - b) (i) Gebhardt Holding shall pay to Purchaser an amount equal to 60.68 percent and (ii) ELBER shall pay to Purchaser an amount equal to 39.32 percent of the difference by which the Company Purchase Price falls short of the Preliminary Purchase Price Company

(any such amount to be paid by Purchaser or Sellers, the "Adjustment Amount").

- 9.2 The Adjustment Amount shall become due on the fifth Business Day after the Effective Date Adjustment Statement has become final and binding on the Parties.
- 9.3 The Adjustment Amount shall bear no additional interest. Section 10.3 remains unaffected.

10. General Rules for Payments; Bank Accounts

- 10.1 Any payments under or in connection with this Agreement shall be made in Euro (unless explicitly agreed otherwise in the following sentence) by wire transfer in immediately available funds without any kind of deductions, bank charges or other charges for the recipient. Fifty percent (50%) of the balance of the Preliminary Purchase Price Company minus (i) the Carve-Out Purchase Price, (ii) the Hold Back Amount, (iii) the Escrow Amount Company and (iv) the Escrow Amount Tax, shall be paid in US dollar, then applying the exchange rate set forth und published by the European Central Bank on the second Business Day prior to the relevant due date.
- 10.2 Except as explicitly set forth otherwise in this Agreement, all payments owed hereunder to Sellers or to Purchaser shall be paid to the bank accounts, as applicable, set forth on Exhibit 11.2.
 - Sellers and Purchaser may notify each other of a change of such bank accounts at any time by giving notice at least five (5) Business Days prior to the respective due date.
- 10.3 Should any Party or any of the Seller Guarantors fail to pay any amount payable by it under or in connection with this Agreement on its due date, such Party or such Seller Guarantor, respectively, shall be in default from the due date of the unpaid amount without further notice being required. Interest shall accrue on the unpaid amount from (and including) the day following the due date up to (and including) the date of receipt of the actual payment at a rate of nine percent (9%) *per annum*, to be calculated on the basis of the actual days elapsed and a year with 360 days. Further claims and remedies of the Parties in connection with such failure shall remain unaffected.

11. Escrow

11.1 Opening of Escrow Account; Purpose; General Terms

- 11.1.1 As soon as reasonably possible after the Signing Date, Sellers and Purchaser shall enter into an escrow agreement substantially in the form attached as Exhibit 11.1.1 (the "Escrow Agreement") with the acting notary (the "Escrow Agent"), who shall be instructed to establish a trust account (*Notaranderkonto*) (the "Escrow Account") as trustee for Sellers and Purchaser.
- 11.1.2 The Escrow Amount Company shall serve as collateral for Purchaser with respect to any claims of Purchaser against any Seller arising out of or in connection with this Agreement. Purchaser

- shall have the right to use any amounts of the Escrow Amount Company in the Escrow Account to satisfy any of its claims under or in connection with this Agreement against any of the Sellers.
- 11.1.3 The Escrow Amount Tax shall serve as collateral for Purchaser with respect to any claims of Purchaser against any Seller arising out or in connection with Section 25 (Taxes) of this Agreement.
- 11.1.4 Any interest accrued on the funds in the Escrow Account shall be credited to the Escrow Account. Any negative interest, account administration costs and other bank fees charged with respect to the Escrow Account and any fees and expenses of the Escrow Agent shall be paid out of the Escrow Account.

11.2 Releases

- 11.2.1 Any funds from the Escrow Account shall only be released by the Escrow Agent to Purchaser, on one hand, and the Sellers, on the other hand, in accordance with concurrent (*übereinstimmenden*) written instructions from the Parties.
- 11.2.2 Purchaser is entitled to request that Sellers, without undue delay, together with Purchaser instruct the Escrow Agent to release any funds in the Escrow Account to Purchaser if and to the extent that
 - c) any Seller is obliged to pay any Adjustment Amount to Purchaser pursuant to Section 9.1b) and Purchaser has requested, in its free discretion, such payment to be made from the Escrow Account; or
 - d) claims of Purchaser against any Seller under or in connection with this Agreement have been (i) acknowledged (*anerkannt*) by such Seller, or (ii) awarded in a legally binding (*rechtskräftig*) decision in principal proceedings (*im Hauptsacheverfahren*) by a competent court or by a competent arbitral tribunal.
- 11.2.3 From and after the expiry of three (3) years after the Closing Date upon the written request of Sellers, Purchaser shall, without undue delay, join Sellers in instructing the Escrow Agent to release to Sellers, in accordance with payment instructions relating to the distribution of payments to Gebhardt Holding and ELBER provided by Sellers, any funds of the Escrow Amount Company in the Escrow Account that remain after deduction of the amounts of
 - a) all outstanding claims with regard to which Purchaser is entitled to request release pursuant to Section 11.2.2; and
 - b) all outstanding claims that have been asserted by Purchaser against Sellers until such date by means of a duly filed lawsuit or request for arbitration in accordance with the terms of this Agreement (the "Asserted Claims").
- 11.2.4 From and after the expiry of three (3) years after the Closing Date, upon the written request of Sellers, Purchaser shall, without undue delay, join Sellers in instructing the Escrow Agent to

release to Sellers, in accordance with payment instructions relating to the distribution of payments to Gebhardt Holding and ELBER provided by Sellers, any funds of the Escrow Amount Tax in the Escrow Account up to an amount of EUR 500,000 that remain after deduction of the amounts of

- a) all outstanding claims with regard to which Purchaser is entitled to request release pursuant to Section 11.2.2; and
- b) all outstanding claims under or in connection with Section 25 (Taxes) of this Agreement that have been asserted by Purchaser against Sellers until such date by means of a duly filed lawsuit or request for arbitration in accordance with the terms of this Agreement (the "Asserted Claims Tax").

The above shall apply *mutatis mutandis* from and after expiry of five (5) years after the Closing Date. From and after expiry of seven (7) years after the Closing Date, the above shall apply to any remaining funds of the Escrow Amount Tax in the Escrow Account.

11.2.5 At any time after the expiry of the relevant escrow period set forth in Sections 11.2.3 and 11.2.4, respectively, upon the written request of Sellers, Purchaser shall, without undue delay, join Sellers in instructing the Escrow Agent to release to Sellers, in accordance with payment instructions relating to the distribution of payments to Gebhardt Holding and ELBER provided by Sellers, any remaining funds of the Escrow Amount Company and/or the Escrow Amount Tax due to be distributed in accordance with Sections 11.2.3 and 11.2.4, in each case if and to the extent that Asserted Claims and/or Asserted Claims Tax have been (i) denied by a final court judgment (rechtskräftiges gerichtliches Urteil) or arbitral award (Schiedsspruch) or (ii) waived by Purchaser.

12. Purchaser's Financing

- Purchaser intends to borrow, under one or more existing or future facilities agreement(s) with certain banks under which, on the terms and subject to the conditions thereof, amongst others, Purchaser will be entitled to borrow funds to satisfy certain payment obligations of Purchaser under and in connection with this Agreement (this/these facilities agreement(s) and any preliminary and/or interim documentation preceding it, as the context requires, the "Purchaser Facilities Agreement").
- 12.2 Sellers shall use commercially reasonable efforts to effect that the relevant Target Group Companies and their respective directors, officers, employees and representatives will, upon and according to Purchaser's reasonable request and to the extent legally permissible, provide cooperation to Purchaser in order to facilitate the arranging, underwriting, syndicating or otherwise preparing and closing of the Purchaser Facilities Agreement. This shall, in particular, and to the extent legally permissible, include the provision of financial and operating data and other information regarding the Target Group Companies as is customarily required for completion of such financings to Purchaser and to the arrangers, underwriters, bookrunners and/or

prospective lenders under the Purchaser Facilities Agreement as well as to rating agencies (including the participation in meetings and/or giving of presentations). Cooperation under this Section 12.2 shall (i) be subject to appropriate measures to safeguard the confidentiality of competitively, commercially and/or otherwise sensitive information and business secrets (which shall, in particular, be provided on an outside-counsel-only basis to the extent legally permissible) and (ii) not be required to the extent that this would (a) comprise the issuing of any securities or the making of any borrowings by any Seller Group Entity or, prior to Closing, by any Target Group Company or (b) unreasonably interfere in any material respect with the business or operations of Sellers or the Target Group Companies.

13.3 It is understood and agreed that the foregoing shall not be interpreted in any way that any Target Group Company shall enter into the Purchaser Facilities Agreement or any other financing arrangements of the Purchaser or incur any liability under any document entered into in connection with it or relating thereto prior to or on the Closing Date. Any such arrangements and documents shall only be entered into and any accession to the Purchaser Facilities Agreement or any other financing arrangements of the Purchaser shall only take place after the Closing has occurred. None of the Target Group Companies are obliged to grant any security interest or enter into any guarantee or similar arrangements to secure any obligations under or in connection with the Purchaser Facilities Agreement, any other financing arrangements of the Purchaser or any other obligations of the Purchaser prior to or on the Closing Date and any such arrangements and documents shall only be entered into after the Closing has occurred.

13. VAT

The Company Purchase Price is a net amount that does not include VAT. If the sale and transfer of the Gebhardt Shares or the ELBER Shares are subject to VAT, the Sellers have the right to opt for VAT according to Section 9 para. 1 German VAT Act after consultation with the Purchaser. However, Purchaser shall only be obliged to pay VAT (due to an option exercised by Sellers or otherwise) in addition to the Company Purchase Price if and to the extent Purchaser can effectively deduct such VAT as input VAT based on a final and binding (formell und materiell bestandskräftig) VAT assessment.

V. Separation from the Seller Group

14. Pre-Closing Carve-out

14.1 Concurrently with the execution of this Agreement, (i) the Company as seller and SMA Holding as purchaser enter into a share purchase and transfer agreement relating to the sale and transfer of all shares in RKT to SMA Holding (part IV of the notarial deed to which this Agreement is attached as part I; the "RKT SPA"), and (ii) the Company as seller and the Sellers as purchaser enter into a share purchase and transfer agreement relating to the sale and transfer of all shares in

SMA Holding GmbH to the Sellers (part III of the notarial deed to which this Agreement is attached as part I; the "SMA Holding SPA").

- 14.2 The Sellers shall procure that RKT SPA and the SMA Holding SPA will be consummated concurrently with this Agreement as follows:
 - a) The Company shall terminate the PLTA on or before the Closing Date.
 - b) On or before the Closing Date, the Company shall terminate RKT's participation in the Cash Pool.
- 14.3 The measures and documents provided for in Section 14.2 are subject to any amendments as may be practically expedient or legally required, provided that amendments that have a more than immaterial impact on the operational, legal, fiscal, tax or economic situation or structure of the Target Group Companies shall require the consent of Purchaser.
- 14.4 Sellers shall keep Purchaser informed about the status of the measures set forth in Section 14.2 and shall deliver to Purchaser appropriate evidence of all material transactions at law or in fact that have been concluded, consummated or effected (as applicable) in order to implement such measures. This shall, in particular, comprise the delivery of (i) copies of all agreements entered into and documents executed by the Target Group Companies in accordance with this Section 14 for the implementation of the measures set forth in Section 14.1 and 14.2 and (ii) written evidence of all registrations and other formal steps required for giving legal effect to the transactions contemplated under such agreements/documents (together the "Material Pre-Closing Carve-out Documentation").
- 14.5 Except for liabilities incurred in the ordinary course of business under the agreements listed in Exhibit 14.5, otherwise in the ordinary course based on already ongoing individual purchase orders or purchase orders issued until the Closing Date or otherwise explicitly stipulated under this Agreement (the "Continuing Liabilities"), with effect as of the Closing Date, neither Purchaser nor any Affiliates of Purchaser (including, for the avoidance of doubt, the Target Group Companies) shall have any liability vis-à-vis Sellers or any Seller Group Entity (including in particular AW Holding GmbH, RKT, Actuator Solutions GmbH and SMA Holding), including but not limited to any liability under any agreement entered into to implement the Pre-Signing Carve-out or the pre-closing carve-out measures set forth in Section 14.1 through 14.4, including the Material Pre-Closing Carve-Out Documentation (such pre-closing carve-out together with the Pre-Signing Carve-out, the "Carve-outs"), or otherwise in connection with the Carve-outs.
- 14.6 Except for the Continuing Liabilities, Sellers and the Seller Guarantors shall not, and the Sellers shall procure that no Seller Group Entity (including in particular AW Holding GmbH, RKT, Actuator Solutions GmbH and SMA Holding) will, after the Closing Date, raise any claim against Purchaser, any Affiliate of Purchaser and/or any Target Group Company, out of or in connection with

- a) The Company's former capacity as direct or indirect shareholder of any Seller Group Entity (including in particular AW Holding GmbH, RKT, Actuator Solutions GmbH and SMA Holding);
- b) The conduct of business of any Target Group Company (including with respect to the supply of goods or the rendering of services by any Target Group Company to any Seller Group Entity (including in particular AW Holding GmbH, RKT, Actuator Solutions GmbH and SMA Holding)); or
- c) The Carve-Outs and any action, step or documentation related to it (unless otherwise reflected in this Agreement, the RKT SPA or the SMA SPA),

in each case irrespective of whether such claim is known or unknown on the Signing Date, actual or contingent, accrued or unaccrued and regardless of the legal basis of such claim (such claims not to be raised the "Indemnified Claims"). From the Effective Date and irrespective of whether it would have been permitted by applicable law to procure that no Indemnified Claims are asserted against Purchaser or any of its Affiliates (including in particular the Target Group Companies), Sellers shall indemnify (freistellen) Purchaser, any Affiliate of Purchaser and/or any Target Group Company from any Indemnified Claims raised by or in the name of any Seller Group Entity (including in particular by AW Holding GmbH, RKT, Actuator Solutions GmbH and SMA Holding) as well as by any third party, such as an insolvency administrator, on their behalf, or by any assignee, and from any costs and expenses arising in connection therewith (being a contract for the benefit of third parties (echter Vertrag zugunsten Dritter) with regard to the Affiliates of Purchaser).

14.7 From the Effective Date, Sellers and the Seller Guarantors shall not, and Sellers shall procure that none of the Seller Group Entities will, raise any claim against any present or former director, officer, board member, employee, advisor or agent of any Target Group Company (in each case, other than each Seller, Seller Guarantor and Seller Group Entity) (each a "Beneficiary") out of or in connection with the Beneficiary's former capacity as director, officer, board member, employee, advisor or agent of any Target Group Company or Seller Group Entity (including any services rendered in such capacity to any Seller Group Entity before the Effective Date), in each case irrespective of whether such claim is known or unknown on the Signing Date, actual or contingent, accrued or unaccrued and regardless of the legal basis of such claim, except solely for claims based on willful or fraudulent misconduct (such claims not to be raised the "Other Indemnified Claims"). From the Effective Date and irrespective of whether it would have been permitted by applicable law to procure that no Other Indemnified Claims are asserted against a Beneficiary, Sellers shall indemnify (freistellen) the relevant Beneficiary from any Other Indemnified Claims raised by or in the name of any Seller Group Entity or any other Affiliate of any Seller (including by any third party, such as an insolvency administrator, on their behalf) or by any assignee, and from any costs and expenses arising in connection therewith (being a contract for the benefit of third parties (echter Vertrag zugunsten Dritter) with regard to any Beneficiary).

- 14.8 If, after the Effective Date, any person or entity asserts any claims against any of the Purchaser, its Affiliates (including in particular the Target Group Companies) or any Beneficiary in connection with any existing or future liability or obligation of any Seller Group Entity (including in particular AW Holding GmbH, RKT, Actuator Solutions GmbH and SMA Holding), irrespective of whether such claim is on the Signing Date known or unknown, actual or contingent, accrued or unaccrued and regardless of its legal basis (including any kind of joint or secondary liability), Sellers shall indemnify (*freistellen*) the relevant party from any such claim and from any costs and expenses arising in connection therewith.
- 14.9 Except for the Continuing Liabilities and except for any claims in connection with or under this Agreement, Purchaser shall not, and shall procure that no Target Group Entity will, after the Closing Date, raise any claim (irrespective of whether such claim is known or unknown on the Signing Date, actual or contingent, accrued or unaccrued and regardless of the legal basis of such claim) against any Seller, any Affiliate of any Seller or Seller Group Entity, out of or in connection with
 - a) the Sellers' former capacity as direct or indirect shareholders of any Target Group Company;
 - b) the PLTA and the termination thereof;
 - c) the conduct of business of any Seller Group Entity (excluding with respect to the supply of goods or the rendering of services by any Seller Group Entity to any Target Group Company until the Closing Date); or
 - d) the Carve-Outs and any action, step or documentation related to it (unless otherwise reflected in this Agreement, the RKT SPA or the SMA SPA).
- 14.10 From the Effective Date and irrespective of whether it would have been permitted by applicable law to procure that no such claims are asserted against Sellers or any of their Affiliates (including in particular the Target Group Companies), Purchaser shall indemnify (*freistellen*) Sellers, any Affiliate of Sellers and/or any Seller Group Entity from any such claims raised by or in the name of Purchaser, any Affiliate of Purchaser or any Target Group Entity as well as by any third party, such as an insolvency administrator, on their behalf, or by any assignee, and from any costs and expenses arising in connection therewith (being a contract for the benefit of third parties (*echter Vertrag zugunsten Dritter*) with regard to the Affiliates of Sellers).
- 14.11 With effect from the Effective Date, except for any Continuing Liabilities and except for any claims in connection with or under this Agreement, Purchaser shall not and shall cause its Affiliates and the Target Group Companies not to, raise any claims against any Leaving Function Holder as well as against present or former director, officer, board member, employee, advisor or agent of any Seller Group Entity, in each case to the extent such claim is not based on willful or fraudulent conduct. From the Effective Date and irrespective of whether it would have been permitted by applicable law to procure that no such claims are asserted against any of the Leaving Function Holders or any other of the aforementioned persons, Purchaser shall indemnify

(*freistellen*) the relevant beneficiary from any of such claims raised by or in the name of the Purchaser, any Affiliate of the Purchaser or any Target Group Entity (including by any third party, such as an insolvency administrator, on their behalf) or by any assignee, and from any costs and expenses arising in connection therewith (being a contract for the benefit of third parties (*echter Vertrag zugunsten Dritter*) with regard to any beneficiary).

14.12 Sections 24.2 shall apply to any claims under or in connection with this Section 14 *mutatis mutandis*. For the avoidance of doubt, Sections 27.2b) and 27.3 shall apply to claims under or in connection with this Section 14 as stipulated therein.

VI. Termination of External Financing Agreements

15. Repayment of External Financing

The Purchaser intends to repay and wishes that the Sellers make the necessary arrangements to repay and terminate and/or cancel, as the case may be, the Facilities made available under the Facilities Agreement, the 2017 Promissory Notes (if applicable), the 2019 Promissory Notes and the Other Reported Debt Agreements (the "Refinancing Debt"). In order to arrange for the repayment of the Refinancing Debt and the termination and/or cancelation, as the case may be, of the relevant underlying agreements as well as the release of the Financing Collateral, the Sellers shall, (i) procure or use reasonable efforts as further specified below to arrange for the repayment of the Refinancing Debt, and (ii) procure the release of the Financing Collateral as follows:

15.1 Repayment and Release Agreements and Arrangements

- 15.1.1 Sellers shall procure that, as soon as reasonably practicable after the Signing Date, the Company enters into one or more repayment and security release agreements executed by all parties to the Facilities Agreement (or the Security Agent and, if required, the Facility Agent on behalf of all parties to the Facilities Agreement) (with executed copies thereof to be delivered by the Sellers to the Purchaser as soon as reasonably practicable after execution) providing for the release, re-assignment and/or retransfer of all Financing Collateral at the Scheduled Closing Date or, if and to the extent such release, re-assignment and/or retransfer is not legally possible or practicable at the Scheduled Closing Date, as soon as reasonably practicable thereafter, conditional upon and subject to and against (*Zug-um-Zug*) the payment of the Facilities Repayment Amount into the Facilities Repayment Accounts on the Scheduled Closing Date and/or notice by the Facility Agent or the relevant creditor or creditor representative to the Security Agent that it has received the Facilities Repayment Amount owing to it (or to the creditors it represents) (the "Release Agreement Facilities").
- 15.1.2 Sellers shall use commercially reasonable efforts to achieve that, as soon as reasonably practicable after the Signing Date, the Company enters into one or more repayment agreements executed by the current lender(s) to the 2017 Promissory Notes (the "Repayment Agreement 2017 Promissory Note") providing for the repayment of the outstanding amounts under the 2017 Promissory Notes on the Scheduled Closing Date as further specified in the Repayment

Agreement 2017 Promissory Note. Sellers shall use commercially reasonable efforts to achieve that, as soon as reasonably practicable after the Signing Date, the Company enters into one or more repayment agreements executed by the current lender(s) to the 2019 Promissory Note (the "Repayment Agreement 2019 Promissory Note") providing for the repayment of the outstanding amounts under the 2019 Promissory Note on the Scheduled Closing Date as further specified in the Repayment Agreement 2019 Promissory Note.

- 15.1.3 Sellers shall use commercially reasonable efforts to achieve that, as soon as reasonably practicable after the Signing Date, the Company enters into one or more release agreements executed by the current lenders under the agreements for which the Company has provided the ASG/RKT Collateral in a form approved by the Purchaser (with executed copies thereof to be delivered by Sellers to Purchaser as soon as reasonably practicable after execution) providing for the release, re-assignment and/or retransfer of all ASG/RKT Collateral on or before the Scheduled Closing Date or, if and to the extent such release, re-assignment and/or retransfer is not legally possible or practicable on or before the Scheduled Closing Date, as soon as reasonably practicable thereafter (the "RKT Release Agreement").
- 15.1.4 Sellers shall procure that, as soon as reasonably practicable after the Signing Date, the Company enters into one or more repayment and release agreements executed by the current lender(s) to the Other Reported Debt Agreements (the "Release Agreements Other Debt" and together with the Release Agreement Facilities, the Repayment Agreement 2017/2019 Promissory Note and the RKT Release Agreement, the "Release Agreements") in a form approved by the Purchaser (with executed copies thereof to be delivered by Sellers to Purchaser as soon as reasonably practicable after execution) providing for the repayment and/or full satisfaction and discharge of the outstanding amounts under the Other Reported Debt Agreements and the release, re-assignment and/or retransfer of all Other Debt Collateral on or before the Scheduled Closing Date or, if and to the extent such release, re-assignment and/or retransfer is not legally possible or practicable on or before the Scheduled Closing Date, as soon as reasonably practicable thereafter, conditional upon and subject to and against (*Zug-um-Zug*) the payment of the outstanding amounts under the Other Reported Debt Agreements on the Scheduled Closing Date as further specified in the Release Agreements Other Debt. It is agreed and understood by all parties to this Agreement that the Release Agreements Other Debt can take any form of release agreement which is typically used for that type of facility provided under the relevant Other Reported Debt Agreement and/or in the relevant local financing market.
- 15.1.5 Sellers shall procure that the Company and all other Target Group Companies, as far as applicable, provide cooperation to Purchaser (i) to facilitate the arranging, underwriting, syndicating or otherwise preparing and finalizing of any new financing arrangements Purchaser intends to enter into to replace the existing financing and (ii) to enable Purchaser to borrow funds under any existing financing arrangements to replace the existing financing. Section 12 shall apply *mutatis mutandis*.

15.2 Repayment of Facilities

- 15.2.1 Purchaser shall pay to the Facility Agent (on behalf of the relevant Borrowers or other Obligors) on the Scheduled Closing Date the aggregate outstanding principal amount of the Facilities together with all accrued interest, commission and fees and all other amounts (including prepayment fees, break costs and fees, costs and expenses which are to be paid or reimbursed by an Obligor pursuant to the terms of the Facilities Agreement) falling due under or in connection with the Facilities (the "Facilities Repayment Amount") in full discharge of the payment obligations (*mit schuldbefreiender Wirkung*) of the Obligors under or in connection with the Facilities and to effect full release of all Financing Collateral as of the Closing Date.
- 15.2.2 For the purpose of the payments of Purchaser pursuant to Sections 15.2.1, Sellers shall procure to notify Purchaser, no later than five Business Days prior to the Scheduled Closing Date, in writing of (i) the Facilities Repayment Amount (or, if the definitive Facilities Repayment Amount is not yet available at that time, an estimate thereof) and (ii) one or more bank accounts (the "Facilities Repayment Account(s)") into which the Facilities Repayment Amount shall be paid (and the amount of the Facilities Repayment Amount to be paid to each Facilities Repayment Account, as applicable). Unless previously notified by Sellers to Purchaser, Sellers shall procure to notify Purchaser of the definitive Facilities Repayment Amount no later than two Business Days prior to the Scheduled Closing Date. It is agreed and understood by the Parties that the definitive Facilities Repayment Amount may be calculated in relation to a date which is supposed to be the Scheduled Closing Date accompanied by a statement which additional amounts need to be paid (in particular daily interest (Stückzinsen)) if the Closing occurs on a date other than the Scheduled Closing Date or the date deemed to be the Scheduled Closing Date for the purposes of the calculation of the Facilities Repayment Amount.
- 15.2.3 Sellers shall procure that the relevant Obligors, if and to the extent reasonably requested by Purchaser:
 - a) provided that all Closing Conditions have been fulfilled or duly waived and the Scheduled Closing Date has been determined in accordance with Section 20.1, serve notices of early termination, cancellation and/or prepayment (as applicable) in the manner and form required by the terms of, and in accordance with the procedures set forth in, the relevant financing agreements (which may be conditional upon the occurrence of Closing) in order to facilitate the repayment of the Facilities on the Scheduled Closing Date and send copies of such notices to Purchaser;
 - b) submit waiver and/or consent requests to the Facility Agent in order to shorten any notice periods for cancellation and/or prepayment of the Facilities; and
 - c) negotiate in good faith the form of the Release Agreements, deliver to Purchaser reasonably advanced drafts of the Release Agreements immediately (*unverzüglich*) after receipt and review thereof and give due consideration to Purchaser's (or its legal advisers') comments on such drafts.

15.3 Repayment of 2019 Promissory Notes

- 15.3.1 Purchaser shall pay to the Paying Agent on the Scheduled Closing Date the aggregate outstanding principal amount of the 2019 Promissory Notes together with all accrued interest, commission and fees and all other amounts (including prepayment fees, break costs and fees, costs and expenses which are to be paid or reimbursed by the Company pursuant to the terms of the 2019 Promissory Notes) falling due under or in connection with the 2019 Promissory Notes (the "**Promissory Note Repayment Amount**") in discharge of the payment obligations (*mit schuldbefreiender Wirkung*) of the Company under or in connection with the 2019 Promissory Notes.
- 15.3.2 For the purpose of the payments of Purchaser pursuant to Sections 15.2.1, Sellers shall use commercially reasonable efforts to notify Purchaser, no later than five Business Days prior to the Scheduled Closing Date, in writing of (i) the Promissory Note Repayment Amount (or, if the definitive Promissory Note Repayment Amount is not yet available at that time, an estimate thereof) and (ii) one or more bank accounts (the "Promissory Note Repayment Account(s)") into which the Promissory Note Repayment Amount shall be paid (and the amount of the Promissory Note Repayment Amount to be paid to each Promissory Note Repayment Account, as applicable). Unless previously notified by Sellers to Purchaser, Sellers shall use commercially reasonable efforts to notify Purchaser of the definitive Promissory Note Repayment Amount no later than two Business Day prior to the Scheduled Closing Date. It is agreed and understood by all parties to this Agreement that the definitive Promissory Note Repayment Amount may be calculated in relation to a date which is supposed to be the Scheduled Closing Date or a date which is likely to be the Scheduled Closing Date accompanied by a statement which additional amounts need to be paid (in particular daily interest (Stückzinsen)) if the Closing occurs on a date other than the Scheduled Closing Date or the date deemed to be the Scheduled Closing Date for the purposes of the calculation of the Promissory Note Repayment Amount.
- 15.3.3 Sellers shall procure that the Company, if and to the extent reasonably requested by Purchaser:
 - a) provided that all Closing Conditions have been fulfilled or duly waived, serves notices of early termination, cancellation and/or prepayment (as applicable) in the manner and form required by the terms of, and in accordance with the procedures set forth in, the 2019 Promissory Notes (which may be conditional upon the occurrence of Closing) in order to facilitate the repayment of the 2019 Promissory Notes on the Scheduled Closing Date and send copies of such notices to Purchaser;
 - b) submit waiver and/or consent requests to the Paying Agent in order to shorten any notice periods for cancellation and/or prepayment of the 2019 Promissory Notes; and
 - c) negotiate in good faith the form of the Release Agreements, deliver to Purchaser reasonably advanced drafts of the Release Agreements immediately (*unverzüglich*) after receipt and review thereof and give due consideration to the Purchaser's (or its legal advisers') comments on such drafts.

15.4 Repayment or Other Discharge of the Other Reported Debt

Sellers shall procure that the Target Group Companies terminate the Other Reported Debt Agreements and repay or otherwise satisfy and discharge all outstanding amounts under the Other Reported Debt no later than on the Scheduled Closing Date.

VII. Further Pre-Closing Covenants

16. Obtaining of Governmental Clearances/Approvals

16.1 Merger Control

- 16.1.1 As soon as reasonably practicable after the Signing Date and in any event within fifteen (15) Business Days following the Signing Date, Purchaser shall make the filings for the merger control clearances in the jurisdictions set out in <u>Exhibit 16.1.1</u> (together the "Merger Control Clearances") with the competent merger control authorities (together the "Merger Control Authorities"). Purchaser shall use commercially reasonable efforts to avoid any declaration of incompleteness by the Merger Control Authorities or any other suspension for the time period of clearance. Where the applicable law requires that a filing must also be made by Sellers, Purchaser shall, subject to the prior written consent of Sellers, make that relevant filing also on behalf of Sellers.
- 16.1.2 Sellers shall closely cooperate with Purchaser for the purpose of the filings set forth in Section 16.1.1 and shall provide without undue delay (*unverzüglich*) all information reasonably requested by Purchaser in this connection concerning Sellers, the Seller Group Entities, the Target Group Companies and the Business, including any additional information and documentary material requested by any Merger Control Authority or other Governmental Entity.
- 16.1.3 Purchaser shall give Sellers reasonable opportunity to review and comment on any drafts of filing or other material documentation prior to their submission to any Merger Control Authority or other competent Governmental Entity (it being acknowledged that certain of such drafts and/or documents may be shared on a confidential outside counsel to counsel basis only) and shall give due consideration to any reasonable comments. Purchaser shall respond as soon as reasonably possible and to the extent the relevant information is available to Purchaser to all inquiries received from any Merger Control Authority or other competent Governmental Entity to whom a filing has been made for additional information or documentation and supplement such filings as reasonably requested by the applicable Merger Control Authority or other competent Governmental Entity. Purchaser undertakes to keep Sellers informed of any material contact with such Merger Control Authority and to the extent legally permissible, provide Sellers with copies of all material relevant documentation in relation thereto and allow Sellers to participate in any material call or meeting with any Merger Control Authority or other Governmental Entity.

- 16.1.4 Purchaser shall in no event be required (i) to offer, or consent to, any obligations, conditions or commitments required by any Merger Control Authority in connection with the Merger Control Clearances or (ii) to contest a decision of any Merger Control Authority or court prohibiting the consummation of the Transaction or parts thereof. Notwithstanding the foregoing, Purchaser shall cooperate with Sellers in good faith and use commercially reasonable efforts to reach the Merger Control Clearances.
- 16.1.5 For the avoidance of doubt, the communication among the Parties in connection with this Section 16.1 shall be subject to (i) compliance with mandatory legal restrictions and (ii) appropriate measures to safeguard the confidentiality of competitively, commercially and/or otherwise sensitive information and business secrets (which shall, in particular, be provided on an outside-counsel-only basis).

16.2 German Foreign Investment Control Clearance

- 16.2.1 As soon as reasonably practicable after the Signing Date and in any event within ten Business Days following the Signing Date, Purchaser shall file an application with the German Federal Ministry for Economic Affairs and Climate Action (*Bundesministerium für Wirtschaft und Klimaschutz*, the "**BMWK**") for the issuance of a compliance certificate pursuant to Section 58 (1) AWV (*Unbedenklichkeitsbescheinigung*) in relation to the Transaction (the "**Compliance Certificate**"). Sections 16.1.2 and 16.1.5 shall apply *mutatis mutandis*.
- 16.2.2 Purchaser shall in no event be required (i) to offer, or consent to, or comply with any obligations, conditions or commitments in connection with the Compliance Certificate or (ii) to contest a decision of the BMWK or any court prohibiting the consummation of the Transaction or parts thereof. Notwithstanding the foregoing, Purchaser shall cooperate with Sellers in good faith and use commercially reasonable efforts to reach the Compliance Certificate.

16.3 U.K. NSI Clearance

- 16.3.1 As soon as reasonably practicable after the Signing Date and in any event within fifteen (15) Business Days following the Signing Date, Purchaser shall submit to the ISU a completed mandatory notification and clearance application under the NSIA2021 for clearance by the ISU/BEIS of the sale and purchase of the Shares under this Agreement. Sections 16.1.2 and 16.1.5 shall apply *mutatis mutandis*.
- 16.3.2 Purchaser shall in no event be required (i) to offer, or consent to, or comply with any obligations, conditions or commitments in connection with such submission or (ii) to contest a decision of the the ISU/BEIS or any court prohibiting the consummation of the Transaction or parts thereof. Notwithstanding the foregoing, Purchaser shall cooperate with Sellers in good faith and use commercially reasonable efforts to obtain the NSI Clearance.

17. Sellers' Conduct of Business Covenants

17.1 General Principle; Consent Requirements

- 17.1.1 Within the period between the Signing Date and the Closing Date (the "Interim Period"), Sellers shall procure that the Target Group Companies (i) conduct their business operations with the care of a prudent businessman (Sorgfalt eines ordentlichen Geschäftsmanns) and consistent with past practice, and (ii) do not take any measure or activity outside the ordinary course of business or inconsistent with past practice unless Purchaser has consented to it.
- 17.1.2 During the Interim Period, and unless explicitly set forth in <u>Exhibit 17.1.2</u>, Sellers shall procure that the Target Group Companies do not take any of the following measures or activities unless Purchaser has consented in writing to it (such consent not to be unreasonably withheld or delayed):
 - a) Divesting or acquiring, directly or indirectly, any shares or any other equity interests in companies or of any business or entering into any type of merger, consolidation or similar transaction;
 - b) Divesting any tangible fixed assets (*Sachanlagevermögen*) with a book value in excess of EUR 50,000 (in words: fifty thousand Euro) in each individual case or EUR 250,000 (in words: Euro two hundred fifty thousand) in the aggregate;
 - c) Acquiring any tangible fixed assets (*Sachanlagevermögen*) with a book value or making capital expenditures each in excess of EUR 125,000 (in words: Euro one hundred twenty-five thousand) in each individual case or EUR 375,000 (in words: Euro three hundred seventy-five thousand) in the aggregate;
 - d) Failing to make any capital expenditures necessary to continue the operations of the Business of the Target Group Companies in a manner in all material respects consistent as conducted on the Signing Date; lit. c) shall remain unaffected;
 - e) Disposing or licensing of any Owned IP Rights or other intellectual property rights or know-how material for the Business or disclosing any technical or confidential information material for the Business outside the ordinary course of business or inconsistent with past practice;
 - f) Granting any transfer by way of security (*Sicherungsübereignung*), pledge, lien or other security interest, any encumbrance (registered or unregistered), any right to use the relevant property or asset (including the granting of any license), any pre-emptive right, option or similar conversion right or any remainder (*Anwartschaftsrecht*) of any property or asset (including any shares/interest in companies and including intellectual property rights and know-how), in each case except as granted in the ordinary course of business and consistent with past practice (in particular leasing);

- g) Paying or otherwise discharging any obligation of persons other than any Target Group Company or issuing of any guarantees, suretyships, letters of credit, performance or warranty bonds or similar instruments in order to secure any such obligations outside the ordinary course of business or inconsistent with past practice;
- h) Procuring of or agreeing to the issuing of any guarantees, suretyships, letters of credit, performance or warranty bonds or similar instruments by persons other than any Target Group Company in order to secure any obligations of any Target Group Company, in each case except as issued in the ordinary course of business and consistent with past practice;
- i) Taking up of any loan or credit or receiving of any other financing (including from any Seller Group Entity and increasing indebtedness under any existing loan, credit or financing), in each case except for financing provided by any other Target Group Company, in excess of EUR 50,000 (in words: Euro fifty thousand) in each individual case;
- j) Granting of any loan or credit or providing of any other financing to any third party (including to any Seller Group Entities), in each case except for financing provided to any other Target Group Company;
- k) Entering into or materially amending any material research or development agreements outside the ordinary course of business or inconsistent with past practice;
- l) Granting or paying any costs, fees, commission, bonus or other benefit that will arise directly as a result of this Agreement or the transactions contemplated herein including (i) fees to any advisors, finders, brokers or agents and (ii) transaction-related bonuses to members of the management or to any employees;
- m) Entering into or materially amending any advisory or consultancy agreements with any advisor or consultant;
- n) Cancelling, waiving or settling any claims or rights exceeding EUR 50,000 (in words: Euro fifty thousand) in each individual case or EUR 125,000 (in words: Euro one hundred twenty-five thousand) in the aggregate;
- o) To the extent not already covered by any other provision of this Section 17.1.2: (i) terminating or materially amending any Material Agreement or entering into any agreement that would have qualified as Material Agreement if it had been entered into prior to the Signing Date; (ii) entering into or materially amending any agreement involving annual payment obligations in excess of EUR 50,000 (in words: Euro fifty thousand) in each individual case, except for any supply agreements entered into in the ordinary course of business and consistent with past practice; (iii) entering into or materially amending or terminating any agreement between any Seller Group Entity and any Target Group Company; or (iv) incurring a default under any Material Agreement or failing to renew any Material Agreement in the ordinary course of business (unless substituted or no longer

required) or take or omit to take any action that could be reasonably expected to cause or result in a breach or default under any Material Agreement;

- p) Hiring or dismissing any Key Employee (except any dismissal for cause (aus wichtigem Grund)); changing to the advantage of a Key Employee of his/her employment conditions (including with respect to remuneration, pension or other post-retirement benefits, paid leave, termination, etc.) outside the ordinary course of business;
- q) Changing the general employment conditions to the advantage of all or groups of the employees (including with respect to remuneration, pension or other post-retirement benefits, paid leave, termination, etc.), in each case except as (i) required by law, provided for by collective bargaining agreements (*Tarifverträge*) or by references to such collective bargaining agreements in the individual employment contracts in effect prior to the Signing Date, or (ii) made in the ordinary course of business and consistent with past practice;
- r) Entering into or material amendment of any collective agreement with any trade union, works council or any other employee representative body;
- s) Initiating, compromising or settling of any legal or administrative proceedings (whether as applicant, plaintiff or defendant) with an economic relevance for the respective Target Group Company that is likely to exceed EUR 25,000 (in words: Euro twenty-five thousand) in each case or in a series of related cases;
- t) Entering into any agreements with public authorities (including Taxing Authorities) or any entities controlled by any public authority which relate to any regulatory matter or other matter governed by public law;
- u) Materially changing any method of accounting or tax reporting, or accounting or tax practice or policy currently applied, as the case may be, except as required by reason of a concurrent change in, or conversion to, the applicable generally accepted accounting principles;
- v) Materially changing the nature or organization of the Business, including (i) launching or ceasing product lines (except in the ordinary course of business) and of branches of business, (ii) introducing new brands or corporate names or material changes to or surrendering existing brands or corporate names, (iii) materially changing business policies (e.g., with respect to customer approach or general pricing, public relations, corporate social responsibility), (iv) materially changing internal structures and policies (e.g., with respect to material responsibilities and reporting lines, compliance procedures), and (v) significantly changing size or structure of the workforce or general working practices;
- w) Incorporating or forming any subsidiaries; entering into any partnership, joint venture, consortium, shareholders' agreement or similar agreement;

- x) Entering into any agreement or assuming of any obligation connected with any of the measures listed in Section 17.1.3a);
- y) Adopting any change in any of its organizational documents, except as required under applicable law;
- z) Liquidating, dissolving, recapitalizing, reorganizing or otherwise winding up all or any portion of the Business or any Target Group Company;
- aa) Selling or transferring, or permitting the sale or transfer of any shares or interests of the Target Group Companies to any person, or in any manner restraining or limiting the rights under such shares or interests or create any liens or encumbrances thereon;
- bb) Paying any dividends or make any distributions outside the Target Group Companies;
- cc) Making any changes (by stock split, reverse stock split, combination, recapitalization, reorganization, formation of new subsidiaries or otherwise) in the capital stock structure of any Target Group Companies;
- dd) Failing to maintain all of the tangible assets of the Target Group Companies in operating condition subject only to ordinary wear and tear;
- ee) Failing to pay or satisfy when due any liability (other than any such liability that is being contested in good faith and by appropriate proceedings);
- ff) Assuming any liability or obligation of any person or entity other than a Target Group Company outside of the ordinary course of business and consistent with past practice in excess of EUR 10,000 (in words: Euro ten thousand) in each individual case or EUR 50,000 (in words: Euro fifty thousand) in the aggregate;
- gg) Failing to maintain all material governmental approvals or Permits as valid and in full force, pursuant to applicable laws;
- hh) Fail to maintain insurance coverage in accordance with the insurance agreements existing on the Signing Date; and
- ii) Entering into any agreement to do any of the foregoing.
- 17.1.3 During the Interim Period, Sellers shall not take or participate in, and shall, to the extent applicable, procure that no other Seller Group Entity will take or participate in, directly or indirectly, any of the following measures or activities, unless Purchaser has consented to it:
 - a) Any material corporate law measures with respect to the Company (and, if and to the extent applicable, the other Target Group Companies) including (i) amendments to the articles of association; (ii) reorganization, dissolution or liquidation; (iii) consent to the conclusion of any enterprise agreement within the meaning of sections 291, 292 AktG or comparable agreements under the laws of other jurisdictions; (iv) authorization to repurchase shares;

- (v) issuance of shares or other equity interest (including options, warrants or conversion rights with respect to such shares or other equity interest); (vi) appointment or dismissal of members of the administrative board (*Verwaltungsrat*); (vii) waiver of claims against current or former members of the administrative board; (viii) appointment of new auditors; (ix) declaration and/or payment of (deemed) dividends or similar distributions (whether in cash or in kind), except if effected prior to the Effective Date; (x) approval (including restatements) of financial statements;
- b) Issuing of any guarantees, suretyships, letters of credit, performance or warranty bonds or similar instruments by any Seller Group Entity in order to secure indebtedness or other obligations of any Target Group Company or procuring the issuing of such instruments by other persons on behalf of any Seller Group Entity;
- c) Divesting of any Shares or permitting any Shares to be subjected to any transfer by way of security (*Sicherungsübereignung*), pledge, lien or other security interest, any pre-emptive right, option or similar conversion right or any remainder (*Anwartschaftsrecht*); or
- d) Soliciting, initiating, or encouraging any inquiries or proposals from, discussing or negotiating with, providing any information to, or considering the merits of any unsolicited inquiries or proposals from, any person or entity relating to any transaction involving the sale of the Business or the assets of the Target Group Companies, or the Shares, or any merger, consolidation, business combination, or similar transaction involving the Target Group Companies.
- 17.1.4 In deviation of Section 36.1, the requests for consent from Sellers and the consents or rejections from Purchaser shall be sent to the respective other Party in text form (*Textform*) via email to the email addresses set forth in <u>Exhibit 17.1.4</u>. Each Party may replace the email addresses set forth for it for the purpose of this Section 17.1.4 by notification to the respective other Party in the form of Section 36.1.
- 17.1.5 During the Interim Period, Sellers shall procure that Sellers and the Target Group Companies (i) take all actions and make all filings required by applicable laws to be made by them in order to consummate the Transaction and (b) use its commercially reasonable efforts to cause the conditions in Section 21 to be satisfied (other than those conditions outside of Sellers' control).
- 18.1.6 For the avoidance of doubt, the Sellers and the Target Group Companies shall not be limited or prohibited hereunder from any measures or activities contemplated under this Agreement, including the execution and implementation of the RKT SPA and the SMA SPA.
- 18.1.6 During the Interim Period, Sellers shall:
 - a) Carry out any and all actions that may be necessary or convenient to maintain full and legal title to the Shares free from any and all liens;

- b) Notify Purchaser of the receipt by the Target Group Companies of (i) any written notice of breach or default of its obligations under any Material Agreement and (ii) any written notice of termination of any Material Agreement; and
- c) Notify Purchaser of the receipt by the Target Group Companies of any material (i) arbitration, (ii) administrative procedure or (iii) other action or litigation with respect to the Target Group Companies.

17.2 Exchange of Function Holders; Agreements with Leaving Individuals

- 17.2.1 Sellers shall procure that, on or before the Scheduled Closing Date:
 - a) the executive directors (*geschäftsführende Direktoren*), managing directors (*Geschäftsführer*) and/or other function holders of the Target Group Companies listed in <u>Exhibit 17.1.2a</u>) (the "**Leaving Function Holders**") will either resign or be removed from their positions specified in <u>Exhibit 17.2a</u>) with effect as of the Closing Date at the latest.
 - b) administrative board or shareholders' resolution(s), as applicable, regarding (i) the appointment of new executive directors (*geschäftsführende Direktoren*) or new managing directors (*Geschäftsführer*), as applicable, of the Target Group Companies (each a "New Function Holder") and (ii) the change in representation for the existing executive directors (*geschäftsführende Direktoren*) or existing managing directors (*Geschäftsführer*), respectively, of the Target Group Companies (each an "Existing Function Holder"), each as specified in Exhibit 17.2.1b);
 - the Company on the one hand and Andreas Gebhardt on the other hand enter into the termination agreement (*Aufhebungsvertrag*) substantially in the form as attached hereto as <u>Exhibit 17.2.1c</u>) (the "**Termination Agreement Andreas Gebhardt**") with effect as of the Effective Date;
 - d) the Company on the one hand and Markus Gebhardt on the other hand enter into the termination agreement (Aufhebungsvertrag) substantially in the form as attached hereto as Exhibit 17.2.1d) (the "Termination Agreement Markus Gebhardt") with effect as of the Effective Date;
 - e) the Company on the one hand and Stephan Schulak on the other hand enter into the termination agreement (*Aufhebungsvertrag*) substantially in the form as attached hereto as <u>Exhibit 17.2.1e</u>) (the "**Termination Agreement Stephan Schulak**") with effect as of the Effective Date;
 - f) the Company on the one hand and each of Julia Jürgensen, Ute Zauleck and Nastasia Gläser on the other hand enter into a customary termination agreement (*Aufhebungsvertrag*) (the "**Employment Termination Agreements**") regarding the termination of their respective employment agreements with the Company, in each case with effect as of the Effective Date and providing for a customary mutual waiver of claims;

- g) the Company on the one hand and Andreas Gebhardt enter into the consultancy agreement substantially in the form as attached hereto as Exhibit 17.2.1g) (the "Andreas Gebhardt Consultancy Agreement") with effect as of the Effective Day and until the expiry of six (6) months following the Effective Date; and
- the Company and the relevant Seller Group Entity enter into a transitional services agreement substantially in the form as attached hereto as Exhibit 17.2.1h) (the "Transitional Services Agreement") according to which certain individuals listed therein will provide the Company with the services further determined in the Transitional Services Agreement with effect as of the Effective Day and until the expiry of three (3) months following the Effective Date.
- 17.2.2 The Sellers will effect that on or before the Closing Date, the Company will terminate the existing D&O insurance of the Target Group Companies with effect as of the end of the current calendar year and a run-off coverage will be provided at no additional cost for Purchaser, the Company or any Target Group Company. The Parties agree that regarding the capacity of Andreas Gebhardt and Markus Gebhardt as executive directors, managing directors or otherwise function holders of any Target Group Company until the Closing Date, only the liability regime of this Agreement shall apply.

17.3 Transfer of Patents

Sellers shall procure that Actuator Solutions GmbH transfers its rights in the patents nos. US 16/833,996; DE 102020107952.4; DE 202019101901.8 and DE 102019108624.8 to the Company without delay (*unverzüglich*) after the Signing Date.

17.4 Amendment of Alfmeier Friedrichs & Rath LLC Put/Call Agreement and Exercise of Call

Sellers shall procure that (i) the Alfmeier Friedrichs & Rath LLC Put/Call Agreement dated and effective 1 January 2020 is amended to provide that the Call (as defined therein) shall be revised to provide that it may be exercised by Alfmeier Corporation at any time after the date of such amendment for a call price as set forth in the amendment ("Call Price") and (ii) the Call is exercised on or around the Closing such that, immediately following the Closing, Alfmeier Corporation holds the entire membership interest and all of equity interests of AF&R and no liabilities or obligations remain due and owing to Friedrichs & Rath, Inc. or its Affiliates under the Alfmeier Friedrichs & Rath LLC Put/Call Agreement or the Amended Operating Agreement Alfmeier Friedrichs & Rath LLC from any Target Group Company following the payment of the Call Price.

17.5 Change of Control Rights

The Parties agree that no contractual counterparty to any Material Agreement providing for a termination right or other right upon a change of control in the relevant Target Group Company shall, unless otherwise agreed between the Parties, be contacted in the Interim Period in order to

obtain a waiver and that, hence, the Purchaser shall have no claims against the Sellers in connection with refraining to do so, again unless otherwise agreed between the Parties.

18. Cooperation by Sellers prior to Closing; Information Rights of Purchaser

- 18.1 Until and including the Closing Date, Sellers shall, and shall procure that the Seller Group Entities and the Target Group Companies will, to the extent reasonably requested by Purchaser, allow Purchaser to prepare the post-Closing integration of the Business into the business of the Purchaser group and cooperate with Purchaser for such purpose, provided, in each case, that the cooperation requested does not unreasonably interfere with the business or operations of Sellers or any of the Target Group Companies and, in each case, subject to compliance with applicable law. In particular, to the extent that any information to be provided to Purchaser is competitively, commercially and/or otherwise sensitive, Sellers shall, to the extent reasonably assessed to be so required, provide such information on an outside-counsel-only basis, subject to appropriate confidentiality obligations.
- 18.2 Any further obligations of Sellers pursuant to this Agreement to provide information or other cooperation to Purchaser remain unaffected from this Section 18.

VIII. Closing

19. Closing Conditions

- 19.1 The obligations of the Parties to take the Closing Actions (the "Closing") pursuant to Section 20 shall be subject to the fulfillment or waiver of the following conditions precedent (*aufschiebende Bedingungen*) pursuant to Section 158 (1) BGB (the "Closing Conditions"):
 - a) The Merger Control Clearances have been obtained or are deemed to be obtained pursuant to applicable laws in the relevant jurisdiction, e.g. because of the expiry of applicable waiting periods or because jurisdiction has been declined, and provided that such Merger Control Clearances are not subject to any conditions (Bedingungen), restrictions (Beschränkungen) or obligations (Handlungspflichten).
 - b) (i) Purchaser has received a Compliance Certificate or another binding decision of, or concluded a public law contract (öffentlich-rechtlicher Vertrag) with, the BMWK confirming that the Transaction does not raise concerns with respect to the public order or security of the Federal Republic of Germany or of another member states of the European Union or in relation to projects or programs of Union interest), provided that such binding Compliance Certificate or decision is not subject to any conditions (Bedingungen), restrictions (Beschränkungen) or obligations (Handlungspflichten), (ii) a Compliance Certificate is deemed to have been issued pursuant to Section 58 (2) AWV or restrictions (Beschränkungen) and obligations (Handlungspflichten) cannot be imposed anymore due to the expiry of the time periods set out in Section 14a AWG, in each case because the time

periods during which the BMWK may initiate a formal foreign investment review or impose restrictions or obligations have expired without the BMWK taking such measures, or (iii) the BMWK has, within the time periods set out in Section 14a AWG imposed restrictions or obligations in relation to the Transaction, without prohibiting it, and Purchaser has agreed to accept such restrictions or obligations (without being in anyway obliged to do so).

- c) The Purchaser has received confirmation from the ISU/BEIS under section 14(8)(b)(ii) or section 26(1)(b) of the NSIA2021 that no further action will be taken in relation to the sale and purchase of the Shares under this Agreement (the "NSI Clearance").
- d) The Sellers have provided the Purchaser with a copy of the executed Release Agreements, evidencing (i) that, subject to any mandatory notarization, filing or de-registration requirements, all Financial Collateral are fully released, terminated and/or retransferred, as applicable, upon payment of the Facilities Repayment Amount into the Facilities Repayment Accounts on the Scheduled Closing Date, (ii) the full discharge of the payment obligations under the 2019 Promissory Note by payment of the Promissory Note Repayment Amount into the Promissory Note Repayment Account(s).
- e) The Assumption Agreements Carve-out have been concluded.
- f) Actuator Solutions GmbH has transferred its rights in the patent families nos. US 16/833,996; DE 102020107952.4; DE 202019101901.8 and DE 102019108624.8 to the Company.
- g) The Sellers have provided the Purchaser with (i) the applicable beneficial ownership declaration and tax residency certification of the Company in respect of interest paid by Alfmeier CZ to the Company in the fiscal years ending 31 December 2018 through 31 December 2021 and dividends paid by Alfmeier CZ to the Company in the fiscal year ending 31 December 2021 and (ii) evidence that Alfmeier CZ has filed with the Czech Taxing Authorities all required notifications of such dividends paid by Alfmeier CZ to the Company in the fiscal year ending 31 December 2021.
- h) The Alfmeier Friedrichs & Rath LLC Put/Call Agreement has been amended in accordance with Section 17.4.
- 19.2 The Parties shall use all commercially reasonable efforts to fulfil the Closing Conditions as soon as possible after the date hereof. The Parties shall notify each other of the fulfillment or the definitive failure (endgültiger Nichteintritt) of any Closing Condition without undue delay (unverzüglich) after obtaining knowledge thereof, and shall include in such notifications all relevant evidencing documents.
- 19.3 The Closing Conditions referred to in Section 19.1b), d), f) g) and h) may be waived by Purchaser only. The other Closing Conditions may only be waived by mutual agreement of the Parties, such agreement to be made in the form of Section 36.1. Any waiver of a Closing Condition may be

made in full or in part. The effect of a waiver of a Closing Condition shall be limited to eliminating the need that such Closing Condition be fulfilled and, unless otherwise agreed, shall not limit or prejudice any claims that a waiving Party may have with respect to any circumstances relating to such Closing Condition not having been fulfilled.

20. Closing

- 20.1 The Closing shall occur (i) at the offices of Gleiss Lutz, Karl-Scharnagl-Ring 6, 80539 Munich, Germany, at 10:00 a.m. on the last Business Day of the calendar month in which all Closing Conditions are fulfilled or duly waived, unless such date is less than five (5) Business Days before the last Business Day of the calendar month then current, in which case the Closing shall occur on the last Business Day of the next calendar month, but in any of the aforementioned cases not before 30 June 2022, or (ii) at such other time, date or place as the Parties have mutually agreed upon in the form of Section 36.1. The date on which the Closing is scheduled to occur according to the foregoing sentence shall be referred to as the "Scheduled Closing Date". The date on which Closing actually occurs shall be referred to as the "Closing Date".
- 20.2 On the Scheduled Closing Date, the Parties shall take or, if applicable, cause to be taken, concurrently (*Zug um Zug*) the following actions (the "Closing Actions"), in each case to the extent not already taken before in accordance with this Agreement:
 - a) If not occurred before, Sellers shall deliver to Purchaser evidence that the Company has terminated the PLTA with effect as of the Effective Date at the latest by delivering a written notice of termination for cause (außerordentliche Kündigung aus wichtigem Grund) to RKT.
 - b) Sellers shall deliver to Purchaser evidence that the shares in SMA Holding have effectively been transferred to Sellers and that the shares in RKT have been effectively transferred to SMA Holding.
 - c) Sellers shall deliver to Purchaser evidence that the administrative board or shareholders' resolutions regarding the appointment of the New Function Holders and the change in representation for the Existing Function Holders have been passed in accordance with Section 17.2b).
 - d) Sellers shall convene a general meeting of the Company granting discharge (*entlasten*) to the Leaving Function Holders substantially in the form as attached hereto as Exhibit 20.2d).
 - e) Sellers shall deliver to Purchaser the executed (i) Termination Agreement Andreas Gebhardt, (ii) Termination Agreement Markus Gebhardt, (iii) Termination Agreement Stephan Schulak, (iv) Employment Termination Agreements, (v) Consultancy Agreement Andreas Gebhardt, and (vi) Transitional Services Agreement.

- f) Sellers shall deliver evidence that the Company on the one hand and Markus Gebhardt on the other hand have entered into the IP Assignment Agreement substantially in the form as attached hereto as <u>Exhibit 20.2f</u>).
- g) Sellers shall deliver to Purchaser the Bring Down Certificate.
- h) Sellers shall deliver evidence that all outstanding amounts under the Other Reported Debt Agreements have been repaid.
- i) Sellers and Purchaser shall execute the Escrow Agreement and procure that (i) the acting notary has duly executed the Escrow Agreement and (ii) the Escrow Account has been established and is validly existing.
- j) Purchaser shall pay the Facilities Repayment Amount to the relevant Facilities Repayment Account.
- k) Purchaser shall pay the Promissory Note Repayment Amount to the relevant Promissory Note Repayment Account.
- 1) Sellers and Purchaser shall enter into the Assumption Agreement.
- m) Each Seller, Andreas Gebhardt, Markus Gebhardt and Johann Vielberth shall deliver to Purchaser a general release of claims against the Target Group Companies substantially in the form attached as <u>Exhibit 20.2m</u>).
- n) The relevant Target Group Companies on the one hand and the relevant Seller Group Entities on the other hand enter into the ancillary agreements substantially in the form as attached as Exhibit 20.2n), and in case the attached agreement is in the form of a term sheet, the Parties enter into a supply which reflects the commercial terms agreed in the term sheet.
- o) Purchaser shall pay the Preliminary Purchase Price Company, the latter as specified in Section 7.2 in the way set forth in Section 7.2.
- p) Sellers shall deliver to Purchaser evidence that the Board of Directors (*Verwaltungsrat*) of the Company has adopted the resolution to establish and confirm an updated share register of the Company reflecting Purchaser as the sole shareholder of the Company.
- q) Sellers shall deliver to Purchaser evidence that the Leaving Function Holders have resigned or have been removed in accordance with Section 17.2a).
- 20.3 The Closing Actions referred to in Sections 20.2a) through 20.2i), 20.2m) and 20.2p) may be waived by the Purchaser only. The other Closing Actions may only be waived by mutual agreement of the Parties, such agreement to be made in the form of Section 36.1 Any waiver of a Closing Action may be made in full or in part. The last sentence of Section 19.3 shall apply mutatis mutandis.

20.4 Immediately after all Closing Actions have been duly taken or waived, the Parties shall execute a closing memorandum substantially in the form as attached hereto as Exhibit 20.4 (the "Closing Memorandum") in which the Parties jointly confirm that (i) the Closing Conditions have been duly fulfilled or waived, (ii) Purchaser has received Sellers' Clearance Confirmation, (iii) the Closing Actions have been duly taken or waived, and, therefore, and (iv) the Closing has occurred. The Parties shall submit a copy of the Closing Memorandum to the acting notary without undue delay following execution of the Closing Memorandum.

IX. Warranties

21. Warranties of Sellers

21.1 Seller Warranties

- 21.1.1 Sellers hereby warrant (*garantieren*) to Purchaser by way of an independent promise of guarantee (*selbständiges Garantieversprechen*) pursuant to Section 311 (1) BGB that the statements set forth in Exhibit 21.1.1-A (the "Seller Warranties") are, taking into account the facts and circumstances set forth in this Agreement, in particular in Exhibit 21.1.1-B (the "Disclosure Schedule"), correct as of the Signing Date and as of the Closing Date, unless a Seller Warranty is explicitly made as of a different date, in which case such Seller Warranty shall be correct only as of such different date.
- 21.1.2 No Seller Warranty shall qualify as a quality agreement within the meaning of Section 434 (2) sentence 1 no. 1, no. 3 and sentence 2 BGB (*Beschaffenheitsvereinbarung*) or as a guarantee of condition within the meaning of Sections 443, 444 BGB (*Garantie für die Beschaffenheit der Sache*).
- 21.1.3 The Seller Warranties set forth in Section 1 of Exhibit 21.1.1-A are referred to as "Fundamental Warranties".
- 21.1.4 Sellers will, not earlier than three Business Days prior to the Scheduled Closing Date, review and confirm (after having made inquiry with the Inquiry Persons) whether the Seller Warranties are still correct. Sellers will provide Purchaser on the Scheduled Closing Date with a confirmation (the "Bring Down Certificate") in which Sellers confirm those Seller Warranties that are, as of the date of the review pursuant to this Section 21.1.4, still correct and, to the extent that any Seller Warranties are, as of the date of the review pursuant to this Section 21.1.4 not still correct, disclose the incorrectness of the Seller Warranties in an amended Disclosure Schedule. Any disclosure in such Bring Down Certificate or amended Disclosure Schedule shall not be deemed to have cured or corrected any incorrectness of the Seller Warranties, including for purposes of (i) the indemnification rights contained in this Agreement, or (ii) the Purchaser's Remedies.

21.2 Sellers' Knowledge

If any Seller Warranty refers to "Sellers' Knowledge", such term shall mean the actual knowledge (*positive Kenntnis*), as of the Signing Date, of any of the persons listed in <u>Exhibit 21.2-A</u> (the "Relevant Persons") and the knowledge such persons could have had as of the Signing Date had they diligently investigated and reviewed the relevant matter, including by making due inquiry with the persons listed in <u>Exhibit 21.2-B</u> (the "Inquiry Persons").

22. Warranties of Purchaser

22.1 General Principle

- 22.1.1 Purchaser hereby warrants (*garantieren*) to Sellers by way of an independent promise of guarantee (*selbständiges Garantieversprechen*) pursuant to Section 311 (1) BGB that the statements in Section 22.2 are correct as of the Signing Date unless a statement is explicitly made as of a different date, in which case such statement shall be correct only as of such different date.
- 22.1.2 None of the statements in Section 22.2 shall qualify as a quality agreement within the meaning of Section 434 (2) sentence 1 no. 1, no. 3 and sentence 2 BGB (*Beschaffenheitsvereinbarung*) or as a guarantee of condition within the meaning of Sections 443, 444 BGB (*Garantie für die Beschaffenheit der Sache*).

22.2 Individual Warranty Statements of Purchaser

- 22.2.1 As of the Signing Date and as of the Closing Date, Purchaser is a limited liability company (*Gesellschaft mit beschränkter Haftung*), duly established and validly existing under the laws of Germany.
- 22.2.2 As of the Signing Date and as of the Closing Date, Purchaser has all corporate power and authority to execute and perform this Agreement and to effect the transactions contemplated therein.
- 22.2.3 There is no legal or administrative proceeding pending against Purchaser or threatened in writing against Purchaser before any court, arbitral tribunal or public authority that in any manner challenges or seeks to prevent, alter or materially delay the Transaction.
- 22.2.4 No insolvency proceedings (*Insolvenzverfahren*) or similar proceedings under applicable laws are pending regarding Purchaser and Purchaser is not required under applicable laws to file for any such proceeding.

X. Purchaser's Remedies; Claim Procedures; Sellers' Remedies

23. Purchaser's Remedies

23.1 Definition and Scope

- 23.1.1 If and to the extent any of the Seller Warranties is incorrect, Purchaser may, at Purchaser's election, request from Sellers that Sellers (x) put Purchaser or, at Purchaser's election, the relevant Target Group Company into the position it would be in if the Seller Warranty had not been incorrect or, if the relevant Seller is unable to achieve this position within 30 Business Days after receipt of a Purchaser Claim Notice, (y) compensate Purchaser or, at Purchaser's election, compensate the relevant Target Group Company by payment in cash for all Losses incurred by Purchaser or the relevant Target Group Company as a result of the incorrectness of the Seller Warranty. Claims of Purchaser pursuant to this Section 23.1.1 in connection with an incorrect Seller Warranty shall be referred to as "Purchaser's Warranty Claims".
- 23.1.2 "Losses" means losses (*Schäden*) within the meaning of Sections 249 et seq. BGB, including (i) indirect or consequential damages (*mittelbare Schäden oder sonstige Folgeschäden*) and (ii) lost profits (*entgangener Gewinn*), in each case of (i) and (ii) if and to the extent reasonably foreseeable (*vernünftigerweise vorhersehbar*), but excluding (i) any reduction in value (*Minderung*) of the Target Group Companies beyond the actual loss incurred, (ii) any further consequential damages (*Folgeschäden*), indirect damages (*mittelbare Schäden*) or lost profits (*entgangener Gewinn*), (iii) frustrated expenses (*frustrierte Aufwendungen*) and (iv) internal expenses and costs (but including, for the avoidance of doubt, any Taxes falling due in connection with a payment received from Sellers in connection with any claim).

23.2 Exclusion of Sellers' Liability

Sellers shall not be liable for any Purchaser's Warranty Claim if and to the extent that:

- a) the facts or circumstances giving rise to the claim have been specifically taken into account as a liability (*Verbindlichkeit*) or provision (*Rückstellung*) in the Final Effective Date Adjustment Statement or otherwise as debt-like items and have actually reduced the Purchase Price;
- b) Purchaser or any Target Group Company has actually received recovery (net of the costs and expenses of such recovery) from a third party with respect to such Purchaser Warranty Claim (with the Purchaser hereby undertaking to pursue any possible claim against third parties to the extent commercially reasonable provided that the foregoing shall not obligate Purchaser to initiate or maintain any lawsuit or arbitration or use such efforts if Purchaser reasonably believes such efforts would be detrimental to or injure Purchaser's reputation or future business prospects in any material respect, in which case the Purchaser shall assign any such claim to the Sellers);

- c) Purchaser or, after the Effective Date, any Target Group Company has caused the facts or circumstances giving rise to the claim;
- d) the Purchaser's Warranty Claim results from or is increased by any change in any law or regulation in writing which has occurred after Closing;
- e) any Losses have only occurred because of Purchaser's breach against any obligations under Sections 24.1 and 25.2; or
- the facts or circumstances giving rise to the claim were disclosed, (i) in this Agreement, or (ii) as of 5:00 p.m. Central Europe Time 28/4/2022 (the "**Data Room Expiry**"), as part of the written information (for the avoidance of doubt, including any Q&A responses until the Data Room Expiry) of the virtual data room operated by Datasite under Project Astute for the purpose of the Transaction (collectively the "**Data Room Documents**"), provided that the disclosure has been made (i) in sufficient detail and specificity to allow a reasonable buyer to identify the factual, legal and economic relevance of such facts and circumstances and to make a reasonably informed assessment in this regard and (ii) in a folder of the virtual data room that would reasonably be expected to contain the disclosure of such facts or circumstances, it being understood that in any case any facts and circumstances made under and in connection with the documents and information included in the Data Room Documents as set forth in Exhibit 23.2f) shall be deemed as not disclosed. This Section 23.2f) shall not apply to Purchaser's Warranty Claims relating to the incorrectness of Fundamental Warranties.

A complete set of electronic copies of the Data Room Documents (buyside view) has been saved on each of four storage devices one of which will be provided to each of Purchaser and each Seller and one of which shall, immediately after the execution of this Agreement, be handed over by the Parties to the acting notary together with a letter from Datasite confirming the completeness of the data saved on the storage devices at the time of Data Room Expiry and clarifying language as to which documents have been uploaded or modified after the Data Room Expiry and accordingly are not deemed to be disclosed pursuant to Section 23.2f)(ii). The notary is hereby instructed (i) to keep the storage device and the confirmation letter from Datasite in safe custody for five (5) years following the Closing Date, provided, however, that, if court or arbitration proceedings pursuant to Section 40.7 are still pending on such date, the term of custody shall, upon any Party's written request to the notary, be extended until three (3) months after such proceedings have been concluded, and (ii) to provide the storage device and the confirmation letter to a court or arbitral tribunal upon written request of such court or arbitral tribunal or Purchaser, on one hand, and the Sellers, on the other hand (in which case the notary shall retain a copy and make customary arrangements with the court or arbitral tribunal to ensure that the original is returned to the notary as soon as possible). After lapse of that period the notary is entitled to destroy the Data Room Documents without further notice. The notary accepts this instruction.

24. Claim Procedures

24.1 Purchaser Claim Notice; Information Rights of Seller

- 24.1.1 Purchaser shall, without prejudice to any further claims of Purchaser in connection with the subject matter, give Sellers notice of any alleged Purchaser's Warranty Claim or any other alleged claim of Purchaser under this Agreement (including any claim under any indemnity) as soon as reasonably practicable, but in any event within forty (40) Business Day after Purchaser has gained positive knowledge of facts or circumstances that are likely to form the basis of such claim, stating in the notice in reasonable detail the then available facts and circumstances and the amount involved, to the extent that such amount has been determined by Purchaser at the time when the notice is given (the "Purchaser Claim Notice").
- 24.1.2 Purchaser shall give, and after Closing shall cause the Target Group Companies to give, in each case as soon as reasonably practicable, such information to Sellers as reasonably requested by Sellers to investigate the relevant facts and circumstances alleged to give rise to the claim notified. Cooperation under this Section 24.1.2 shall (i) be subject to appropriate measures to safeguard the confidentiality of the information given to Sellers and (ii) not be required to the extent that this would (a) unreasonably interfere with any attorney-client privilege or similar privilege of Purchaser and/or the Target Group Companies or (b) unreasonably interfere in any material respect with the business or operations of Purchaser or the Target Group Companies.

24.2 Third Party Claim Procedure

- 24.2.1 If a third party asserts or threatens to assert any claim against Purchaser or any Target Group Company, or if Purchaser or any Target Group Company is subjected to any audit or examination by any public authority (other than a Taxing Authority) or any public authority (other than a Taxing Authority) issues or threatens to issue a decision, order or decree against Purchaser or any Target Group Company, in each case based on facts or circumstances that are likely to give rise to a Purchaser's Warranty Claim (each a "Third Party Claim"), Purchaser shall, to the extent legally and practically possible, give Sellers notice of such Third Party Claim within fifteen (15) Business Days (unless a shorter time period applies under the Third Party Claim) after Purchaser has gained positive knowledge of the Third Party Claim. Section 24.1.2 shall apply accordingly.
- 24.2.2 Sellers shall be entitled to decide within four weeks after receipt of the notice of the Third Party Claim, to assume the defense against the Third Party Claim and to take all measures on behalf of the relevant Target Group Company or Purchaser which Sellers deem necessary or appropriate in connection with the defense against the Third Party Claim (including the assertion of counterclaims against the respective third party). In doing so, Sellers shall give due consideration to the legitimate interests of the Target Group Companies or Purchaser, as the case may be, and shall keep the costs in reasonable relation to the possible Losses. Sellers shall closely coordinate their defense with Purchaser and Purchaser shall cooperate with Sellers as reasonably requested by Sellers for the purpose of defending the Third Party Claim. If the relevant Third Party Claim proves to be justified by binding decision or written agreement after Sellers have assumed the

defense of the Third Party Claim, the underlying facts as determined in the relevant decision shall be deemed to be acknowledged in principle (*dem Grunde nach anerkannt*) in the relationship between Sellers and Purchaser and Sellers shall bear all costs and expenses incurred by them in connection with the defense against a third party claim.

- 24.2.3 If Sellers decide not to assume the defense against the Third-Party Claim, Purchaser shall be entitled to take any and all measures that Purchaser deems necessary in connection with the defense against the Third Party Claim, cooperating with Sellers for the purpose of defending the Third Party Claim. If the relevant Third Party Claim proves to be justified (*begründet*) or if other liabilities of the Target Group Companies or Purchaser are created, for example by the conclusion of court settlements, Sellers shall bear all reasonable costs and expenses incurred by the Target Group Companies or Purchaser in connection with the defense against a Third Party Claim.
- 24.2.4 Sections 24.2.2 and 24.2.3 shall not apply if the Third Party Claim results in a claim of the Purchaser under the W&I Insurance.
- 24.2.5 Purchaser shall be under no obligation under this Section 24.2 to perform or omit any act, or to cause any Target Group Company to perform or omit any act, if such performance or omission (as the case may be) may jeopardize Purchaser's coverage under the W&I Insurance and the non-observance of Sellers' rights under this Section 24.2 in such case shall not result in any adverse consequences to Purchaser's rights and claims under this Agreement, if and to the extent Purchaser's rights and claims are covered under the W&I Insurance.

24.3 Further Provisions regarding Seller Warranties

- 24.3.1 Sellers do not make any representations, warranties or guarantees regarding the Shares or any of the Target Group Companies other than the Seller Warranties as explicitly set forth in and based on the terms of this Agreement (in particular no guarantees with respect to any projections, estimates or budgets delivered or made available to Purchaser or its representatives of future revenues, future results of operations, future cash flows, future financial condition or the future business operations of the Company). Purchaser hereby expressly confirms and agrees to acquire the Shares and the Target Group Companies based upon Purchaser's own inspection, examination and evaluation/assessment with respect thereto, including its due diligence investigation, without reliance upon any express or implied representations, warranties or guarantees of any nature made by any Seller except for the Seller Warranties.
- 24.3.2 The Parties agree that the rights and remedies which the Purchaser may have with respect to any breach of Seller Warranties and Losses under Section 26 (*Indemnities*) are limited to the rights and remedies explicitly contained herein and any and all other rights and remedies of any legal nature are excluded. In particular, without limiting the generality of the foregoing, the Purchaser hereby waives any claims under statutory representations and warranties (sections 434 et seq. BGB), irrespective of whether any defects (*Mängel*) exist on the Signing Date or may arise in the period between the Signing Date and the Closing Date. In addition, any claims of the Purchaser related to statutory contractual or pre-contractual obligations (sections 280 to 282, 311 para. 2

and 3 BGB), frustration of contract (*Störung der Geschäftsgrundlage*, section 313 BGB) or tort (sections 823 et seqq. BGB) are excluded to the largest extent legally permissible. The Purchaser shall not have any right to rescind, cancel, contest or otherwise terminate this Agreement or exercise any right or remedy which would have a similar effect except as otherwise explicitly stated in this Agreement.

XI. Taxes

25. Taxes

25.1 Definitions

In this Agreement, the following terms shall have the following meanings:

"Indemnifiable Tax" means any Tax imposed on any Target Group Company by any Taxing

Authority and relating to the time period ending on or before the Effective Date (for the avoidance of doubt, except for any transfer taxes imposed in

connection with the transactions contemplated herein).

"Tax" means any tax within the meaning of Section 3 AO and equivalent taxes

under the laws of any other jurisdiction (including any withholding tax), customs duties and social security contributions, obligations to repay unlawfully granted state aid, administrative fines and any other public levies, together in each case with any interest, penalty, fine or addition thereto and irrespective of whether owed by the respective Target Group Company as a Tax payer, due as a secondary liability, contractually or otherwise; deferred

taxes shall not fall within the definition of Taxes.

"Taxing Authority" means any competent public authority in charge of imposing or collecting

any Tax.

"Tax Indemnification Claim" means any claim of Purchaser under Section 25.2.

25.2 Tax indemnification

25.2.1 Sellers shall pay to Purchaser

a) an amount equal to any (internal and external) expenses incurred by a Target Group Company for the preparation of any transfer pricing documentation (however in no event exceeding an aggregate amount \$ 100,000 for the Target Group as a whole) that the relevant Target Group Company is legally required to prepare and that has not been properly

- prepared before the Effective Date for any transactions relating to any time period ending on or before the Effective Date for which the statutory prescription period (*Festsetzungsverjährung*) has not yet expired;
- b) an amount equal to any Indemnifiable Tax incurred by AF&R Mexico, on or before the Effective Date, in connection with any certificates of origin issued by any of the Target Group Companies to obtain preferential tariff treatment, not complying with applicable trade regulations and laws, as well as any (internal and external) expenses; and
- c) an amount equal to any Indemnifiable Tax, provided that the Tax Indemnification Claim shall be reduced if and to the extent:
 - aa) the respective Indemnifiable Tax has been paid up to and including the Effective Date;
 - bb) a specific liability or provision for the relevant Indemnifiable Tax is included in the Effective Date Financial Debt Company and has actually reduced the Purchase Price; or
 - any Target Group Company has within seven (7) years after the Effective Date received any casheffective Tax benefit resulting from the Indemnifiable Tax for periods beginning after the Effective
 Date in particular due to the lengthening of any amortization or depreciation periods, a step-up in
 the Tax basis of assets, the non-recognition of liabilities or provisions (*Phasenverschiebung*) or a
 transfer pricing adjustment (herein collectively "**Tax Benefits**"), provided that the Tax Benefit shall
 be discounted at a rate of three point five per cent (3.5%) p.a. as from the Effective Date.
- 25.2.2 Any amounts payable to Purchaser under this Section 25.2 shall be payable and due within five (5) Business Days prior to the due date of the respective Tax payment to be made by the relevant Target Group Company.
- 25.2.3 After the Effective Date, Purchaser shall pay to Sellers an amount equal to any refund of any Taxes actually obtained by any Target Group Company after the Closing Date to the extent they relate to any period prior to the Effective Date, provided that (i) Sections 25.2.1c)aa) to cc) shall apply mutatis mutandis and (ii) no refund or adjustment shall be payable in respect of reimbursements paid to AF&R Mexico in respect of any VAT Taxes, subject to the next sentence. The Purchaser agrees that once the VAT suspension is fully remediated and Purchaser has reconciled the VAT payments (both pre and post Closing) against the reimbursement for those payments Purchaser will pay Sellers any excess. Fully remediated means, that all of the steps to remediate the reporting system and correct *pedimentos* are completed, the VAT suspension has been lifted or reinstated and the relevant VAT reimbursements have been received. Any amounts payable to the Sellers under this Section 25.2.3 shall be payable and due within five (5) Business Days after receipt of the respective Tax repayment by the relevant Target Group Company.

25.3 Cooperation on Tax Matters

- 25.3.1 As soon as reasonably practicable but no later than five (5) Business Days after the Signing Date, Sellers shall provide Purchaser with the relevant information in order to prepare the drafts of any real estate transfer tax notifications which have to be made by Purchaser or any of the Target Group Companies with respect to the execution or consummation of this Agreement.
- 25.3.2 After the Effective Date, the Purchaser and the Sellers shall reasonably cooperate, and shall cause their representatives to reasonably cooperate, with each other in connection with all Tax matters relating to any Taxes payable by the Target Group Companies for any period ending on or before the Effective Date, including the preparation and filing of any Tax return or the conduct of any Tax audits, investigations or other proceedings. Purchaser shall ensure, as legally permissibly that the Tax returns (*Steuererklärung*) of the Target Group Companies for any periods up to and including the Effective Date will be duly prepared and filed by the Target Group Companies. The cooperation between the Purchaser and the Sellers shall include the making available by each such Party to the respective other Parties of all books and records and the assistance of all officers and employees, to the extent reasonably required in connection with such Tax matters. The Purchaser shall notify the Sellers of any proposed audit or assessment relating to any Tax with respect to which the Sellers may reasonably expect to incur any liability hereunder. Purchaser shall, however, be under no obligation under this Section 25.3 to perform or omit any act, or to cause any Target Group Company to perform or omit any act, if such performance or omission (as the case may be) may jeopardize Purchaser's coverage under the W&I Insurance.
 - Following the Closing Date, the Purchaser shall as soon as reasonably practicable apply for a prompt company audit of the Target Group Companies (*Betriebsprüfung*).
- 25.3.3 Without limiting the generality of the other provisions of this Section 25.3, after the Effective Date, the Purchaser may cause the Target Group Companies to amend all open U.S. Tax Returns of Alfmeier Corporation to include the U.S. Internal Revenue Service Form 5471 with respect to each of the Subsidiaries, and the Sellers shall reimburse the Purchaser for the out-of-pocket costs of such amendments in an amount not to exceed EUR 135,000.
- 25.3.4 Without limiting the generality of the other provisions of this Section 26.3, after the Effective Date, the Purchaser may cause AF&R to file any application, or otherwise voluntarily undertake to participate in any program of voluntary disclosure, for amnesty or other relief from penalty for AF&R's failure to report or pay (i) income and/or franchise Taxes in California and Texas and (ii) gross receipts Tax in Ohio and Tennessee. Sellers shall reimburse AF&R and/or the Purchaser for all reasonable and documented out-of-pocket third-party fees and expenses paid by AF&R or the Purchaser in connection with preparing, filing and negotiating any such voluntary disclosures and related submissions in an amount not to exceed EUR 30,000. Following the execution of any voluntary disclosure agreement or receipt of a certificate (such agreement or certificate, a "VDA"), the amount of any sales Tax (including any interest, penalties or additions to such Tax) attributable to the time period ending on or before the Effective Date with respect to the VDA shall be treated as an Indemnifiable Tax for purposes of this Section 26.

- 25.3.5 With respect to AF&R Mexico's self-correction process regarding its VAT/IMMEX/PROSEC compliance, Sellers agrees to cooperate in full with any information request from Purchaser with respect to the status of the self-correction process, including but not limited to, pedimentos review, VAT calculations, Annex 24 files, Bill of Materials, VAT payments, Annex 30 reports, communication with the Mexican Taxing Authorities, status update meetings, amended Tax returns and any other information deemed necessary to review by the Sellers. Notwithstanding the provisions of this section, Sellers agrees to cause AF&R and AF&R Mexico to provide full copies of any documentation filed to the relevant Mexican Taxing Authorities within five (5) business days after its filing date.
- 25.3.6 Any failure of Purchaser to comply with any of its information and cooperation obligations set forth in Section 25.3.2 shall release Sellers from their respective indemnification obligation under this Section 25 if and to the extent that Sellers prove that they are prejudiced by such failure, unless Purchaser can demonstrate that the relevant Indemnifiable Tax assessment has been in compliance with applicable Tax law. There is a rebuttable presumption (widerlegbare Vermutung) that Sellers are prejudiced if, in breach of Section 25.3.2, (i) Sellers have not been notified before the expiration date of the relevant remedy of a relevant Tax assessment received by any of the Target Group Companies after the Closing Date or (ii) a specific remedy explicitly requested to be filed against a relevant Tax assessment has not been filed.

25.4 As-If Assessment

25.4.1 In the event Taxes related to tax assessment periods beginning before and ending after the Effective Date ("Straddle Tax Period"), the portion of Taxes related to the period prior to and including the Effective Date and the portion of Taxes related to the period beginning after the Effective Date shall be determined as if the Straddle Tax Period were a tax assessment period which ends on the Effective Date.

25.5 Time Limitations

Claims under this Section 25 shall not be time-barred before the expiration of six (6) months after the statutory prescription (*Festsetzungsverjährung*) of the relevant Tax but in no event earlier than six (6) months after the Effective Date.

25.6 Treatment of Payments

All payments to be made by Sellers to Purchaser or by Purchaser to Sellers based on or pursuant to Sections 21 to 23 or 25 shall constitute a reduction or an increase of the Company Purchase Price, as the case may be.

25.7 Tax Elections

Upon reasonable advance written notice from Purchaser, Sellers shall use commercially reasonable efforts to achieve, together with the Target Group Companies and Purchaser, AF&R

to make the alternative election pursuant to Section 6226(a) of the Code (and any comparable election for state Tax laws) with respect to all taxable periods of the Company prior to the Closing Date, after receipt of notice of final partnership adjustment from the U.S. Internal Revenue Service or another Taxing Authority.

25.8 PRC Bulletin 7 Tax

- 25.8.1 The Parties acknowledge and agree that reporting will be made with the applicable Taxing Authorities of the Peoples Republic of China ("PRC") in connection with Article 9 of Bulletin 7 with respect to the Transactions. Accordingly, promptly after the Signing Date, the Purchaser shall, and no later than thirty (30) days after the Signing Date, prepare and, upon prior written consent of the Sellers (not to be unreasonably withheld or delayed) file all such information and documents as required under Article 9 and Bulletin 7 (the "Bulletin 7 Reporting") with respect to the Transaction (and, in particular, with respect to the indirect share transfer of Alfmeier Automotive Systems (Shanghai) Co., Ltd. under this Agreement); provided, that such information and documents shall be prepared solely on the basis of a non-taxable position and no other Tax Return shall be filed, nor any Taxes paid, in connection with Bulletin 7 except as otherwise set forth in this Section 25.8).
- 25.8.2 Upon the Purchaser's request following the Signing Date, the Seller shall provide, and/or cause the Company to provide, to the Purchaser all information, documents and assistance reasonably necessary in connection with the Bulletin 7 Reporting to be made. The Purchaser shall, no less than ten (10) days before it submits the Bulletin 7 Reporting to any PRC Taxing Authority, (i) confer with the Sellers in good faith regarding the timing and venue of its proposed filings, (ii) provide the Sellers with copies of all Bulletin 7 Reporting documents it proposes to submit, (iii) consider in good faith any reasonable comments raised by the Sellers thereon that are provided to the Purchaser in writing no more than seven (7) days after the date such Bulletin 7 Reporting documents are provided to the Purchaser, (iv) include explicit representations in the Bulletin 7 Reporting documents that the Sellers, instead of the Purchaser, are solely responsible for the associated Tax filings and liabilities, if applicable, in connection with the indirect share transfer of Alfmeier China under this Agreement, and will file any required Tax Returns and settle relevant Taxes pursuant to Bulletin 7, and (v) promptly after the filing of the Bulletin 7 Reporting, deliver to the Sellers a copy of the submitted version of the Bulletin 7 Reporting documents and an acknowledgement or receipt in respect of the filing issued by the appropriate PRC Taxing Authority or a duplicate of the submitted version of the Bulletin 7 Reporting documents. Any filing requires the prior written consent of the Sellers (not to be unreasonably withheld or delayed).
- 25.8.3 Where applicable, the Sellers shall file required Tax Returns and pay or cause to be paid the Tax amount due with the appropriate PRC Taxing Authorities and other Governmental Entities in connection with the transfer of Shares under this Agreement before the respective filing and payment due dates as specified in Bulletin 7 (the "Bulletin 7 Taxes"), including any amended or supplemental Tax Returns that may be required. The Sellers shall provide Purchaser with a copy

of such Tax Returns (including amended or supplemental Tax Returns) at least ten (10) days before submitting the Tax Returns to the appropriate PRC Taxing Authority, and shall consider in good faith any reasonable comments raised by the Purchaser thereon that are provided to the Sellers in writing no more than seven (7) days after the date such Tax Returns are provided to the Purchaser. Notwithstanding the foregoing, upon the Sellers or the Purchaser receiving a payment or assessment notice issued by relevant PRC Taxing Authorities (a "PRC Payment Notice") for any PRC Bulletin 7 Taxes with respect to Alfmeier China, the Party that received the PRC Payment Notice shall deliver a written notice to the other Party with a copy of the PRC Payment Notice, and the Sellers shall pay or cause to be paid the amount shown as due and payable on the PRC Payment Notice to the relevant PRC Taxing Authorities in settlement of such Bulletin 7 Taxes in accordance with the PRC Payment Notice and applicable PRC Tax laws. The Sellers shall provide to the Purchaser copies of the Tax payment receipt from the competent PRC Taxing Authorities indicating that the Sellers or one or more of their Affiliates has paid the necessary Bulletin 7 Tax amount.

- 25.8.4 If the Sellers or the Purchaser receive any other written communication of any PRC Governmental Authority in relation to the transactions contemplated hereunder and/or Bulletin 7 Reporting, such Party shall promptly forward the communication to the other Party. Upon the Sellers' request following the execution of this Agreement, the Purchaser shall, and following the Closing shall cause the relevant Target Group Companies to, provide to the Sellers all information, documents and assistance reasonably necessary in connection with any Tax filings to be made in connection with Bulletin 7 Taxes.
- 25.8.5 The Sellers and the Purchaser shall use commercially reasonable efforts following the execution of this Agreement to discuss in good faith the anticipated allocation of the Purchase Price for purposes of Bulletin 7.

XII. Indemnities

26. Indemnities

26.1 Indemnity regarding data protection

After the Closing, Sellers shall indemnify (*freistellen*) Purchaser and the Target Group Companies from and against any Losses which have been incurred from and after the Closing Date by any of them and relating to the time period prior to and including the Closing Date out of or in connection with the matters and/or facts set out in <u>Exhibit 26.1</u>.

26.2 Indemnity regarding intellectual property rights

After the Closing, Sellers shall indemnify (*freistellen*) Purchaser and the Target Group Companies from any Losses which have been incurred from and after the Closing Date by any of them out of or in connection with the matters and/or facts set out in Exhibit 26.2.

26.3 Indemnity regarding employment matters

After the Closing, Sellers shall indemnify (*freistellen*) Purchaser and the Target Group Companies from any Losses which have been incurred from and after the Closing Date by any of them out of or in connection with the matters and/or facts set out in Exhibit 26.3.

- 26.4 Indemnity regarding environmental matters
- 26.4.1 After the Closing, Sellers shall indemnify (*freistellen*) Purchaser and the Target Group Companies from any Losses which have been incurred from and after the Closing Date by any of them out of or in connection with the matters and/or facts set out in Exhibit 26.4.1.
- 26.4.2 Any claims of Sellers pursuant to section 24 para. 2 German Federal Soil Protection Act (*Bundes-Bodenschutzgesetz*) and any similar statutory claims against Purchaser or the Target Group Companies shall be excluded.
- 26.5 Indemnity for undisclosed Debt and further Identified Risks

After the Closing, Sellers shall indemnify (*freistellen*) Purchaser and the Target Group Companies from any Losses which have been incurred from and after the Closing Date by any of them out of or in connection (i) with any Debt to the extent unpaid at Closing and not specifically accounted for (or exceeding the amount accounted therefore) in the Final Effective Date Adjustment Statement, (ii) any Seller Transaction Expenses or (iii) the matters and/or facts set out in Exhibit 26.5.

26.6 Cooperation

Section 24.2 shall apply to any claims under or in connection with this Section 26 (the "**Indemnity Claim**") mutatis mutandis.

- 26.7 Limitations
- 26.7.1 Indemnity Claims shall become time-barred (verjähren) three (3) years after the Effective Date.
- 26.7.2 The overall liability of Sellers for any Indemnity Claim shall be limited to an amount equal to fifteen percent (15%) of the Company Purchase Price.
- 26.7.3 Exclusion of Liability

Sellers shall not be liable for, and Purchaser shall not be entitled to an Indemnity Claim if and to the extent that

a) the matter to which the claim relates is specifically provided for in the Effective Date Adjustment Statement as a write-off (*Abschreibung*), value adjustment (*Wertberichtigung*), liability (*Verbindlichkeit*) or provision (*Rückstellung*) or is otherwise recognised as debt-like item, and has been explicitly taken into account in the calculation of the Purchase Price:

- b) Purchaser or any Target Group Company has actually received recovery from a third party with respect to such Indemnity Claim (with the Purchaser hereby undertaking to pursue any possible claim against third parties to the extent commercially reasonable possible);
- c) the Indemnity Claim arises, or the amount of the Indemnity Claim is increased, as a result of changes in law or federal supreme court jurisprudence which occurred after the Closing Date; or
- d) any Losses have occurred because of Purchaser's breach against its obligation under Section 24.2 with respect to the relevant Indemnity Claim, if unless, and to the extent, such Losses did not only occur because of such breach.

26.7.4 Treatment of Payments

Section 25.6 shall apply to any payments under this Section 26 mutatis mutandis.

XIII. Limitations

27. Limitation of Claims

27.1 Monetary Limitations / W&I Insurance

- 27.1.1 Sellers shall not be liable for any Purchaser's Warranty Claim if
 - a) such claim does not exceed EUR 100,000 (in words: Euro one hundred thousand) (the "**De Minimis Amount**"), provided, however, that where a series of related Purchaser's Warranty Claims arises out of a single event, cause or circumstance, such claims shall be aggregated and treated as a single Purchaser's Warranty Claim for the purpose of the foregoing; or
 - b) the aggregate amount of all Purchaser's Warranty Claims that exceed the De Minimis Amount does not exceed EUR 1,000,000 (in words: Euro one million) (the "**Threshold**").

If Sellers are liable for a Purchaser's Warranty Claim according to the foregoing, such liability shall, for the avoidance of doubt, not only be for the portion exceeding the De Minimis Amount and the Threshold but for the entire amount of such claim. The limitations of this Section 27.1.1 shall not apply to claims based on the incorrectness of any Fundamental Warranty.

27.1.2 Purchaser takes out a warranty & indemnity insurance under an insurance policy to be executed on or around the Signing Date, a final draft of which is attached hereto as Exhibit 27.1.2 (the "W&I Insurance" and the relevant insurance provider(s) as identified therein, the "Insurer"). The Purchaser undertakes to execute the W&I Insurance on or around the date hereof and not to subsequently change or terminate the W&I Insurance.

- 27.1.3 The liability of Sellers for any Purchaser's Warranty Claims to the extent covered by the W&I Insurance, except for claims (i) based on the incorrectness of the Seller Warranties (or parts thereof) set out in <u>Exhibit 27.1.3</u> and (ii) the incorrectness of Seller Warranties based on information set forth in the Bring Down Certificate (the "**Exempted Claims**"), shall be excluded and be EUR 0.00 (in words: zero Euro) (the "**W&I Liability Exclusion**"). The liability of Sellers for any Purchaser's Warranty Claims not (or partly not) covered by the W&I Insurance, except for breaches of the Seller Warranties in Exhibit 21.1.1-A under Section 3 shall be limited to the aggregate amount of fifteen percent (15%) of the Purchase Price.
- 27.1.4 The liability of Sellers for any Tax Indemnification Claims shall be excluded and be EUR 0.00 (in words: zero Euro) only if and to the extent such Tax Indemnification Claim is covered by the W&I Insurance. The liability of Sellers for any Tax Indemnification Claims not covered by the W&I Insurance shall be limited to the aggregate amount of fifteen percent (15%) of the Purchase Price.
- 27.1.5 The aggregate liability of Sellers for any and all Purchaser Warranty Claims, Tax Indemnification Claims and Indemnity Claims shall be limited to the amount of fifteen percent (15%) of the Purchase Price in total. The foregoing shall not apply (i) to any specific performance claims for transfer of title to the Gebhardt Shares and the Elber Shares, (ii) for breaches of the Seller Warranties in Exhibit 21.1.1-A under Section 3, and (iii) any claims of Purchaser under or in connection with Section 14 and the Carve-outs, which shall be limited in the aggregate to the Purchase Price.
- 27.1.6 To the extent any Purchaser's Warranty Claim or Tax Indemnification Claim is covered by the W&I Insurance (excluding in any case the Exempted Claims), Purchaser's sole recourse for any such claims shall be only against the Insurer. Any claim of the Purchaser under this Agreement which also constitutes a Purchaser's Warranty Claim or Tax Indemnification Claim (partially) covered by the W&I Insurance shall be asserted under the W&I Insurance against the Insurer first. Purchaser expressly acknowledges, and the other Parties agree, that the risk of non-implementation of the W&I Insurance as well as the validity and collectability risk in respect of the W&I Insurance and hence, the risk to successfully claim and/or recover from the Insurer any damages of the Purchaser in connection with any breaches of any of the Seller Warranties which would be covered in case of a valid W&I Insurance shall solely and irrevocably rest with the Purchaser. Under the W&I Insurance, the Insurer shall not be entitled to subrogate against Sellers except if the payment under the W&I Insurance or any loss as defined in the W&I Insurance arises out of Sellers' fraud (Arglist) or willful misconduct (Vorsatz).

27.2 Time Limitations

All claims of Purchaser arising under or in connection with this Agreement shall become time-barred (*verjähren*) three (3) years after the Effective Date, except for

a) claims resulting from a breach of any of the Seller Warranties, which shall become time-barred (*verjähren*) twenty-four (24) months after the Effective Date;

- b) claims resulting from a breach of any of the Fundamental Warranties or claims under or in connection with Section 14 (*Carve-out*), which shall become time-barred (*verjähren*) ten (10) years after the Effective Date;
- c) claims pursuant to Section 25 (Taxes), which shall become time-barred (*verjähren*) pursuant to Section 25.5;
- d) claims pursuant to Section 26 (Indemnities), which shall become time-barred (*verjähren*) pursuant to Section 26;
- e) all claims of Purchaser based on willful misconduct (*Vorsatz*) of obligations under this Agreement by the Sellers or fraud (*Arglist*), each of which shall be time-barred in accordance with the statutory limitation periods.

27.3 Mitigation of Damages

Section 254 BGB and the principles of setting-off any advantages (*Vorteilsausgleichung*) and mitigation of damages (*Schadensminderungspflicht*) shall apply.

28. Intentional or Fraudulent Behavior; Limitations

- 28.1 The limitations of liability in this Agreement, including limitations with regard to the term Losses and the limitations set forth in Section 21 and Sections 23 to 27, shall not apply in case of intentional or willful misconduct (*Vorsatz*) or fraud (*Arglist*) by a Seller with regard to such Seller only, in which case, statutory law shall apply. However, the Sellers' liability for fraud, willful deceit (*arglistige Täuschung*) and willful misconduct (*Vorsatz*) under and in connection with this Agreement shall also include the actions, measures and declarations of the persons listed in Exhibit 28.1, being understood that the actions, measures and declarations of the aforementioned persons are attributed to all Sellers. Sellers' liability for fraud, willful deceit (*arglistige Täuschung*) and willful misconduct (*Vorsatz*) by any other person (including vicarious agents (*Erfüllungsgehilfe*) of the Sellers within the meaning of section 278 BGB) is, under and in connection with this Agreement, comprehensively and for all purposes excluded, as far as legally permissible.
- 28.2 To the extent legally permissible and subject to Section 28.1 above, there shall be no imputation of willful intent, fraud, or fault to the Sellers or from one Seller to the other Seller (*keine Wissenszurechnung für Zwecke des Vorsatzes, keine Vorsatz-, Arglist- oder Verschuldenszurechnung*). To the extent legally permissible, the Purchaser shall have no claims or rights out of or in connection with this Agreement based on willful intent, fraud or fault of the Sellers if such claims were based on, or made in connection with, the attribution of knowledge to the Sellers or from one Seller to the other Sellers.

XIV. Post-Closing Covenants

29. Business Records; Post-Closing Assistance; Required Financial Statements

- 29.1 Sellers shall procure that, as of Closing, all books, accounts and other records relating to the Business of the Target Group Companies existing at the Target Group Companies (collectively the "Business Records") are available and accessible for Purchaser in the premises of the respective Target Group Company. If and to the extent it is revealed after Closing that Sellers are in possession of certain Business Records, Sellers shall provide Purchaser with the originals of such Business Records, or, to the extent the originals are not available, with copies thereof. The Parties will agree in good faith in the Interim Period on the transition of email inboxes, ensuring appropriate confidentially measures, as well as similar tasks in light of Andreas Gebhardt and Markus Gebhardt.
- 29.2 Until the expiry of the statutory book-keeping period, the Parties shall provide and shall procure that the Target Group Companies and the Seller Group Entities, respectively, as well as their representatives provide each other at reasonable request and at the requesting Party's cost with copies of all documents, information and assistance which are in the reasonable opinion of the requesting Party relevant for its legitimate business interests (including for any legal or administrative proceedings), provided that such exchange of information shall occur during regular business hours and under reasonable circumstances and in compliance with applicable laws and further provided that the requesting Party enters into a customary confidentiality undertaking in relation to any confidential information.
- 29.3 Sellers shall procure that those of the Seller Group Entities in which the Company sells its interest under the Carveout Agreements prepare and provide to Purchaser and the Company, without undue delay (*unverzüglich*) after the
 Effective Date, for purposes of inclusion into the consolidated financial statements of the Target Group, the financial
 reporting package for the last calendar month prior to the Effective Date (and any other calendar months prior to the
 Effective Date that have not yet been reported to Purchaser or another Target Group Company prior to the Effective
 Date, if any), in each case with the general content, in the same format and consistently applying the accounting
 principles as in the previous months.
- 29.4 From and after the Closing Date, Sellers shall, at Purchaser's written request, use their commercially reasonable efforts to procure to Purchaser as soon as practicable, but in no event later than 30 days after the Closing Date, all financial statements and financial and other information required to be included in, or required to facilitate the preparation of, pro forma or other financial statements (including applicable pro forma adjustments) and related notes, or financial information required to be included in (i) Parent's filings with the SEC on Form 8-K pursuant to Item 2.01 and 9.01 thereof and any financial statements required thereunder, in each case, in relation to the Transaction and/or the Target Group Companies (the "Form 8-K Information"), (ii) Parent's other reports to be filed or furnished with the SEC pursuant to the U.S. Securities Exchange Act of 1934, as amended, and the rules and regulations thereunder (the

"Exchange Act"), and any financial statements required thereunder (the "Exchange Act Information"), and (iii) any registration statement under the U.S. Securities Act of 1933, as amended, and the rules and regulations thereunder (the "Securities Act") prepared by Purchaser or any of its Affiliates, including financial statements prepared in accordance with Rule 3-05 of Regulation S-X for fiscal periods completed prior to the Closing Date and financial information through the Closing Date reasonably necessary to reflect the Transaction and/or the Target Group Companies in Parent's financial statements or to prepare pro forma financial statements reflecting the Transaction and/or the Target Group Companies (together with the Form 8-K Information and the Exchange Act Information, the "Required Financial Information"), but in all cases, only to the extent such Required Financial Information is not otherwise reasonably available to Parent or Purchaser and in each case subject to confidentiality requirements. For purposes of clarification, if financial statements covering different periods could constitute the Required Financial Information under the Exchange Act or the Securities Act and the rules and regulations promulgated thereunder, then Parent or Purchaser shall determine which such periods will constitute the Required Financial Statements for purposes of this Agreement.

- Sellers shall reasonably cooperate with Parent's and its Affiliates' preparation of all such pro forma or other financial statements or other information described above, and will use its commercially reasonable efforts to obtain (including by providing requested representations to such accountants), if so required under the Securities Act or the Exchange Act, an unqualified audit opinion from (i) each of the auditing firms set forth on Exhibit 30.5 or (ii) to the extent such auditing firms are not subject to and in compliance with the requirements of the Public Company Accounting Oversight Board (the "PCAOB"), another independent auditor selected by Parent, which firm shall be subject to and in compliance with the requirements of the PCAOB, with respect to such financial statements (which audit opinion shall comply with the requirements of the U.S. Securities and Exchange Commission (the "SEC")) and, in either case, the consent of such firm to the inclusion or incorporation by reference of their audit opinion in any registration statement filed by Parent or any of its Affiliates pursuant to the Securities Act (which audit opinion and consent shall also constitute Required Financial Information).
- 29.6 From and after the Closing, Sellers shall, at Parent's written request, use its commercially reasonable efforts (including by providing requested customary representations on a non-reliance basis to such accountants) to obtain (i) consents from the auditing firm determined in accordance with Section 30.5 to the inclusion of their audit opinion in registration statements filed by Parent or any of its Affiliates for so long as audited financial statements relating to the Target Group Companies and/or the Transaction are required to be included or incorporated by reference in such registration statements and (ii) such customary comfort letters from such auditing firm as the underwriters of any securities offering made pursuant to such registration statements may reasonably request. For the avoidance of doubt, it is understood and agreed that the Required Financial Information shall include combined audited financial statements of the Target Group Companies on a consolidated basis.

29.7 From and after the Closing Date, for a transition period, each Seller shall reasonably cooperate with Parent and its accountants and auditors and provide to Parent and its accountants and auditors, during normal business hours and upon reasonable prior written notice, but without unreasonably disrupting its business, access to such information, books and records related to the Company as Parent may reasonably request in connection with the preparation by Parent of historical financial statements related to the Target Group Companies as may be required to be included in any filing under the Exchange Act and the regulations promulgated thereunder, including Regulation S-X, to be reported on a current report on Form 8-K filed in connection with the transactions contemplated by this Agreement and any other filing as may be required under applicable law. Without limiting the foregoing, such cooperation shall include: (i) where appropriate, the signing of customary management representation letters on a non-reliance basis as are required in connection with such audit and (ii) as reasonably requested by Parent, access, during normal business hours and upon reasonable prior written notice, but without unreasonably disrupting its business, to appropriate individuals with knowledge of the historical financial information related to the the Target Group Companies for preparation of such financial statements. Parent shall reimburse each Seller for all out-of-pocket expenses and reasonable external costs incurred by such Seller or its Affiliates in connection with this Section 30.7 and Sections 30.4 through 30.6, and Parent shall be deemed an express third party beneficiary of this Section 30.7 and Sections 30.4 through 30.6.

30. Non-Compete; Non-Solicitation

- 30.1 Non Compete
- 30.1.1 Subject to Section 30.1.2, Sellers and Seller Guarantors shall not, and shall cause the Seller Group Entities not to, conduct a Competing Business in the territory of the countries listed in Exhibit 30.1.1 until the expiry of three (3) years from the Effective Date. "Competing Business" means the Business as conducted by the Target Group Companies on the date hereof. For the avoidance of doubt, the solicitation of customers from the Target Group Companies shall constitute the conduct of a Competing Business.
- 30.1.2 Nothing in this Section 30 shall prevent any Seller, Seller Guarantor or Seller Group Entity from:
 - a) maintaining and continuing the business activities set forth in Exhibit 30.1.2a); or
 - b) acquiring any participation in entities conducting a Competing Business not exceeding five percent (5%) of the share capital or voting rights.
- 30.2 Non-Solicitation

Until the expiry of two (2) years from the Effective Date, Sellers and Seller Guarantors shall not, and shall cause the Seller Group Entities not to, actively solicit the service or employment of any employee of a Target Group Company, provided that the foregoing shall not prevent Sellers, Seller Guarantors and other Seller Group Entities from hiring any such person that approaches

Sellers, Seller Guarantors or another Seller Group Entity on his or her own initiative or by responding to a general job advertisement or recruitment campaign that is not specifically targeted at persons engaged by the relevant Target Group Company. It is agreed that Nastasia Gläser, Julia Jürgensen, Ute Zauleck and Stephan Schulak will each start employment with a Seller Group Entity as of Closing, and, hence, this non-solicitation covenant does not apply to those individuals.

30.3 Consequences of Breaches

- 30.3.1 In the event of any breach of Section 30.1 or 30.2 by any Seller or any Seller Guarantor or the Affiliates or related persons of any of them, such Seller or Seller Guarantor shall be liable to pay a contractual penalty in the amount of EUR 250,000 per breach if such breach is not cured by such Seller or Seller Guarantor within five (5) Business Days after having been notified of such breach by Purchaser. In case of a continuous breach which is not stopped by any Seller within five (5) Business Days after having been notified of such breach by Purchaser, an additional contractual penalty of EUR 500,000 shall become due and payable every 14 days of such continued breach. Moreover, in such case the Sellers and Seller Guarantors shall indemnify and hold Purchaser harmless from any damage, cost or expense, including, without limitation, reasonable professional advisors' fees, resulting from such breach. Such indemnification shall not relieve the Sellers or Seller Guarantors from their obligations under the non-compete and the non-solicitation covenant.
- 30.3.2 The Parties agree that the remedies available to Purchaser under or pursuant to this Agreement are not sufficient to hold Purchaser and the Target Group Companies fully harmless against the detriments suffered from any breach, and that therefore Purchaser shall be entitled to enforce any claims for specific performance by the Sellers of the non-compete and/or non-solicitation obligation (*Unterlassungs- und Beseitigungsansprüche*) by injunctive relief (*einstweiliger Rechtsschutz*) without having to specifically establish irreparable harm or injury and without providing a bond or other collateral (*Sicherheitsleistung*).

31. Non-Disparagement

Until the expiry of five (5) years from the Effective Date, Sellers shall not, and shall cause the Seller Group Entities not to, make any comment or statement or engage in any other behaviour, directly or indirectly, that in any way disparages or could result in the disparagement of Purchaser or any Target Group Company. Notwithstanding the foregoing, nothing in this Section 31 shall prohibit Sellers or any Seller Group Entity from making truthful statements when required by law or binding administrative or court order.

XV. Miscellaneous

32. Withdrawal from this Agreement

- 32.1 Sellers jointly or the Purchaser shall, until the Closing has occurred, be entitled to withdraw (*zurücktreten*) from this Agreement by written notice to the respective other Party if
 - a) not all Closing Conditions have been fulfilled or waived by 31 December 2022 (the "Long-Stop Date"); or
 - b) it has become impossible to fulfil any of the Closing Conditions and such Closing Condition has not been waived within ten (10) Business Days after it has become impossible to be fulfilled,

provided that a Party shall not be entitled to base a withdrawal on one of the foregoing reasons if such Party is responsible for the non-fulfilment of, or the impossibility to fulfil, the relevant Closing Condition.

- 32.2 Sellers jointly or the Purchaser shall, until the Closing has occurred, be entitled to withdraw (*zurücktreten*) from this Agreement by written notice to the respective other Party if
 - a) not all Closing Actions have been taken or waived within ten (10) Business Days from the Scheduled Closing Date; or
 - b) it has become impossible to perform any of the Closing Actions and such Closing Action has not been waived within ten (10) Business Days after it has become impossible to be performed;

provided that a Party shall not be entitled to base a withdrawal on one of the foregoing reasons if such Party is responsible for the non-performance of, or the impossibility to perform, the relevant Closing Action due to a breach of its obligations hereunder in relation to the relevant Closing Action.

A withdrawal (*Rücktritt*) pursuant to this Section 32 shall not prejudice or limit any rights and claims (including those against any Seller Guarantor) that (i) the withdrawing Party may have due to the reason for the withdrawal, or (ii) any Party may have due to a breach of this Agreement by the other Party prior to such withdrawal (including damages due to the other Party's breach of obligations under Section 19 and/or Section 20.2). The provisions of this Section 32, Section 10 and Sections 33 through 40 shall survive a withdrawal.

33. Liability of Sellers

33.1 Up to the relevant Escrow Amounts which have not been released to the Sellers in accordance with Section 11.2 at the relevant point in time, Sellers shall be jointly and severally liable

(gesamtschuldnerische Haftung) for any and all Purchaser's claims under or in connection with this Agreement.

33.2 For Purchaser's claims under this Agreement exceeding the Escrow Amounts which have not been released to the Sellers in accordance with Section 11.2 at the relevant point in time, Sellers shall be severally liable (teilschuldnerische Haftung) pro rata to their holding of the Company Shares except for any claims relating to the shareholdings of any Seller in the Company which shall be given by each Seller as an individual debtor (Einzelschuldner) only with respect to the shares to be sold and transferred hereunder by such Seller.

34. Obligations of Seller Guarantors

- 34.1 Each of Andreas Gebhardt and Markus Gebhardt
- shall be jointly and severally liable (*gesamtschuldnerische Haftung*) with Gebhardt Holding for any and all obligations of Gebhardt Holding under and in connection with (i) any performance claims for transfer of title to the Gebhardt Shares, (ii) intentional or fraudulent behaviour of Gebhardt Holding (including by the persons listed in Exhibit 28.1 in accordance with Section 28.1), (iii) Section 14 and the Carve-Outs, (iv) Section 26 (Indemnities) and (v) the post-closing covenants under Sections 30 (Non-Compete; Non-Solicitation) and 31 (Non-Disparagement); and
- d) hereby undertakes to use all shareholder rights and all other influence as shareholders of Gebhardt Holding to ensure that Gebhardt Holding fully complies with its obligations under this Agreement.

34.2 Johann Vielberth

- a) shall be jointly and severally liable (*gesamtschuldnerische Haftung*) with ELBER for any and all obligations of ELBER under and in connection with (i) any performance claims for transfer of title to the ELBER Shares, (ii) intentional or fraudulent behavior of ELBER (including by the persons listed in <u>Exhibit 28.1</u> in accordance with Section 28.1) (iii) Section 14 and the Carve-Outs, (iv) Section 26 (Indemnities) and (v) the post-closing covenants under Sections 30 (Non-Compete; Non-Solicitation) and 31 (Non-Disparagement); and
- b) hereby undertakes to use all shareholder rights and all other influence as shareholder of ELBER to ensure that ELBER fully complies with its obligations under this Agreement.
- 34.3 Any joint and several liability of Andreas Gebhardt and Markus Gebhardt for obligations of Gebhardt Holding and/ or Johann Vielberth for obligations of ELBER, as the case may be, under this Section 34 only applies if and to the extent that the Purchaser has obtained a corresponding final court judgement (*rechtskräftiges gerichtliches Urteil*) or arbitral award (*Schiedsspruch*) against Gebhardt Holding and/or ELBER, as the case may be, and Gebhard Holding or ELBER, have not fullfiled their obligations as set forth therein. This Section 34.3 shall not apply in case

Gebhardt Holding and/or, ELBER, as the case may be, is insolvent or overindebted within the meaning of Section 17 et seq. Insolvency Code and a respective proceeding has been opened over the assets of Gebhardt Holding and/or ELBER, as the case may be, or rejected due to a lack of assets.

35. Confidentiality

- 35.1 For the purpose of this Agreement, "Confidential Information" shall mean (i) this Agreement, all contents thereof and the content of the negotiations relating to it, (ii) any information relating to the Parties, the Seller Guarantors and their Affiliates (excluding the Target Group Companies) provided by or on behalf of a Party or a Seller Guarantor in connection with this Agreement, (iii) any information provided by or on behalf of a Party or a Guarantor in connection with the performance of this Agreement and (iv) any information relating to the Target Group Companies, provided, however, that such last-mentioned information shall be Confidential Information for Purchaser only until the Effective Date and for Sellers only as from the Effective Date.
- 35.2 Each Party and each Seller Guarantor shall protect any Confidential Information against unauthorized access by third parties and/or accidental disclosure with the same degree of care as used with regard to its own confidential information of a similar nature and shall at least employ reasonable measures to keep the information secret (angemessene Geheimhaltungsmaβnahmen) as required by Section 2 no. 1b) GeschGehG. The Parties and the Seller Guarantors shall not disclose any Confidential Information unless and to the extent the disclosure is
 - a) (i) made to a Party's, Seller Guarantor's or any of the Target Group Companies' directors, officers, employees, professional advisors or agents (including auditors, lenders, banks and insurance carriers) or the agents' professional advisors, in each case having a need to know such information in connection with their function and/or engagement, (ii) required for the due fulfillment of the relevant Party's or Seller Guarantor's obligations under this Agreement, or (iii) required for the fulfillment of any information or reporting obligations towards direct or indirect shareholders of the relevant Party or Seller Guarantor, provided for all such cases ((i) through (iii)) that the receiving person/entity is bound to confidentiality obligations similar to and no less stringent than those set forth by this Section 35 (either contractually or under statutory secrecy obligations), and further provided that the disclosing Party is responsible for any breach of the confidentiality obligations by any person or entity to whom it has made available Confidential Information;
 - b) required by any applicable law, stock exchange regulation or public authority or applicable self-regulatory organization (including the SEC or Nasdaq);
 - c) required for the purpose of any arbitral or judicial proceedings arising out of this Agreement;

- d) limited to information that is generally known to the public other than as a result of a breach of a duty of confidentiality; or
- e) explicitly agreed between the Parties and the Guarantors.
- 35.3 The Parties agree that the non-disclosure agreement dated 3 December 2021 (the "NDA"), shall remain in force and effect and shall expire on the Effective Date, and that, from the Signing Date until the Effective Date, the provisions of the NDA shall apply in addition to the provisions of Sections 35.1 and 35.2 limited, however, to the extent required to comply with applicable law.
- 35.4 The provisions of this Section 35 shall be effective as from the Signing Date and shall expire five (5) years after the Effective Date.

36. Notices

36.1 If not required otherwise by mandatory law, all notices, legal declarations (*Willenserklärungen*) or other communications hereunder (including, for the avoidance of doubt, the making of any claims and the exercise of any other rights under or in connection with this Agreement) shall be made in writing, with personal signature (*eigenhändige Unterschrift*), and in the English language and shall be (i) sent by registered letter (*Einschreiben*), (ii) delivered personally, or (iii) sent by email as attached pdf-copy, in each case to the persons at the addresses set forth below, or to such other persons/addresses as may be designated by the respective Party or Seller Guarantor to the relevant other addressees in the same manner:

To Gebhardt Holding:

Gebhardt Holding GmbH Attn.: Andreas Gebhardt / Markus Gebhardt Industriestraße 5 91757 Treuchtlingen Germany Email:

with a copy to:

Germany

Attn.: Dr. Hubertus Schröder Alter Wall 32 20457 Hamburg Germany Email: and Attn.: Dr. Malte Schwertmann Feuerbachstraße 20a 91522 Ansbach

Email:

To ELBER:

ELBER GmbH

Attn.: Dr. Johann Vielberth Im Gewerbepark C 25 93059 Regensburg

Germany Email:

with a copy to:

Attn.: Dr. Hubertus Schröder

Alter Wall 32
20457 Hamburg
Germany
Email:

To Andreas Gebhardt:

Gebhardt Holding GmbH

Attn.: Andreas Gebhardt / Markus Gebhardt

Industriestraße 5 91757 Treuchtlingen

Germany Email:

with a copy to:

Attn.: Dr. Hubertus Schröder

Alter Wall 32 20457 Hamburg Germany

Email:

and

Attn.: Dr. Malte Schwertmann

Feuerbachstraße 20a 91522 Ansbach

Germany Email:

To Markus Gebhardt:

Gebhardt Holding GmbH

Attn.: Andreas Gebhardt / Markus Gebhardt

Industriestraße 5 91757 Treuchtlingen Germany Email:

with a copy to:

Attn.: Dr. Hubertus Schröder Alter Wall 32 20457 Hamburg Germany

Email: and

Attn.: Dr. Malte Schwertmann

Feuerbachstraße 20a 91522 Ansbach Germany Email:

To Johann Vielberth:

ELBER GmbH Attn.: Dr. Johann Vielberth Im Gewerbepark C 25 93059 Regensburg Germany Email:

with a copy to:

Attn.: Dr. Hubertus Schröder Alter Wall 32 20457 Hamburg Germany Email:

To Purchaser:

Gentherm GmbH Attn.: Thomas Stocker Rudolf-Diesel-Straße 12 85235 Odelzhausen Germany Email:

with a copy to:

Gentherm Incorporated Attn.: Wayne Kauffman, General Counsel 21680 Haggerty Road Northville, MI 48167 United States Email:

and

Gleiss Lutz Hootz Hirsch PartmbB Rechtanwälte, Steuerberater Attn.: Dr. Ralf Morshäuser Karl-Scharnagl-Ring 6 80539 Munich Germany Email:

and

Honigman LLP
Attn.: Michael Ben
Matthew VanWasshnova
660 Woodward Avenue
Detroit, MI 48226-3506
United States
Email:

36.2 The receipt of copies of notices, legal declarations and other communications hereunder by the Parties' or Seller Guarantors' advisors shall not constitute or substitute the receipt of such communication by the relevant Party or Seller Guarantor itself. Any notices, legal declarations and other communications hereunder shall be deemed received by a Party or Seller Guarantor regardless of whether any copy of such communication has been sent to or received by an advisor of such Party or Seller Guarantor.

37. Public Announcement

The Parties shall consult with each other before issuing any press release or otherwise making any public statement or making any other public (or non-confidential) disclosure (whether or not in response to an inquiry) regarding the terms of this Agreement and the transactions contemplated hereby, and no Party shall issue any such press release or make any such public statement or disclosure without the prior approval of the other Parties (which approval shall not be unreasonably withheld, delayed or conditioned). Notwithstanding the foregoing, the Parties

agree to the public announcement attached hereto as Exhibit 37, which may be released by any Party at any time after the date hereof. Nothing in this Section 37 shall limit the ability of Purchaser or its Affiliates (a) to issue any press release or make any public statement or any other public (or non-confidential) disclosure (whether or not in response to an inquiry) regarding the terms of this Agreement and the transactions contemplated hereby to the extent it determines that such press release, public statement or other public disclosure is required by applicable law, stock exchange regulation or public authority or applicable self-regulatory organization (including the SEC or Nasdaq), and (b) to conduct customary investor and analyst calls concerning the transactions contemplated hereby, to make internal announcements to their employees, and to make disclosures to potential sources of financing for the transactions contemplated hereby or to credit rating agencies. Sellers acknowledge and agree that Purchaser may file this Agreement and other documents described in this Agreement and summaries hereof and thereof, with the SEC and Nasdaq.

38. SAES Getters, Reimbursement of Duty Drawback

- 38.1 The Parties acknowledge the Alfmeier Target Group Companies' long-standing supply arrangement with SAES Getters SPA. During the three-month period after the Closing Date, Purchaser shall cause the Alfmeier Target Group Companies to work in good faith to discuss and negotiate with SAES Getters SPA the formalization of long-term supply agreement for SMA-wire, and the Sellers shall cooperate with the Alfmeier Target Group Companies in such efforts.
- 38.2 If, after the Closing Date, AF&R actually receives a drawback of duties paid prior to the Closing Date by AF&R and/ or AF&R Mexico on solenoids imported into Mexico prior to the Closing Date, Purchaser shall cause AF&R to pay (i) to Gebhardt Holding an amount equal to 60.68 percent of the amount actually received by AF&R in respect of such drawback, but in all cases, net of any reasonable and documented out-of-pocket costs and expenses (including, without limitation, applicable commissions and consultant payments) incurred by or on behalf of AF&R Mexico in the collection of such drawback (the "Net Drawback Amount") and (ii) to ELBER an amount equal to 39.32 percent of the Net Drawback Amount, in each case, within 30 days of the date on which AF&R Mexico actually receives such Net Drawback Amount. Notwithstanding the foregoing, the aggregate Net Drawback Amount shall not exceed more than EUR 601,000.

39. Costs and Expenses; Transfer Taxes

39.1 Each Party and each Seller Guarantor shall bear its own costs and expenses, including the fees of its advisors, incurred in connection with the preparation, negotiation, execution and performance of this Agreement. To the extent that any Target Group Company has incurred any Seller Transaction Expenses or incurs any Seller Transaction Expenses after the Signing Date, such Target Group Company hereby assigns to Sellers, and Sellers hereby assume from such Target Group Company, such Seller Transaction Expenses to the Sellers, and Sellers shall discharge such Seller Transaction Expenses in full.

39.2 Unless otherwise expressly specified herein, all notarial fees, registration fees, fees and charges in connection with any regulatory requirements (including any merger control filings, foreign investment control clearances and further governmental clearances) and any other similar fees, charges and costs as well as any sales, transfer or stamp taxes, or other similar charges payable in connection with the execution or performance of this Agreement shall be borne by Purchaser, except for any fees, charges or costs or any sales, transfer or stamp taxes, or other similar charges incurred in connection with the measures set forth in Section 14.1 and 14.2, which shall be borne by Sellers.

40. General Provisions

- 40.1 Purchaser shall be entitled to assign its rights and claims under this Agreement (i) to any of its Affiliates and (ii) as security (*Sicherungsabtretung*) to the lenders, agents or other finance parties under the Purchaser Facilities Agreement. Apart from that, no Party shall be entitled to assign any of its rights or claims hereunder without the prior consent of the respective other Party (such consent to be granted in the form of Section 36.1).
- 40.2 The Parties and the Seller Guarantors shall not be entitled to exercise any right to set-off, retention or other right to refuse performance (*Aufrechnung, Zurückbehaltung oder sonstige Leistungsverweigerungsrechte*) with respect to any obligation under or in connection with this Agreement, except in case the respective claim was expressly acknowledged (*anerkannt*) by the respective counterparty in the form of Section 36.1 or has been awarded in a legally binding (*rechtskräftig*) decision in principal proceedings (*im Hauptsacheverfahren*) by a competent court or arbitral tribunal.
- 40.3 Unless explicitly provided otherwise herein, this Agreement shall not grant any rights to, and is not intended to operate for the benefit of, any third parties, including the Seller Guarantors, and any Target Group Company (*kein echter Vertrag zugunsten Dritter*).
- 40.4 Unless expressly set forth otherwise herein, amendments to this Agreement (including to this Section 40.4) shall only be valid if explicitly referring to this Agreement and made in written form within the meaning of Section 126 (1) alt. 1 and (2) BGB. Section 127 (2) BGB shall not apply. Stricter requirements under mandatory law shall remain unaffected.
- 40.5 This Agreement (including all Annexes and Exhibits hereto) contains the entire agreement between the Parties and the Seller Guarantors with respect to the subject matter hereof and supersedes all prior agreements and understandings with respect thereto, except for the NDA with regard to which Section 35.3 shall apply.
- 40.6 This Agreement shall be governed by, and construed in accordance with, German law, excluding the German conflict of law rules and excluding the United Nations Convention on Contracts for the International Sale of Goods (CISG).

- 40.7 All disputes arising out of or in connection with this Agreement (including the breach, termination or validity thereof) shall be finally settled, without recourse to the ordinary courts of law, by arbitration in accordance with the Arbitration Rules of the German Arbitration Institute (*Deutsche Institution für Schiedsgerichtsbarkeit e.V.*), as amended from time to time. The arbitral tribunal shall be comprised of three (3) arbitrators. The seat of the arbitration shall be Munich, Germany. The language of the arbitration shall be English. Documents originating in the German language may be submitted in the German language without an English translation. The right to obtain injunctive relief (*vorläufigen Rechtsschutz*) before state courts shall not be excluded.
- 40.8 Should any provision of this Agreement be or become invalid or unenforceable, the other provisions of this Agreement shall remain in full force and effect. In such case, the Parties and the Guarantors shall agree on such valid and enforceable substitute provision(s) that correspond(s) as closely as possible with the intentions of the Parties and the Guarantors at the time of the conclusion of this Agreement. This shall apply mutatis mutandis in the event that this Agreement contains any unintended gaps (unbeabsichtigte Vertragslücken).

* * * **

ANNEX 2.1 (DEFINITIONS, LIST OF DEFINITIONS)

1. Definitions

In this Agreement, the following terms shall have the following meanings:

"2022 Business Plan" as defined in Section 25.1 of the Seller Warranties

"Affiliate" means any affiliate (verbundenes Unternehmen) within the meaning of Sections 15

et seq. AktG, however regardless of its qualification as undertaking

(Unternehmenseigenschaft).

"AF&R" means Alfmeier Friedrichs & Rath, LLC.

"AF&R Mexico" means Alfmeier Friedrichs & Rath de México, S.A. de C.V. (previously Alfmeier

de México, S.A. de C.V.)

"AktG" means the German Stock Corporation Act (Aktiengesetz).

"Alfmeier China" means Alfmeier Automotive Systems (Shanghai) Co., Ltd.

"Alfmeier CZ" means Alfmeier CZ s.r.o.

"AO" means the German Fiscal Code (*Abgabenordnung*). "Applied Subsidies" as defined in Section 16.1 of the Seller Warranties

"AWG" means the German Foreign Trade and Payments Act (*Außenwirtschaftsgesetz*).

"AWV" means the German Foreign Trade and Payments Regulation

(Außenwirtschaftsverordnung).

"BEIS" means the Department for Business, Energy and Industrial Strategy and the

secretary of state, as referred to in the NSIA2021.

"BIS" as defined in Section 21.5 of the Seller Warranties

"BGB" means the German Civil Code (Bürgerliches Gesetzbuch).

"Bulletin 7" means the Tax notice issued by the PRC State Administration of Taxation titled

the "State Administration of Taxation's Bulletin on Several Issues of Enterprise Income Tax on Income Arising from Indirect Transfers of Property by Non-resident Enterprises" (State Administration of Taxation Bulletin [2015] No. 7), as may be amended or supplemented from time to time, including any similar or replacement PRC Law on the Tax treatment of the offshore indirect transfer of "China Taxable Property" and including any similar Applicable Laws in the PRC against the

avoidance of PRC Tax.

"Business Day" means any day other than a Saturday or Sunday on which banks are open for

business in Munich, Germany.

"Business Information" as defined in Section 11.1 of the Seller Warranties

"CARES Act" means the U.S. Coronavirus Aid, Relief, and Economic Security Act of 2020, Pub.

L. 116-136, as amended.

"Code" means the U.S. Internal Revenue Code of 1986, as amended.

"Consolidated Financial Statements" as defined in Section 5.1 of the Seller Warranties

"Contract" means any indenture, mortgage, note, bond, financing commitment, loan

agreement, license, sublicense, contract, lease, sublease, agreement, instrument of record or other instrument or obligation, in any case, whether written or oral, but in each case which is legally binding on a Target Group Company, the Sellers or any

of Andreas Gebhardt, Markus Gebhardt or Johann Vielberth.

"Copyleft Software" as defined in Section 10.1a) of the Seller Warranties

"Debt" means any item which would have qualified as Effective Date Financial Debt

Company as of the Effective Date but has not been included in the Effective Date Adjustment Statement, including to the extent not fully captured therein, the Call

Price

"Employees" as defined in Section 14.1 of the Seller Warranties

"Employee Benefit Plans" as defined in Section 14.8 of the Seller Warranties

"Environmental Conta-mination" means any Hazardous Materials that exist in the soil, groundwater or surface water.

"Environmental Law" means any applicable law relating to or imposing liability for the protection,

pollution or contamination of the environment or the generation, use, treatment, storage, handling, transportation, disposal, clean-up or release of Hazardous

Materials.

"ERISA Affiliate"

means any Person that, together with any Target Group Company is (or at any relevant time has been or could be) considered a single employer under Section 414

of the Code.

"EStG" means the German Income Tax Act (Einkommensteuergesetz).

"EUR" means Euro or Euros.

"Exchange Rate" means the exchange rate of the relevant currency versus the Euro as determined

by the European Central Bank on its website at https://www.ecb.europa.eu/stats/policy_and_exchange_rates/euro_reference_exchange_rates/html/index.en.html as published (i) for the relevant date if the relevant date is a Business Day, or (ii) for the last Business Day prior to the relevant date if the relevant date is not a Business

Day.

"GeschGehG" means the German Code on the Protection of Trade Secrets (Gesetz zum Schutz von

Geschäftsgeheimnissen).

"GewStG" means the German Trade Tax Act (Gewerbesteuergesetz).

"Governmental Entity" means any (a) foreign, domestic, federal, territorial, state, provincial or local

government, (b) governmental, administrative, regulatory, quasi-governmental, monetary, fiscal or judicial authority, body, court, tribunal, arbitral body, commission, board, bureau, agency or instrumentality, or (c) political or other department, agency, subdivision, department or branch of any of the foregoing,

including any regulatory or oversight organization or association.

"Hazardous Materials" means any pollutant, contaminant or other substance identified or designated as

hazardous under applicable law.

"HGB" means the German Commercial Code (Handelsgesetzbuch).

"Information Technology" as defined in Section 12.1 of the Seller Warranties

"InsO" means the German Insolvency Code (*Insolvenzordnung*).

"Insurance Policies" as defined in Section 15.1 of the Seller Warranties

"IP Rights" as defined in Section 10.1b) of the Seller Warranties

"ISU" means the Investment Security Unit as formed and operating under the NSIA2021.

"Key Employees" as defined in Section 14.2 of the Seller Warranties

"KStG" means the German Corporate Income Tax Act (Körperschaftssteuergesetz).

"Leased Real Estate" as defined in Section 9.5 of the Seller Warranties
"Licensed IP Rights" as defined in Section 10.1c) of the Seller Warranties

"Material Agreements" as defined in Section 13.1 of the Seller Warranties

"Material Assets" as defined in Section 8.1 of the Seller Warranties

"Material Customers" as defined in Section 25.1 of the Seller Warranties

"Material Vendors" as defined in Section 25.1 of the Seller Warranties "Monthly Financial Reporting" as defined in Section 5.2 of the Seller Warranties

"NSIA2021" means the United Kingdom National Security and Investment Act 2021.

"OFAC" as defined in Section 21.5 of the Seller Warranties

"Open Source Software" as defined in Section 10.1f) of the Seller Warranties

"Owned IP Rights" as defined in Section 10.1e) of the Seller Warranties

"Owned Real Estate" as defined in Section 9.1 of the Seller Warranties

"Permits" as defined in Section 17.1 of the Seller Warranties

"PPP Lender" means TD Bank, N.A.

"PPP Loan" means the Debt evidenced by that certain promissory note dated on or about 22

April 2020, made by AF&R in favor of the PPP Lender, in the principal amount of

\$3,141,395.00.

"PPP Loan Requirement" means the applicable requirements of the CARES Act, including, without

limitation, the requirement that (i) the PPP Loan shall constitute a "covered loan" under, and as defined in, Section 1102 of the CARES Act (as in effect on the Closing Date) and (ii) the proceeds of the PPP Loan shall be used only for the purposes permitted under Section 1102 of the CARES Act (as in effect on the

Closing Date).

"Real Estate Permits" as defined in Section 9.9 of the Seller Warranties

"Registered IP Rights" as defined in Section 10.1d) of the Seller Warranties

"Related Party" as defined in Section 4.1 of the Seller Warranties

"Related Party Agreements" as defined in Section 4.1 of the Seller Warranties

"Sanctioned Person" as defined in Section 21.5 of the Seller Warranties

"Sanctions" as defined in Section 21.5 of the Seller Warranties

"SDN List" as defined in Section 21.5 of the Seller Warranties

"Seller Group" means Sellers and their Affiliates, except for the Target Group Companies.

"Seller Group Entity" means any person or entity that is a member of the Seller Group.

"Seller Transaction Expenses" means those fees, commissions, costs and expenses incurred or payable by any

Target Group Company, any Seller Group Entity or Seller Guarantor in connection with the Transaction, including (a) the fees of legal advisors, accountants, investment bankers, financial advisors, valuation firms or other professionals, and (b) any fees and expenses associated with attempting to fulfill or fulfilling any of the

Closing Conditions.

"Signing Date" means the date of this Agreement.

"State Department"	as defined in Section 21.5 of the Seller Warranties
"StGB"	means the German Criminal Code (Strafgesetzbuch).
"Subsidies"	as defined in Section 16.1 of the Seller Warranties
"Target Group Company Interests"	as defined in Section 3.2 of the Seller Warranties
"Third Party Developer"	as defined in Section 10.12 of the Seller Warranties
"Third Party Interests"	as defined in Section 9.4 of the Seller Warranties
"Transfer Tax"	means all federal, state, local or foreign transfer, sales, use, documentary, registration, real property transfer, mortgage recording, stamp duty or similar Taxes or charges (but excluding, for the avoidance of doubt, VAT), in each case including any interest and penalties relating thereto.
"U.S. Person"	as defined in Section 21.5 of the Seller Warranties
"VAT"	means (i) any tax imposed in compliance with the Council Directive of 28 November 2006 on the common system of value added tax (EC Directive 2006/112) and (ii) any other tax of a similar nature, whether imposed in a member state of the European Union in substitution for, or levied in addition to, such tax referred to in (i) above, or imposed elsewhere.
"ZPO"	means the German Code of Civil Procedure (<i>Zivilprozessordnung</i>).

2. List of Definitions

The following is a list of all defined terms used in this Agreement, indicating the page of the Agreement where the definition is introduced:

2017 Promissory Notes	9	Andreas Gebhardt	5
2019 Promissory Notes	9	Andreas Gebhardt Consultancy Agreement	37
2022 Amendment	9	AO	72
2022 Business Plan	72	Applied Subsidies	72
Adjustment Amount	18	ASG/RKT Collateral	9
AF&R	72	Asserted Claims	19
AF&R Mexico	72	Asserted Claims Tax	20
Affiliate	72	Assumption Agreement	12
Agreement	6	Assumption Agreements Carve-out	12
AktG	72	AW Holding	6
Alfmeier China	72	AWG	72
Alfmeier Czechia	72	AWV	72
Alfmeier Target Group	5	BEIS	72
Alfmeier Target Group Company	5	Beneficiary	23

BGB	72	Escrow Amount Company	13
BIS	72	Escrow Amount Tax	13
BMWK	30	Escrow Amounts	13
Bring Down Certificate	42	EStG	73
Bulletin 7	72	EUR	73
Bulletin 7 Reporting	52	Exchange Act	59
Bulletin 7 Taxes	52	Exchange Act Information	59
Business	6	Exchange Rate	73
Business Day	73	Exempted Claims	55
Business Information	73	Existing Function Holder	36
Business Records	58	Facilities	9
Call Price	37	Facilities Agreement	8
CARES Act	73	Facilities Repayment Account(s)	27
Carve-out Purchase Price	12	Facilities Repayment Amount	27
Carve-outs	22	Facility Agent	8
Cash Pool	9	Final Effective Date Adjustment Statement	17
Closing	38	Financing Collateral	9
Closing Actions	40	Form 8-K Information	58
Closing Conditions	38	Fundamental Warranties	42
Closing Date	40	Gebhardt Shares	8
Closing Memorandum	42	Gehbardt Holding	5
Code	73	GeschGehG	74
Company	5	GewStG	74
Company Purchase Price	11	Governmental Entity	74
Company Shares	8	Hazardous Materials	74
Competing Business	60	HGB	74
Compliance Certificate	30	Hold Back Amount	12
Confidential Information	64	Indemnifiable Tax	48
Consolidated Financial Statements	73	Indemnified Claims	23
Continuing Liabilities	22	Indemnity Claim	54
Contract	73	Information Technology	74
Copyleft Software	73	Inquiry Persons	43
Data Room Documents	45	InsO	74
Data Room Expiry	45	Insurance Policies	74
De Minimis Amount	55	Insurer	55
Debt	73	Interim Period	31
Disclosure Schedule	42	IP Rights	74
Effective Date	10	ISU	74
Effective Date Adjustment Statement	14	Johann Vielberth	5
Effective Date Balance Sheet	14	Key Employees	74
Effective Date Cash Company	11	KStG	74
Effective Date Financial Debt Company	11	Leased Real Estate	74
Effective Date Working Capital Company	12	Leaving Function Holders	36
ELBER	5	Lenders	8
ELBER Shares	8	Licensed IP Rights	74
Employee Benefit Plans	73	Long-Stop Date	62
Employees Employees	73	Losses	44
Employment Claims	12	Markus Gebhardt	5
Employment Termination Agreements	36	Material Agreements	74
Environmental Contamination	73	Material Assets	74
Environmental Contamination Environmental Law	73	Material Customers	74
ERISA Affiliate	73	Material Pre-Closing Carve-out Documentation	22
Escrow Account	18	Material Vendors	74
Escrow Account Escrow Agent	18	Merger Control Authorities	29
<u> </u>	18	Merger Control Clearances	29
Escrow Agreement	10	Micigal Control Cicalances	29

Monthly Financial Reporting	74,	RKT SPA	21
, , ,	75	Sanctioned Person	75
NDA	65	Sanctions	75
Neutral Expert	16	Scheduled Closing Date	40
New Function Holder	36	SDN List	75
NSI Clearance	39	SEC	59
Obligors	8	Securities Act	59
OFAC	75	Security Agent	8
Open Source Software	75	Seller	5
Other Indemnified Claims	23	Seller Group	75
Other Reported Debt	9	Seller Group Entity	75
Other Reported Debt Agreements	9	Seller Guarantor	5
Other Reported Debt Collateral	9	Seller Guarantors	5
Owned IP Rights	75	Seller Warranties	42
Owned Real Estate	75	Sellers	5
Parent	6	Sellers' Knowledge	42
Parties	5	Shares	8
Party	5	Signing Date	76
Paying Agent	9	SMA Holding	6
PCAOB	59	SMA Holding SPA	22
Permits	75	State Department	76
PLTA	9	StGB	76
PPP Lender	75	Straddle Period Tax	51
PPP Loan	75	Subsidiaries	8
PPP Loan Requirement	75	Subsidies	76
PRC	52	Target Group	5
PRC Payment Notice	53	Target Group Company	5
Preliminary ELBER Amount	13	Target Group Company Interests	76
Preliminary Gebhardt Amount	13	Target Working Capital Company	12
Preliminary Purchase Price Company	12	Tax	48
Pre-Signing Carve-out	6	Tax Benefits	49
Promissory Note Repayment Account(s)	28	Taxing Authority	48
Promissory Note Repayment Amount	28	Termination Agreement Andreas Gebhardt	36
Purchase Price	11	Termination Agreement Markus Gebhardt	36
Purchaser	5	Termination Agreement Stephan Schulak	36
Purchaser Claim Notice	46	Third Party Claim	46
Purchaser Facilities Agreement	20	Third Party Developer	76
Purchaser's Warranty Claims	44	Third Party Interests	76
Real Estate Permits	75	Threshold	55
Reference Pro-Forma Balance Sheet	14	Transaction	6
Refinancing Debt	25	Transfer Tax	76
Registered IP Rights	75	Transitional Services Agreement	37
Related Party	75 75	U.S. Person	76
Related Party Agreements	75 75	VAT	76
Release Agreement Facilities	25	VAT Determination Date	13
Release Agreements	26	VAT Reinstatement	13
Release Agreements Other Debt	26	VAT Reinstatement Compensation	13
Relevant Persons	43	VDA	50
Repayment Agreement 2017 Promissory Note	26	W&I Insurance	55
Repayment Agreement 2019 Promissory Note	26	W&I Liability Exclusion	56
Required Financial Information	59	ZPO	76
RKT	8	210	70
RKT Release Agreement	26		
Terra residuade i ignociment	20		

ANNEX 2.2 (LIST OF EXHIBITS)

[Exhibit P-(C) Pre-Signing Carve-out agreements]

Exhibit 3.2 Subsidiaries

[Exhibit 4.1 Facilities Agreement Amendment Agreement]

Exhibit 4.2 Financing Collateral

Exhibit 4.4 PLTA

Exhibit 4.5 Cash Pool Agreement

Exhibit 4.6-A Other Reported Debt Agreements

Exhibit 4.6-B Other Reported Debt

Exhibit 4.7 ASG/RKT Collateral

Exhibit 6.1-A Minutes of the general meeting Company

Exhibit 6.1-B Approval of the board of directors Company
Exhibit 6.2 Shareholders' resolution Gebhardt Holding

Exhibit 6.3 Shareholders' resolution ELBER

Exhibit 7.1.1 Sample Calculation of Company Purchase Price

Exhibit 7.1.2 Effective Date Cash Company, Effective Date Financial Debt Company and

Effective Date Working Capital Company

Exhibit 7.2.2-A Assumption Agreements Carve-out

Exhibit 7.2.2-B Assumption Agreement

Exhibit 8.1.2a) Accounting Policies and Consolidation Principles

Exhibit 10.2 Bank Accounts

Exhibit 11.1.1 Draft Escrow Agreement

Exhibit 14.5 Pre-Signing/Pre-Closing Carve-Out Agreements
Exhibit 16.1.1 Merger Control Clearances relevant for Closing

Exhibit 17.1.2 Permitted Actions Interim Period
Exhibit 17.1.4 Conduct of Business Contacts
Exhibit 17.2.1a) Leaving Function Holders

Exhibit 17.2.1b) New and Existing Function Holders

Exhibit 17.2.1c) Termination Agreement Andreas Gebhardt
Exhibit 17.2.1d) Termination Agreement Markus Gebhardt
Exhibit 17.2.1e) Termination Agreement Stephan Schulak
Exhibit 17.2.1g) Andreas Gebhardt Consultancy Agreement

Exhibit 17.2.1h) Transition Services Agreement

Exhibit 20.2d) Discharge of Leaving Function Holders

Exhibit 20.2f)

Exhibit 20.2m)

General Release of Claims

Exhibit 20.2n)

Ancillary agreements

Exhibit 20.4

Closing Memorandum

Exhibit 21.1.1-A

Seller Warranties

Exhibit W.2.3 Silent Partnerships and Domination or Profit and Loss Transfer Agreements of

Target Group Companies

Exhibit W.2.4 Corporate Documents of Target Group Companies

Exhibit W.2.6 Governing Bodies of Target Group Companies

Exhibit W.5.1 Financial Statements

Exhibit W.5.2 Monthly Financial Reporting

Exhibit W.8.6 Bank Accounts of Target Group Companies

Exhibit W.9.1

Exhibit W.9.5

Leased Real Estate

Exhibit W.10.2

Owned IP Rights

Exhibit W.10.5

Exhibit W.10.8

Exhibit W.10.11

Open Source Software

Exhibit W.13.1

Material Agreements

Exhibit W.14.1 Employees
Exhibit W.14.2 Key Employees

Exhibit W.14.6 Agreements with Trade Unions, Company Practices and General Commitments

Exhibit W.14.8Employee Benefit PlansExhibit W.14.18Employee Time CreditsExhibit W.15.1Insurance Policies

Exhibit W.16.1 Subsidies

Exhibit W.20.1 Disclosure Compliance

Exhibit W.22.1 Warranty and Product Liability Claims

Exhibit W.23.2 Hazardous Materials

Exhibit W.25.1 Material Customers and Material Vendors

Exhibit 21.1.1-B Disclosure Schedule

Exhibit 21.2-A Persons Relevant for Sellers' Knowledge

Exhibit 21.2-B **Inquiry Persons** Exhibit 26.1 **Indemnity Data Protection** Exhibit 26.2 **Indemnity IP Rights** Exhibit 26.3 **Indemnity Employment Matters Exhibit 26.4.1 Indemnity Environmental Matters** Exhibit 26.5 **Indemnity Identified Risks Exhibit 27.1.2 W&I** Insurance Policy Exhibit 28.1 Sellers' Liability Persons Exhibit 29.5 Accounting Firms **Exhibit 30.1.1** Countries with Non-Compete **Exhibit 30.1.2a)** Permitted Business Activities Exhibit 37 Public Announcement

Project Astute

Exhibit 21.1.1-A

Seller Warranties

EXHIBIT 21.1.1-A (SELLER WARRANTIES)

XVI.

Fundamental Warranties

1. Corporate Status and Authorization of Sellers and Seller Guarantors

- 1.1 Gebhardt Holding and ELBER are each a limited liability company, duly established and validly existing under the laws of Germany.
- 1.2 All Sellers and Seller Guarantors have all corporate power and authority to execute and perform this Agreement and to effect the transactions contemplated therein and the execution and performance of this Agreement does not violate any provisions of the articles of association, certificate of incorporation, bylaws or equivalent constitutional document of any Seller. With the exceptions of the clearances and approvals set forth in Section 17 of this Agreement, Sellers or Seller Guarantors are not obligated to obtain consents of third parties (including government authorities) to enter into or perform this Agreement.
- 1.3 This Agreement has been duly executed on behalf of all Sellers and Seller Guarantors and constitutes legal, valid and binding obligations of all Sellers and Seller Guarantors that are enforceable in accordance with their terms.
- 1.4 The execution and performance of this Agreement and the transactions contemplated therein do not trigger rights of any third parties under any agreement to which any Seller or Seller Guarantor is a party to or violate any law or any judicial or governmental order binding on any Seller or Seller Guarantor.
- 1.5 There is no legal or administrative proceeding pending (*rechtshängig*) against any Seller or Seller Guarantor or threatened in writing against any Seller or Seller Guarantor before any court, arbitral tribunal or public authority that in any manner challenges or seeks to prevent, alter or delay any transactions contemplated under this Agreement.
- No insolvency proceedings (*Insolvenzverfahren*) or similar proceedings under applicable laws have been applied for by any Seller or Seller Guarantor or, to Sellers' Knowledge, have been applied for by third parties, have been rejected on account of lack of assets or are pending regarding any Seller or Seller Guarantor and no Seller or Seller Guarantor is required under applicable laws to apply for any such proceeding or is unable to pay its debts (*zahlungsunfähig*).

2. Corporate status of the Target Group Companies

2.1 The Target Group Companies are duly established and validly existing under their respective jurisdictions of incorporation or formation, and are qualified to do business in their respective jurisdictions. AF&R Mexico was incorporated as, and currently is, a stock corporation with variable capital (*Sociedad Anónima de Capital Variable*). Each Target Group Company has all

necessary corporate power and authority to own, operate or lease the properties and assets owned, operated or leased by it and to carry on its business as currently conducted.

- 2.2 The corporate information in the Preamble (A) and (B) and Sections 3 and 4 of this Agreement is true and correct. No Target Group Company holds either directly or indirectly (including through an escrow agent or trust agreement) any share or interest in any other legal entity other than in other Target Group Companies and AW Holding, SMA Holding and RKT or is obligated to acquire any such share or interest.
- 2.3 Except as listed in Exhibit W.2.3, no Target Group Company is a party to any silent partnerships, domination or profit and loss transfer agreements (*Beherrschungs- oder Gewinnabführungs-verträge*) or other agreements within the meaning of section 291 *et. seq.* AktG or analogous provisions under the applicable laws of their respective jurisdictions of incorporation that would entitle any person or entity other than the Company to participate in the profits of or to exercise control over any Target Group Company. They are also under no obligation to enter into any such agreements.
- Exhibit W.2.4 contains a correct and complete list of the applicable articles of association, by-laws, partnership agreements, shareholders' agreements, joint venture agreements and similar constitutional documents (including any amendments thereto) of, or arrangements relating to the interests in, the Target Group Companies. Except as set out in Exhibit W.2.4, correct and complete copies of such documents have been made available to Purchaser in the Datasite on or before the third Business Day prior to the Signing Date by or on behalf of Sellers.
- 2.5 As of the Signing Date, no registration or application for registration in any commercial register or any similar public register of any Target Group Company is pending, and there is no matter that is required to be registered in any such register, but has not been registered. As of the Signing Date, no shareholder resolution of any Target Group Company is being challenged or threatened in writing to be challenged.
- 2.6 No Target Group Company has a supervisory board, advisory board, administrative board or any similar governing body, except as listed in <u>Exhibit W.2.6</u>.
- 2.7 No insolvency proceedings (*Insolvenzverfahren*) or similar proceedings under applicable laws have been applied for by any Target Group Company, or to Sellers' Knowledge, by third parties, have been rejected on account of lack of assets or are pending regarding any Target Group Company and no Target Group Company is required under applicable laws to apply for any such proceeding. To Sellers' Knowledge, there are no circumstances that would justify such proceedings or any actions seeking to void or challenge this Agreement under insolvency law. No Target Group Company has a negative net worth (*ist insolvenzrechtlich überschuldet*) or is unable to pay its debts as they fall due (*zahlungsunfähig*) under the applicable laws of its jurisdiction of incorporation.

- None of the Target Group Companies' corporate bodies has adopted a resolution that (i) will come into effect after the Signing Date, or (ii) would constitute any obligation of any Target Group Company to pay out any share in its profit after the Signing Date.
- 2.9 There are no claims pending or, to Sellers' Knowledge, threatened against any Target Group Company which are outstanding or unsettled of any current or previous member of any corporate body of any Target Group Company and no claims were pending or, to Sellers' Knowledge, threatened at any time during the past three years, in each case, other than unpaid salary or wages for services rendered or expense reimbursement in the ordinary course of business that are unpaid or unsatisfied due to timing considerations and not because there are actual disputes.
- 2.10 The list of shareholders or share register and capital variations registry book of each Target Group Company (in each case, to the extent applicable and to the extent statutorily required and for the past five (5) years), which are in the possession of such Target Group Company, are complete and correct.
- 2.11 The legal reserve of AF&R Mexico has been created and duly accrued for as required by Mexican law.

3. The Shares and the Target Group Company Interests

- 3.1 The Gebhardt Shares, the ELBER Shares and all shares and interests that the Company holds in any other Target Group Company are validly issued in compliance with all applicable laws. The contributions on the Gebhardt Shares, the ELBER Shares and all shares and interests that the Company directly or indirectly holds in any other Target Group Company have been fully made, have not been repaid or returned in any way and there are no obligations to make further contributions on any of the Gebhardt Shares, the ELBER Shares or any shares and interests that the Company holds in any other Target Group Company.
- 3.2 Gebhardt Holding is the sole and unrestricted legal and beneficial owner of the Gebhardt Shares. ELBER is the sole and unrestricted legal and beneficial owner of the ELBER Shares. The Target Group Companies holding shares or interests in other Target Group Companies (collectively, the "Target Group Company Interests") are the unrestricted legal and beneficial owners of such shares or interests, respectively. The Target Group Company Interests represent 100% of the registered capital and voting rights in the respective Target Group Company (except for AF&R, in which the Company holds 86.8% of the registered capital and voting rights). The Gebhardt Shares, the ELBER Shares and the Target Group Company Interests are free and clear of any liens, encumbrances and other rights of third parties. Except as listed in Exhibit W.2.3, the Gebhardt Shares, the ELBER Shares and the Target Group Company Interests are not subject to any trust arrangements, joint venture agreements, silent partnerships, sub-participations or similar agreements. There are no pre-emptive rights, options or other rights of any third party to purchase or acquire, or any rights of third parties to encumber, any of the Gebhardt Shares, the ELBER Shares or the Target Group Company Interests.

- The Gebhardt Shares, the ELBER Shares and the Target Group Company Interests are not subject to any resolution or agreement providing for their redemption or repurchase by their respective issuer or to any similar measures under applicable law. There is no (actual or contingent) obligation of any Target Group Company to issue, and no person or entity is entitled to request from any Target Group Company the issuance of, new shares or other equity interests (including convertible bonds, options or similar rights). There are no contributions for future capital increases (aportaciones para futuros aumentos de capital) recorded in the books and records of AF&R Mexico. No obligation or decision exists for any person to contribute to the equity capital of Alfmeier CZ.
- 3.4 The share certificates that represent the capital stock of AF&R Mexico have been duly and validly issued by AF&R Mexico in accordance with Mexican applicable laws. Such share certificates are in the possession of the Company. True, correct and complete copies of such share certificates have been made available to Purchaser in the Datasite prior to the Signing Date by or on behalf of Sellers.

XVII.

Other Seller Warranties

4. No Related Party Agreements

- 4.1 No Target Group Company has entered into any agreements, arrangements and other legal relationships, in each case, in writing, (i) with any Seller Group Entity or any person or entity that is a related person of a Seller Group Entity within the meaning of Section 138 (1) or (2) InsO (together, a "Related Party"), or (ii) which are for the benefit of any Related Party (as third party beneficiary or otherwise; the agreements, arrangements and legal relationships under (i) and (ii) collectively, the "Related Party Agreements").
- 4.2 To Sellers' Knowledge, all transactions between any of the Target Group Companies and any of its Affiliates or any of the Related Parties (including any Seller Group Entity) have been executed on arms' length terms (in particular with respect to transfer pricing and the OECD Guidelines), with no intention of tax base erosion or profit shifting, and within the ordinary course of business of the respective Target Group Company. All corporate approvals, consents or other documents required for valid execution and performance of such transactions were duly obtained.

5. Financials

The audited financial statements of the Company and the audited consolidated financial statements of the Alfmeier Target Group for the financial year ending on 31 December 2021, in each case, consisting of a balance sheet, a profit and loss statement and notes (*Anhang*) for the period between 1 January 2021 until 31 December 2021 (collectively, the "Consolidated Financial Statements"), copies of which are attached hereto as Exhibit W.5.1, have been prepared (i) with the care of a prudent businessman (*Sorgfalt eines ordentlichen Geschäftsmanns*),

(ii) in all material respects in accordance with IFRS, and applicable generally accepted accounting principles (Grundsätze ordnungsgemäßer Buchführung) and (iii) complying with the principle of accounting and valuation consistency (Bilanzierungs- und Bewertungskontinuität), in particular by applying the valuation criteria (Bewertungskriterien), depreciation principles (Abschreibungsgrundsätze) and election rights (Wahlrechte) consistent with past practice. The Consolidated Financial Statements present, in accordance with the laws and principles specified in the foregoing sentence and as of the date of their preparation, (to the extent audited) a true and fair view of or (to the extent not audited) do not materially misstate the assets and liabilities, financial condition and results of operations (Vermögens-, Finanz- und Ertragslage) of the Alfmeier Target Group on a consolidated basis as of the end of, and for, the period referenced therein.

5.2 Monthly Financial Reporting

The consolidated monthly financial reportings concerning the Alfmeier Target Group and its Business for the period from January until March 2022, a copy of which is attached hereto as Exhibit W.5.2 (the "Monthly Financial Reporting"), have been prepared (i) with the care of a prudent businessman, (ii) in all material respects in accordance with IFRS and applicable generally accepted accounting principles (*Grundsätze ordnungsgemäßer Buchführung*) and (iii) complying with the principle of accounting and valuation consistency (*Bilanzierungs- und Bewertungskontinuität*), in particular by applying the valuation criteria (*Bewertungskriterien*), depreciation principles (*Abschreibungsgrundsätze*) and election rights (*Wahlrechte*) consistent with past practice. The Monthly Financial Reporting, in accordance with the standards and rules specified in the foregoing sentence, do not materially misstate the assets and liabilities, financial condition and results of operations (*Vermögens-, Finanz- und Ertragslage*) of the Alfmeier Target Group and the Business as of the end of, and for, the period referenced therein.

5.3 Books and Records

The books and records (including accounting and tax records) of the Target Group Companies are in all material aspects correct, up-to-date and complete and have been kept in accordance with applicable laws and are in the unrestricted possession of the Target Group Companies, including through third party service providers in the ordinary course.

5.4 No Undisclosed Liabilities; PPP Loan

5.4.1 To Sellers' Knowledge, except for liabilities that (a) (i) are specifically and fully accrued for or reserved against in the Consolidated Financial Statements, or (ii) have arisen in the ordinary course of business consistent with past practices since the date of the Consolidated Financial Statements, and (b), in the case of clause (ii), are normal and usual in amount and do not directly or indirectly result from, arise out of or relate to any breach of Contract, indemnification obligation, breach of warranty, tort, infringement, default, non-performance or violation of law, or any proceeding, no Target Group Company has, and, to Sellers' Knowledge, no event has

occurred or facts, circumstances or conditions exist which (with or without due notice or lapse of time or both or the occurrence of any further event or condition) would reasonably be expected to result in any Target Group Company having, any obligations or liabilities (whether actual, contingent or otherwise) whatsoever of a type required to be reflected on a balance sheet prepared in accordance with German GAAP, Mexican GAAP, U.S. GAAP or IFRS, as the case may be.

- 5.4.2 To Sellers' Knowledge, AF&R has used all of the proceeds of the PPP Loan solely for allowable purposes pursuant to, and in accordance with, the PPP Loan Requirements and has complied, and is in compliance with, the PPP Loan Requirements. To Sellers' Knowledge, all applications, documents and other information submitted to PPP Lender, the U.S. Small Business Administration or any other Governmental Entity with respect to the PPP Loan were true and correct in all respects as of the applicable time of submission or the periods set forth therein.
- 5.4.3 (i)The PPP Loan has no outstanding balance, (ii) any and all commitments by the PPP Lender to lend to AF&R in respect of the PPP Loan have been terminated, (iii) to Sellers' Knowledge, all of AF&R's obligations under the documents evidencing the PPP Loan have been satisfied, terminated and released, (iv) to the extent that any liens have attached to the assets of AF&R or otherwise secured, or could be deemed to have secured, the PPP Loan, all such liens have been terminated, discharged, and released in full, and (v) the PPP Loan and all guarantees, security agreements, promissory notes and other documents executed in connection therewith have been satisfied, terminated and released and have no further force and effect.

6. Conduct of Business

From 1 January 2022 and up to the Signing Date, (i) the Business has been conducted with the care of a prudent businessman (*Sorgfalt eines ordentlichen Geschäftsmanns*) and in the ordinary course of business, and in a manner consistent with past practice, subject to the addressing of incidents and challenges generally affecting the industry in which the Business operates (e.g. the Ukraine war, the supply chain issues or the COVID-19 pandemic) and (ii) no material adverse change has occurred with respect to the Target Group as a whole or the Business which does not generally affect the industry in which the Target Group Companies operate.

7. No Finder's Fee

No Target Group Company has paid or is obliged to pay a finder's fee or any other type of fee, renumeration, brokerage fee, commission or benefit to any of its directors, officers or employees or to any finder, broker, agent, advisor or other third party (including any Seller Group Entity) in connection with this Agreement, the agreement regarding certain options for the sale and purchase of all shares in COSMIQ Industrieverwaltungs- und Vermietungs-GmbH entered into concurrently with this Agreement between Gebhardt Holding and the Purchaser (the "COSMIQ Option Agreement"), the Transaction or any other transactions contemplated herein or in the COSMIQ Option Agreement.

8. Material Assets

- 8.1 The Target Group Companies own, lease, license or otherwise have a legal right to use all tangible and intangible moveable assets that are used in or necessary to carry on the Business as conducted at any time since 1 January 2022 (together the "Material Assets") and all Material Assets are in the exclusive possession or under the direct control of and subject to unrestricted access by the relevant Target Group Company.
- 8.2 All Material Assets that are owned by any Target Group Company are free of liens or encumbrances and other rights of third parties and there are no pre-emptive rights, options or other rights of any third party to purchase, acquire or use, or any rights of third parties to encumber, any of the Material Assets, in each case, except for those rights arising by operation of law or granted in the ordinary course of business (including by retention of title (*Eigentumsvorbehalt*)) and consistent with past practice.
- 8.3 To Sellers' Knowledge, the fixed Material Assets are in a good state of preservation and efficiency except for usual wear and tear, have been regularly serviced and maintained in the ordinary course of business and allow the Alfmeier Target Group Companies to run the Business in the same manner and scope as conducted on the date hereof. No usual capital expenditures regarding the fixed Material Assets have been deferred since December 31, 2020.
- 8.4 To Sellers' Knowledge, the inventories (*Vorräte*) of the Target Group Companies are in good and useable condition and are, in the case of finished goods, of a quality and quantity saleable in the ordinary course of business, in each case, except for obsolete, damaged, defective or slow-moving items for which adequate reserves have been established in accordance with past practice.
- 8.5 To Sellers' Knowledge, the receivables of the Target Group Companies are collectible without any retention rights, rights to set-off or deduction, except for any such rights granted in the ordinary course of business and consistent with past practice.
- 8.6 Exhibit W.8.6 sets forth complete and accurate information with respect to all bank and savings accounts, money market accounts and safe deposit boxes of Target Group Companies, including each account or box number associated therewith, the bank routing number, the name and address of the institution where such account or box is held and, where available, the name of each applicable Target Group Company's primary contact at such bank, the purpose of each such account or box (e.g. payroll, checking, etc.), the owner of such account or box and all persons authorized to deliver instructions to the institution in respect of such account or box.
- 8.7 AF&R Mexico has a legal right to use all equipment, machinery and Material Assets used by such company for its *maquila* activities, including equipment and machinery owned by other Target Group Companies.

9. Real Estate

- Exhibit W.9.1 contains a complete and correct list of all real estate that is owned or subject to hereditary buildings rights (*Erbbaurechte*), other in-rem real property rights (*grundstücksgleiche Rechte*) or equivalent rights in the relevant jurisdictions held by any Target Group Company (the "Owned Real Estate") setting out for each Owned Real Estate the owner, co-owners (if any), type of legal title, location, size, registration details, type of use, existing encumbrances and other rights of third parties. Except as set forth otherwise in Exhibit W.9.1, the Target Group Companies are the unrestricted legal and commercial owner of the Owned Real Estate. There is no obligation of any Target Group Company (i) to sell, transfer or otherwise dispose of any Owned Real Estate or any part thereof or to create or amend any encumbrances or other rights of third parties thereon or (ii) to acquire any further real estate or rights equivalent to real estate.
- 9.2 All real estate assets owned by Alfmeier CZ that are subject to registration have been registered with the Czech cadastral register as being owned by Alfmeier CZ. All easements created for the benefit of Alfmeier CZ have been duly registered with the Czech cadastral register. Alfmeier CZ has not executed any documents which are required to be filed or registered with the Czech cadastral register that have not yet been so filed or registered.
- 9.3 All consideration for the acquisition of any of the Owned Real Estate has been paid in full and none of the purchase agreements with respect to such real estate has been withdrawn by the respective seller.
- 9.4 Except as listed in Exhibit W.9.1, the Owned Real Estate is free of any encumbrances, pending restitution entitlement or claim, or expropriation claim (together the "**Third Party Interests**"), no person has successfully claimed any such Third Party Interest, and no person is in possession or occupation of any of them or any part thereof. There are no pending disputes regarding boundaries, ownership, Third Party Interests, use, access or other matters relating to any Owned Real Estate, and to Sellers' Knowledge, no such disputes are threatened in writing.
- Exhibit W.9.5 contains a complete and correct list of all real estate that is leased or sub-leased by any of the Target Group Companies as lessee (the "Leased Real Estate") setting out for each Leased Real Estate the location, size, type of use, and, in relation to the lease or sub-lease agreements, the parties, date of agreement, monthly rent and ancillary payments. Each of the lease or sub-lease agreements is valid, in full force and enforceable in accordance with its terms, establishes the right of the relevant Target Group Company to use such premises for its business. The Target Group Companies and, to Sellers' Knowledge, all other parties, have complied with all provisions (including rent payment) in all material respects. None of the counterparties under any such lease or sub-lease agreement has given written notice to the relevant Target Group Company of, or made any claim in writing with respect to, any breach or default thereunder, and, to Sellers' Knowledge, there is no fact giving rise to such a breach or default such that a counterparty would be entitled to terminate it or restrict the rights of the Target Group Company thereunder in any material respect.

- The Target Group Companies have the right pursuant to any applicable laws to use the Owned Real Estate, the Leased Real Estate and the premises, constructions and fixtures installed thereon, in each case, as required to carry on the Business as conducted on the date hereof, and such use does not violate any rights of third parties or conflict with any contracts to which any Target Group Company is a party or is otherwise bound. All construction, installations and facilities, which are situated on the Owned Real Estate and the Leased Real Estate and which require regulatory approval, as well as the current use of those items, have been granted unconditional and perpetual approval with regulatory finality or grandfathered status (bestandskräftig) and do not violate any other currently applicable laws.
- 9.7 To Sellers' Knowledge, there are no outstanding payments, and the Target Group Companies have satisfied their respective obligations, regarding services and utilities related to the Leased Real Estate (including insurance and the obligation to pay deposits or performance bonds (*fianzas*)). No Target Group Company is a sublessor or grantor under any sublease or other instrument granting to any other person any right to possess, lease, occupy or use any Leased Real Estate.
- 9.8 There is no outstanding written notice or dispute as to any violation of the relevant planning and zoning laws or regulations or any written allegations of breaches of planning and zoning laws or regulations in relation to any of the Owned Real Estate or Leased Real Estate.
- 9.9 Any building, construction, installation or facility on any Owned Real Estate has been constructed and all other construction works on/in each of the Owned Real Estate have been performed, in all material respects, in accordance with legally effective zoning and building plans or permits where such permit was required and their use for the purposes for which they are being used (in particular, but not limited to, the activities of the Business) has been approved by the public authorities through legally effective decisions or approvals), as well as any hygiene and environmental decisions or approvals, where such decisions or approvals were required. The current use of each Owned Real Estate, and the current by any Target Company of the Leased Real Estate, constitutes lawful and permitted use under the applicable laws and the applicable licences, authorisations and permits relating to zoning, construction and use of each Owned Real Estate and each Leased Real Estate (the "Real Estate Permits"). Each such Real Estate Permit is in full force and effect and unconditional (or subject only to a condition which has been satisfied) and does not require any work to be carried out in order for it to be maintained in force. No revocation or, where such Real Estate Permits may need to be renewed, non-renewal of any such Real Estate Permit is threatened in writing.
- 9.10 No real estate other than the Owned Real Estate and the Leased Real Estate is used or is necessary to carry on the Business as conducted on the date hereof.
- 9.11 To Sellers' Knowledge, the Owned Real Estate, the Leased Real Estate and the premises, constructions and fixtures thereon are in a good state of preservation and efficiency except for usual wear and tear and have been regularly serviced and maintained in the ordinary course of business. To Sellers' Knowledge, no circumstances exist that would require expenditures in

excess of EUR 100,000 per case by any Target Group Company in respect of the Owned Real Estate, the Leased Real Estate or the premises, constructions and fixtures thereon within the next three (3) years.

9.12 There exists no pending or, to Sellers' Knowledge, threatened in writing condemnation, confiscation, eminent domain proceeding, dispute, claim, demand or similar proceeding with respect to, or which could materially and adversely affect, the continued use and enjoyment of the Owned Real Estate, the Leased Real Estate or the premises, constructions and fixtures thereon. To Sellers' Knowledge, there are no circumstances that would entitle any Governmental Entity or other person or entity to take possession or otherwise restrict use, possession or occupation of the Owned Real Estate, the Leased Real Estate or the premises, constructions and fixtures thereon.

10. IP Rights

10.1 Definitions

- f) "Copyleft Software" means any Open Source Software (as defined below) that is freely available under license conditions according to which any software incorporated in, distributed with, or derived from it, (i) is to be disclosed or distributed in source code form, (ii) has to be made available at no charge, (iii) restrict one's ability to charge for distribution of or to use software for commercial purposes or (iv) is subject to any other transparency obligations or limitations for a commercial use.
- g) "IP Rights" means any patents, design rights or design patents whether registered or unregistered –, copyrights, trademarks, applications to register and rights to apply for any of the afore-mentioned rights, and renewals of such rights, rights in unpatented technical and other know-how (whether patentable or not), and any other intellectual or industrial property rights of any nature whatsoever in any part of the world, whether registered or unregistered. For the purpose of this definition.
 - (i) "patents" shall include (i) all patents, utility models, and all applications for such rights, and (ii) all other registered intellectual property rights for inventions or other technical developments, and applications therefor worldwide, including for example provisionals, substitutions, extensions, reissues, re-exams, renewals, divisionals, continuations, continuations-in-part, continued prosecution applications, supplemental protection certificates, and term extensions under applicable patent or other applicable law;
 - (ii) "trademarks" shall include trademarks, service marks, trade names (*Unterneh-menskennzeichen*) and other proprietary rights to any words, names, slogans, symbols, logos, tones, positions, devices or combinations thereof used to identify, distinguish and indicate the source or origin of goods or services; rights in and to internet domain names; and the business goodwill associated with each of the foregoing;

- (iii) "know-how" shall include any invention, discovery, development, data, information, process, method, technique, trade secret, composition of matter, formulation, article of manufacture or other know-how, and any physical (including electronic) embodiments of any of the foregoing, in each case, unless part of the public domain; and
- (iv) "copyrights" shall include all copyrights, rights to use and/or exploit copyrights (*Nutzungs- und/oder Verwertungsrechte*), ancillary rights and sui-generis rights e.g. database rights existing under the applicable copyright laws, whether in published or unpublished works and whether registered or unregistered; rights to computer programs according to section 69b German Copyright Act ("*vermögensrechtliche Befugnisse*"), rights to compilations, collective works and derivative works of any of the foregoing and moral rights in any of the foregoing; and registrations and applications for registration for any of the foregoing and any renewals or extensions thereof.
- h) "Licensed IP Rights" means all IP Rights that any Target Group Company is entitled to use on the basis of a license, a covenant not to sue or a similar agreement with a third party.
- i) "Registered IP Rights" means IP Rights that are registered in an appropriate register anywhere in the world, including applications for such registrations.
- j) "Owned IP Rights" means all IP Rights (whether registrable or not) that are owned by any Target Group Company or co-owned among multiple Target Group Companies.
- k) "Open Source Software" means any software including its source code which is freely available to the public under license conditions which permit any person to use, copy, study, improve, modify or change such software in modified or unmodified form.
- Exhibit W.10.2 contains a correct and complete list of all (i) Owned Registered IP Rights and, to Sellers' Knowledge, (ii) non-registered Owned IP Rights that are material for the conduct of the Business (including, in particular, all rights in know-how, technical drawings and proprietary software pertaining to the Business), setting out for each Owned IP Right at least the legal and commercial owner(s), type and subject matter, and, if applicable, the territory for which it is registered or applied for, the registration or applications numbers, and the formal registrant(s).
- 10.3 All Owned Registered IP Rights and all non-registered Owned IP Rights and all Licensed IP Rights, are valid (rechtsbeständig), in full force and enforceable. Each Target Group Company has taken all steps necessary for the prosecution, registration, administration, maintenance, renewal (including payment of any applicable renewal fees) and protection of Owned Registered IP Rights and the non-registered Owned IP Rights and Licensed IP Rights, and, to Seller's Knowledge, nothing has been done or omitted to be done by which the Owned IP Rights or Licensed IP Rights may cease to be valid.

- 10.4 Except as disclosed in Exhibit W.10.4, the Target Group Companies are the sole legal and commercial owners of the Owned IP Rights as set out in Exhibit W.10.2 and are free to use and dispose of the Owned IP Rights. The Owned IP Rights are free of any encumbrances and other rights of third parties and there are no pre-emptive rights, options or other rights of any third party to purchase, acquire, license or have assigned, or any rights of third parties to encumber, any Owned IP Rights, except for mandatory moral rights (*Urheberpersönlichkeitsrechte*) to the extent that they cannot be waived and non-exclusive licenses granted in the ordinary course of business.
- 10.5 Exhibit W.10.5 contains, except with regard to any off-the-shelf software and Open Source Software, as of the Signing Date, a correct and complete list of all Licensed IP Rights, setting out for each Licensed IP Right at least (i) the type and subject matter of the Licensed IP Right, (ii) the parties (name and address), date and term of the agreement constituting or providing for the Licensed IP Right and, if applicable, (iii) the territory or other restrictions which apply to the Licensed IP Right.
- 10.6 To Sellers' Knowledge, for the period of three (3) years prior to the Signing Date, all agreements constituting or providing for any Licensed IP Rights are valid, in full force and enforceable in accordance with their terms, and all material provisions have been complied with and performed by all parties thereto in all material respects.
- 10.7 To Sellers' Knowledge, no IP Rights other than the Owned IP Rights and the Licensed IP Rights are necessary or used to carry on the Business as conducted on the date hereof without violating any IP Rights of third parties. To Sellers' Knowledge, no Owned IP Rights or Licensed IP Rights will be adversely affected by this Agreement, the COSMIQ Option Agreement, the Transaction or any other transactions contemplated herein.
- 10.8 To Sellers' Knowledge, except as set out in <u>Exhibit W.10.8</u>, no Alfmeier Target Group Company in connection with the Business is infringing any IP Rights of any third party, and, to Sellers' Knowledge, no third party is infringing any Owned IP Rights or Licensed IP Rights.
- 10.9 No legal or administrative proceeding relating to any Owned IP Right or Licensed IP Right (including, but not limited to, any limitation on its use, enforceability, validity or ownership) is pending or, to Sellers' Knowledge, has been threatened or announced in writing by or against the Target Group Companies, and, to Sellers' Knowledge, no circumstances exist that can reasonably be expected to justify or give rise to any such proceeding.
- 10.10 The Target Group Companies have in the past three (3) years not violated, and, to Sellers' Knowledge, there is no reasonable basis to believe that any Target Group Company violates any applicable law on unfair competition or unfair trade practices. The Target Group Companies have not received any written warning alleging a Target Group Company of such act, and no proceeding of this kind is pending before any court or tribunal, office or other authority.
- 10.11 <u>Exhibit W.10.11</u> contains a correct and complete list of all Open Source Software used by the Alfmeier Target Group Companies in connection with the Business. The Target Group Companies have complied with all terms and conditions applicable to any Open Source Software

used. No Owned IP Rights are subject to any terms and conditions of any Copyleft Software. The use of Open Source Software does not restrict or prevent the commercializing of any products or services of the Business.

- 10.12 All IP Rights that have been developed or created by any current or former employee or freelancer of any Target Group Company or by any other person in a similar position (if any, each a "Third Party Developer"), in each case, to the extent developed or created in the course of doing work for such Target Group Company and that are necessary to carry on or are used in the Business, have been transferred to the Target Group Company or, where a transfer is impossible due to mandatory law, the Target Group Company, respectively, have been granted an exclusive and unlimited right to use the relevant IP Right. As for service providers of any Target Group Company or any other person in a similar position of any Target Group Company, to Sellers' Knowledge, the IP Rights that they have been developed or created, in each case, to the extent developed or created in the course of doing work for such Target Group Company and that are necessary to carry on or are used in the Business, a sufficient license for the use of these IP Rights by the relevant Target Group Company for the Business is in place.
- 10.13 The Target Group Companies have complied with the applicable provisions relating to employee inventions, including the German Act on Employee Inventions (Gesetz über Arbeitnehmererfindungen), the Czech Copyright Act (Czech Act No. 121/2000 Coll, as amended, in Czech: autorský zákon), in each case, to the extent applicable, and applicable Mexican laws. In particular, the German Target Group Companies have formally and correctly laid (when applicable: unrestricted) claim (Diensterfindungen (solange anwendbar: unbeschränkt) in Anspruch genommen) to the rights in service inventions underlying the Owned IP Rights, if applicable and to the extent that they are necessary to carry on in the Business. Also, the Target Group Companies have paid all remuneration and other compensation relating to any commercial exploitation of Registered IP Rights made prior to the Closing Date to persons entitled thereto pursuant to applicable laws such as the German Act on Employee Inventions, the Czech Copyright Act and/ or the Czech Civil Code (Czech Act No. 89/2012 Coll., as amended, in Czech: občanský zákoník), in each case, to the extent applicable, or pursuant to contractual arrangements, if applicable. The Target Group Companies are not obliged to pay any remuneration or other compensation for any use of the Owned IP Rights made prior to the Closing Date.

11. Protection of Business Information

- 11.1 To Sellers' Knowledge, all actions required in accordance with best industry practice to conserve and protect the confidentiality and value of all know-how and trade secrets relating to the Business, including, the know-how and trade secrets comprised in the Owned IP Rights as set out in Exhibit W.10.2 (together the "Business Information") have been taken.
- To Sellers' Knowledge, no Business Information (i) has been made available to any third party during the past five (5) years prior to the Signing Date except under confidentiality and restricted

use agreements or (ii) is available to, or used by, any third party against the intent of the relevant Target Group Company.

- 11.3 All Business Information is available in the Business.
- 11.4 To Sellers' Knowledge, no records, technological systems, data or information necessary for and/or used in the Business of the Target Group Companies are dependent upon or accessible by any means, which are not under the exclusive ownership and/or direct control of the Target Group Companies.

12. Information Technology

- 12.1 The Target Group Companies own, lease, licence or otherwise have a legal right to use all information and communications technology including computer systems, hardware, software/firmware, telecommunications systems, servers, websites, network and data center infrastructure, related equipment and associated documentation that is used for or required to carry on the Business as conducted on the date hereof and on the Closing Date (together the "Information Technology").
- 12.2 All Information Technology that is owned by any Target Group Company is free of encumbrances and other rights of third parties and there are no pre-emptive rights, options or other rights of any third party to purchase or acquire, or any rights of third parties to encumber, any of such Information Technology, in each case, except for rights arising by operation of law or granted in the ordinary course of business.
- 12.3 To the extent that the right to use, or to receive any services relating to, any Information Technology is granted by contractual agreement, such agreements cannot be ordinarily terminated by the relevant third party (i) with a notice period of less than three (3) months, or (ii) due to the consummation of this Agreement.
- 12.4 All software that is part of the Licensed IP is licensed to the relevant Target Group Company under licenses that, in all material respects, are commercially reasonable and authorize the use of the software in the manner and on the scale on which it is currently used.
- 12.5 In the period of three (3) years prior to the Closing Date, none of the Target Group Companies has disclosed, delivered or licensed to any person or agreed or obligated itself to disclose, deliver or license to any person, or permitted the disclosure or delivery to any escrow agent or other person of, nor has there been any unauthorized or inadvertent disclosure of, any source code used in the Business, other than disclosures to employees, contractors and consultants involved in the development or maintenance of the products of the relevant Target Group Company, and subject to a written agreement containing terms as to confidentiality.
- 12.6 To Sellers' Knowledge, no event has occurred, and no circumstance or condition exists, that results in any Target Group Company being legally or contractually required to disclose, deliver or license to any third party any source code for software used in the Business, other than

disclosures to employees and consultants involved in the development and maintenance of the products of the relevant Target Group Company or otherwise in the ordinary course of business and consistent with past practice.

- 12.7 To Sellers' Knowledge, the Information Technology is in good working order, fit for the purpose for which it is used and performs in accordance with its specifications and documentation and as otherwise required for the Business. All Information Technology that is owned by any Target Group Company is supported by technically competent and trained employees or dedicated IT services (including by third parties), as the case may be, and has, to Sellers' Knowledge, in the case of hardware and software, appropriate customary hardware and software support and maintenance agreements in place to minimise the risk that any error, breakdown, failure or security breach occurs and to ensure that if any such event occurs it does not cause a disruption to the Business.
- 12.8 To Sellers' Knowledge, the Information Technology has not suffered any outage, unauthorized intrusion, breakdown or other failure in the past 24 months (unless immaterial in the case of a breakdown or failure) and no circumstances exist that can reasonably be expected to give rise to any such incident. To Sellers' Knowledge, the Target Group Companies have taken reasonable steps and implemented reasonable procedures to ensure that the Information Technology used in connection with the operation of the Business are free of any disabling codes or instructions and any "back door," "time bomb," "logic bomb", "Trojan horse," "worm," "drop dead device", "bug", "virus" or other destructive programs, scripts routines or hardware components that permit unauthorised access or unauthorised disablement or erasure of such Information Technology or data or other software of users.
- 12.9 To Sellers' Knowledge, the Target Group Companies have, in a manner consistent with applicable industry and otherwise commercially reasonable standards, implemented back-up and disaster recovery technology and procedures, hold available facilities, and have taken all other steps to safeguard the Information Technology and, if required, to bypass any malfunctions. To Sellers' Knowledge, the Target Group Companies have implemented all current security patches, updates and upgrades that are generally available for the Information Technology.
- 12.10 To Sellers' Knowledge, the use of the Information Technology by or on behalf of the Target Group Companies has not infringed, and, to Sellers' Knowledge, there is no reasonable basis to believe that they do infringe any rights (including IP Rights) of any third party (including, without limitation, those of the Seller Group Entities). No such infringement has been asserted in writing during the past three (3) years.

13. Material Agreements

13.1 <u>Exhibit W.13.1</u> contains a correct and complete list as of the Signing Date of all of the following agreements pertaining to the Business that involve at least one party other than a Target Group Company and under which either party or both parties have not yet completely fulfilled their primary contractual obligations (*Hauptleistungspflichten*) (the "Material Agreements"), setting

out for each Material Agreement the type of agreement, the parties (name and address), date, term and any ancillary agreements, amendments and side letters:

- 1) agreements with any Material Customer or any Material Vendor;
- m) agreements with distributors, sales agents and other resellers or sales representatives accounting (each) for sales in the last full fiscal year of more than EUR 50,000;
- n) agreements that (i) limit the freedom of the Business to compete in any line of business with any third party or (ii) provide for any exclusivity provisions restricting the Business;
- o) agreements on strategic alliances, joint ventures, joint development of products, profit sharing or other forms of cooperation;
- p) loan agreements, capital leases or other financing instruments (as lender or borrower) unless listed in the Exhibits to Section 5 of the Purchase Agreement and except for capital leasing agreements in the ordinary course and with annual payments not exceeding EUR 50,000;
- q) guarantees, security agreements, comfort letters, indemnities or similar instruments granted for any liabilities of any third parties unless listed in the Exhibits to Section 5;
- r) agreements regarding swaps, options, forward transactions, futures and other derivatives;
- s) agreements to acquire, sell, transfer or dispose over any fixed assets (*Anlagevermögen*) with a fair market or replacement value exceeding, in each case, EUR 50,000;
- t) agreements on the construction of fixed assets (*Anlagevermögen*) or other capital expenditures exceeding, in each case, a value of EUR 150,000;
- u) lease agreements entered into as lessee or lessor relating to any real estate or other assets providing, in each case, for an annual rent exceeding EUR 75,000, to the extent not listed under Exhibit 9.5;
- v) agreements constituting or providing for any Licensed IP Rights other than off-the-shelf software;
- w) research and development agreements with third parties;
- x) agreements granting the right to use, or to receive any services relating to, any Information Technology that exceed a value of EUR 50,000 annually;
- y) agreements to acquire, sell, transfer or dispose over interests in other companies or businesses;

- z) agreements with public authorities (including tax authorities) or any entities controlled by any public authority which relate to any regulatory matter or other matter governed by public law;
- aa) agreements between any Target Group Company on the one hand and any Seller Group Entity or Related Party on the other hand except for any agreements entered into in connection with any Carve-out that are described in the Agreement;
- bb) to the extent not already covered by any other provision of this Exhibit 23.1.1-A: (i) long-term agreements (*Dauerschuldverhältnisse*) which provide for recurring claims or obligations exceeding, in each case, a value of EUR 250,000 per year and which may not be terminated on less than six months' notice and without penalty; (ii) agreements that have been concluded outside the ordinary course of business or not at arm's length; and
- cc) any commitment to enter into any of the agreements referred to in this Section 13.1.
- 13.2 (i) Each Material Agreement is in full force and effect and enforceable against the parties thereto in accordance with its terms, (ii) no party to any Material Agreement has terminated or is entitled to terminate or has announced in writing a termination of any Material Agreement, (iii) no Target Group Company, or, to Sellers' Knowledge, any other party to any Material Agreement is in default under or breach of any of its obligations under the Material Agreement, and (iv) to Sellers' Knowledge, no facts or circumstances exist which might invalidate or give rise to a ground for termination, avoidance or repudiation of any Material Agreement.
- 13.3 No party to any Material Agreement (other than any Target Group Company) is entitled to terminate, materially amend or claim any other rights under any Material Agreement solely as a result of the execution or performance of this Agreement, the COSMIQ Option Agreement or the transactions contemplated herein or therein (change of control).
- 13.4 To Sellers' Knowledge, no Target Group Company is due to pay any contractual penalty, damages, default interest or default charge and, to Sellers' Knowledge, there are no facts or circumstances which could result in such obligation of any Target Group Company.
- 13.5 To Sellers' Knowledge, no Target Group Company has agreed to a restriction on its right to conduct its business or compete freely with any other business, including a restriction on the geographic scope of its business, a restriction on the kind of business that it is entitled to carry on or a restriction obliging it to deal exclusively with a person.

14. Employment and Employee Benefits

14.1 <u>Exhibit W.14.1</u> contains a correct and complete list, as of the relevant date set forth in the Exhibit, of the employees of each Target Group Company (together the "**Employees**"), setting out for each individual on an anonymous basis the employer, department, position, start of employment, year of birth, part time arrangements, notice periods for termination, fixed term (if any), special protection from dismissal, indication on whether the employment has been terminated with the

notice period still running, annual gross base salary, performance-related and other material benefits (including performance, transaction and retention bonuses, stock options, profit participation rights, company cars to the extent not included in the budget of the Target Group) as well as post-contractual non-compete covenants and contractual severance claims.

- Exhibit W.14.2 contains a correct and complete list, as of the date set forth therein, of all employees, managing directors and officers (*Geschäftsführer / Mitglied eines Geschäftsführungsorgans*) of any of the Target Group Companies with a gross annual target salary (including base salary and target bonus) in excess of EUR 75.000 (together the "**Key Employees**"). None of the Key Employees have (i) announced in writing, given or received notice of termination or (ii) announced in writing, made, requested or received an offer to enter, or entered, into any amendment or termination agreement with the relevant Target Group Company.
- 14.3 Except as set forth in Exhibit W.14.1 or Exhibit W.14.2, no employee, managing director or officer of any Target Group Company (i) will have any termination rights or financial claims due to the execution or performance of this Agreement, the COSMIQ Option Agreement, or the transactions contemplated herein or therein, (ii) has been promised in writing any payment or benefit by any Target Group Company or Seller Group Entity in connection with or with a view to the Transaction or (iii) waived, at any time during the past three years, any right to receive any compensation from any Target Group Company in excess of EUR 25,000.
- 14.4 None of the outside labour personnel (including freelancers and leased employees, irrespective of whether leased from external agencies or within the Target Group) deployed for any Target Group Company has to be classified, or were classified at any time during the past three (3) years, as an employee of any Target Group Company pursuant to applicable law.
- 14.5 All compensation obligations (including obligations to deduct, withhold and pay social security contributions) of the Target Group Companies vis-à-vis or in respect of their current and former employees, directors and officers (including bonuses and profit sharing, as applicable in each jurisdiction) have been fulfilled or properly provided for, in each case, during the past three (3) years prior to the Closing Date.
- Exhibit W.14.6 contains a correct and complete list of all (i) agreements with unions (*Gewerkschaften*) (including, in case of AF&R Mexico, any *contrato colectivo de trabajo* and in case of Alfmeier CZ, any *kolektivní smlouva*), works councils (*Betriebsräten*) or similar organizations entered into by any of the Target Group Companies or applicable to any of the Target Group Companies due to membership in an employers' association and (ii) general commitments (*Gesamtzusagen*), in each of the cases (i) and (ii) to the extent still binding on any of the Target Group Companies. To Sellers' Knowledge, there are no employee representatives of any such unions, works councils or similar organizations operating within, or at any facility of, any Target Group Company.
- 14.7 Each Target Group Company is, and has been during the past three (3) years, in compliance with all agreements, practices and commitments set forth in Section 14.6. None of such agreements,

practices or commitments set forth under Section 14.6 restrict any Target Group Company's freedom to dismiss any of its employees or to change the terms of their employment by way of payment obligations in case of dismissals or amendments to terms of employment. All restructuring measures, in particular all measures described in the reconciliation of interests agreements (*Interessenausgleiche*) and social plans provided, have been completed.

- Exhibit W.14.8 contains a correct and complete list by jurisdiction of all pension commitments (including retirement, widows', dependents' and disability pensions), early or partial retirement schemes and other employee benefit plans (whether subject to ERISA or not, whether oral or written, whether fully insured or self-insured, whether qualified or nonqualified, whether funded or unfunded, on a defined benefit or defined contribution basis, or otherwise) relating to retirement, early retirement, nonqualified retirement, deferred compensation, severance, change in control, termination, redundancy, retention, equity based, bonus, incentive compensation, commission, death, disability, post-retirement health and welfare medical benefits or anniversary payments, under which any Target Group Company or any ERISA Affiliate has any obligations (such commitments and plans together the "Employee Benefit Plans"). In Germany, no old-age part-time scheme (*Altersteilzeit*) exists.
- 14.9 All Employee Benefit Plans comply and have been established, amended, administered, funded, maintained and operated in accordance with their terms and applicable law, in each case, in all material respects. To Sellers' Knowledge, any adjustments of pensions under the Employee Benefit Plans have been made as required by applicable law and the relevant contractual provisions, and no backlog adjustments are required to be made for periods prior to the Closing Date. All contributions owed under the Employee Benefit Plans by any Target Group Company have been duly and timely paid. To Sellers' Knowledge, there is no current or threatened underfunding of an Employee Benefit Plan.
- 14.10 Each Employee Benefit Plan that is intended to meet the requirements of a "qualified plan" under Code Section 401(a) may rely upon a current opinion or advisory letter from the Internal Revenue Service. All required reports and descriptions (including, without limitation, Form 5500 annual reports, Form 1094-Cs, Form 1095-Cs, summary annual reports, and summary plan descriptions) with respect to each Employee Benefit Plan have been properly and timely filed and/or distributed to participants and other applicable individuals in accordance with the applicable requirements of ERISA and the Code. No Employee Benefit Plan provides, and none of the Target Group Companies or any ERISA Affiliate has any current or potential obligation to provide, or to pay for or subsidize any amount of, post-employment, post-ownership, or post-termination health, life or other welfare benefits to any Person other than (including for a longer period than) as required by Part 6 of Subtitle B of Title I of ERISA or Section 4980B of the Code or similar state Law. No Employee Benefit Plan is and none of the Target Group Companies maintains, sponsors, participates in, contributes to, or has any obligation to contribute to, or has any other current or contingent liability or obligation (including on account of an ERISA Affiliate) under Title IV of ERISA or with respect to (i) any "multiemployer plan" (as defined in Section 3(37) of ERISA) or (ii) any "defined benefit plan" (as such term is defined in Section 3(35) of ERISA) or any other

plan that is or was subject to Section 412 or 430 of the Code or Section 302 or Title IV of ERISA. No Target Group Company has any Liability on account of at any time being considered a single employer with another Person under Section 414 of the Code. There has been no non-exempt "prohibited transaction" within the meaning of Section 4975 of the Code or Section 406 of ERISA and no breach of fiduciary duty (as determined under ERISA) has occurred with respect to any Employee Benefit Plan. There is no proceeding (other than routine and undisputed claims for benefits) pending or, to Sellers' Knowledge, threatened in writing with respect to any Employee Benefit Plan or against the assets of any Employee Benefit Plan, and, to Sellers' Knowledge, there is no fact or circumstance that could give rise to any such proceeding. Neither an Employee Benefit Plan nor any fiduciary thereof is the subject of an audit, investigation or examination by any Governmental Entity. None of the Target Group Companies or any ERISA Affiliate has any actual or potential obligation to indemnify, reimburse, or gross-up any Person for any Taxes or interest or penalties that may be imposed, incurred, or accelerated, including under Section 409A or 4999 of the Code. No payment, amount or benefit that could be, or has been, received (whether in cash or property or the vesting of cash or property or the cancellation of indebtedness) by any current or former employee, officer, stockholder, director or other individual service provider of any Target Group Company or as a result of or in connection with the execution and delivery of this Agreement or the consummation of the transactions contemplated by this Agreement could reasonably be expected to be characterized as an "excess parachute payment" (as defined in Section 280G(b)(1) of the Code or any corresponding provision of state, local, or foreign Tax law) or otherwise would not be deductible by reason of Code Section 280G or would be subject to an excise tax under Code Section 4999. With respect to each Employee Benefit Plan, each Target Group Company has delivered to Purchaser true, correct and complete copies of the current plan document, and the most recent summary plan descriptions and summaries of material modifications, the most recent favorable determination letter, or opinion letter received from or issued by the Internal Revenue Service (as applicable), and all related trust agreements, insurance Contracts, and other funding arrangements.

- During the past three (3) years, there has not been any strike, lockout or labour interruption or any lawsuit, dispute or proceeding before any court, conciliation committee (*Einigungsstelle*), arbitrator or public authority with any union (*Gewerkschaft*), works council (*Betriebsrat*) or similar organization, in each case, affecting or involving any Target Group Company. No circumstances exist that can reasonably be expected to justify or give rise to any of the foregoing.
- 14.12 No Target Group Company is involved in any legal proceedings before any court with any current or former employee, director or officer that has a value of the claim (*Streitwert*) or could result in losses in excess of EUR 25.000, in each case, excluding costs and fees, it being understood that any proceedings relating to the same or similar circumstances will be aggregated. No such proceeding has been threatened or announced, in each case, in writing, against any Target Group Company and, to Sellers' Knowledge, no circumstances exist that can reasonably be expected to justify or give rise to any such proceeding.

- 14.13 Each Target Group Company is in compliance with all applicable laws pertaining to employment and employment practices in all material respects (including, in the case of AF&R Mexico, the Mexican Ley Federal del Trabajo and the amendments thereto and to the Mexican Social Security Law, Law of the Institute of the National Fund for Employees' Consumption, Federal Tax Code, Income Tax Law, Value Added Tax Law, Federal Law of Employees at the Service of the State, and the Law Regulating Fraction VIII Bis of Section B of Article 123 of the Mexican Constitution, all made in connection with the subcontracting of personnel by Mexican Persons, published in the Federal Official Gazette on 23 April 2021, as well as any regulations derived thereof). Alfmeier CZ has complied with its obligations arising under the applicable laws relating to social security and health insurance. All obligations of Alfmeier CZ resulting from the employment or similar relationships with its current or former employees and relationships with its executive directors are being performed or appropriate reserves are created in respect of such obligations, regardless of whether such obligations are owed to the employees or public authorities.
- 14.14 There are no agreements for the provision of consultancy or similar services akin to employment between Alfmeier CZ and any person. To Sellers' Knowledge, Alfmeier CZ does not have and did not have during the past three (3) years any relations with persons carrying out dependent activities (in Czech závislá práce) for Alfmeier CZ otherwise than as employees.
- 14.15 All employment relationships with Employees have been operated in accordance with applicable immigration and work permit rules.
- 14.16 Other than within the agreements with each of Mr. Stephan Schulak and Mr. Thomas Konetznick that have been made available to Purchaser in the Datasite on or before the third Business Day prior to the Signing Date by or on behalf of Sellers, no change of control clauses are in place.
- 14.17 Other than within the agreements with each of Mr. Markus Gebhardt, Mr. Andreas Gebhardt and Mr. Stephan Schulak that have been made available to Purchaser in the Datasite on or before the third Business Day prior to the Signing Date by or on behalf of Sellers, no post-contractual restrictive obligations in agreements with Key Employees are in place.
- 14.18 <u>Exhibit W.14.18</u> contains an overview of all time credits accrued by overtime work and outstanding vacation entitlements from previous years for Employees in Germany as of 22 April 2022.
- 14.19 No payment or repayment obligations regarding short time work allowances (*Kurzarbeitergeld*) or similar subsidies are outstanding or owed by the Target Group Companies. To Sellers' Knowledge, calculations of short-time work allowances and top-up payments in Germany have been made correctly and short-time work (*Kurzarbeit*) has been introduced and conducted legally, which includes that the preconditions for short-time work have been met at all time.
- 14.20 No Seller, Seller Group Entity, any of its Affiliates or any other Sellers' related person has actively solicited the service or employment of any employee of a Target Group Company for any time period as of or after Closing.

15. Insurance

- Exhibit W.15.1 contains a complete and correct list of all insurance policies relating to the Business (the "Insurance Policies"), setting forth for each policy the type of insurance, insurer, policyholder, co-insured parties, policy number, period of insurance, insured sum and annual premiums.
- 15.2 To Sellers' Knowledge, the Insurance Policies provide insurance coverage for the Business against such risks and to such an extent as is in line with applicable industrial standards and required to comply with any applicable laws and contractual obligations.
- 15.3 The Target Group Companies have complied with all obligations under the Insurance Policies (including the payment of all premiums when due).
- 15.4 All Insurance Policies are in full force and effect and no notice of termination in writing has been issued or received by any Target Group Company. No Insurance Policy will terminate or may be terminated or modified as a result of the consummation of the transactions contemplated by this Agreement or the COSMIQ Option Agreement, and coverage under all Insurance Policies, as in effect as of the Closing Date, will continue, in the event of their termination, with respect to events occurring prior to their termination. No Target Group Company has received any notice in writing of premium increase with respect to, or alteration of coverage under, any Insurance Policy. All premiums due on the Insurance Policies have been paid.
- 15.5 There are no claims in writing pending under any Insurance Policy for matters exceeding EUR 25,000, and no claims for matters exceeding EUR 50,000 (whether or not pending under any Insurance Policy) have been questioned, denied or disputed by the insurer in the past three years. To Sellers' Knowledge, there are no circumstances or facts which could (i) become a claim exceeding EUR 25,000 under any Insurance Policy not yet reported to any insurer or (ii) relieve any insurer under any Insurance Policy from satisfying in full any claim by any Target Group Company under any Insurance Policy.
- The Insurance Policies are, in terms of types and amounts, of the nature customarily carried by persons conducting a business similar to the Target Group Companies and, to Sellers' Knowledge, are sufficient for compliance with all applicable laws and contracts or agreements to which any Target Group Company is a party or by which it is bound.

16. Subsidies

Exhibit W.16.1 contains a complete and correct list of all public grants (*Zuschüsse*), allowances (*Zulagen*), subsidies (*Subventionen*), investment grants (*Investitionszuschüsse*) and other public aids to any Target Group Company, (i) that have been granted in the last three (3) years, and (ii) under which, including the obligations under any ancillary provisions in the respective orders or agreements thereto, any of the Target Group Companies still has any obligations, or (iii) with regard to which an application is still pending (the public aids applied for as set forth in no. (iii)

together the "Applied Subsidies"; the Applied Subsidies after having been granted to the respective applicant and the public aids set forth in no. (i) and (ii) together the "Subsidies").

- 16.2 The Target Group Companies have complied with the terms of the Subsidies in all respects. No Subsidy will become repayable or subject to any other modification as a consequence of the consummation of this Agreement, the COSMIQ Option Agreement or the transactions contemplated hereby or thereby. The Subsidies are not being challenged by any third party and no legal or administrative proceeding regarding a modification or revocation of any Subsidy is pending, threatened or announced, in each case, in writing. No circumstances exist that can reasonably be expected to justify or give rise to any such challenge, modification or revocation.
- 16.3 Regarding to the Applied Subsidies, the respective Target Group Companies have fulfilled all requirements for the granting of the respective subsidies and, to Sellers' Knowledge, no circumstances exist that would reasonably be expected to prevent or delay such granting.
- 16.4 The terms of the Subsidies and the expected terms of the Applied Subsidies do not impede the continuance of the Business as conducted on the Signing Date or the Closing Date. The terms of the Subsidies and the expected terms of the Applied Subsidies do not provide for any obligation to maintain a certain number of employees or operations at any site of the Business.

17. Permits

- 17.1 To Sellers' knowlegede, the Target Group Companies validly hold and have in full force and effect all material public approvals, licenses, permits, accreditations, certifications and other authorizations and have submitted all notifications under applicable public law that are required to carry on the Business as conducted on the date hereof (including with respect to any Target Group Company's use, possession or occupation of the Owned Real Estate, the Leased Real Estate or the premises, constructions and fixtures thereon or, in case of AF&R Mexico, the IMMEX Program and the PROSEC Program) (together the "Permits").
- All of the Permits are held in the name of the respective Target Group Company and not in the name of any Seller Group Entities, any personnel or other representatives of the Target Group Companies or otherwise on behalf of the Target Group Companies. The Permits will, to Sellers' Knowledge, not be adversely affected by the consummation of this Agreement, the COSMIQ Option Agreement or the transactions contemplated hereby or thereby. The Target Group Companies have complied within the past three (3) years with the Permits in all material respects.
- 17.3 The Permits are not being challenged in writing by any third party and no proceeding regarding a modification or revocation of any Permit is pending, threatened or announced, in each case, of threatened or announced in writing. To Sellers' Knowledge, no circumstances exist that can reasonably be expected to justify or give rise to any such challenge, modification or revocation, or which might prejudice the renewal of any Permit.

18. Day-1-Readiness

As of the Closing Date, the Target Group Companies own, hold or have otherwise available (including rights of use) all tangible and intangible assets, rights and other resources and are party to all agreements reasonably required to conduct the Business as an operationally separate business in a manner in all material respects consistent as conducted on the Signing Date.

19. Proceedings

- 19.1 Except as disclosed in the Disclosure Schedule, no Target Group Company is involved (as party or otherwise) in any civil, criminal or administrative action, lawsuit, investigation or other proceeding before any court, arbitrator or public authority that has a value of the claim (*Streitwert*) or, to Seller's Knowledge, could result in losses in excess of EUR 75,000, in each case, excluding costs and fees, it being understood that any proceedings relating to the same or similar circumstances will be aggregated.
- No proceeding as referred to in Section 19.1 has been threatened or announced, in each case, in writing, against any Target Group Company or any Seller Group Entity in connection with the Business and, to Sellers' Knowledge, no circumstances exist that can reasonably be expected to justify or give rise to any such proceeding.
- 19.3 There is no decision in writing by any court, arbitrator or public authority nor any settlement that (i) requires payments from, or provides for other losses to be incurred by any Target Group Company or any Seller Group Entity in connection with the Business in an amount as set forth in Section 19.1 or (ii) impedes the continuance of the Business as conducted on the Signing Date.

20. Compliance

- Unless listed on Exhibit W.20.1, the Business is, and, to the extent any non-compliance in the past is, or may still become relevant as of today, has been in past three (3) years, carried out in compliance with all applicable laws, in particular with laws on corruption, unfair competition (including antitrust law), data protection, money laundering, sanctions or export control.
- 20.2 Neither any Target Group Company nor, to Sellers' Knowledge, any officers, directors or employees of any Target Group Company are subject to or otherwise involved in any administrative or criminal inquiry, investigation or enforcement proceeding regarding any actual or potential infringement of any laws, including any laws on corruption, unfair competition (including antitrust law), data protection, money laundering, sanctions or export control.
- 20.3 No proceeding as referred to in Section 20.2 has been threatened or announced to a Target Group Company, in each case, in writing, and, to Sellers' Knowledge, no circumstances exist that can reasonably be expected to justify or give rise to any such proceeding.

- Neither any Target Group Company nor, to Sellers' Knowledge, any officers, directors or employees of any Target Group Company have during the past three (3) years, directly or indirectly,
 - dd) used funds of any Target Group Company for unlawful contributions, gifts, entertainment or other expenses relating to political activities;
 - ee) made any unlawful payment to foreign or domestic government officials or employees or to foreign or domestic political parties or campaigns from a Target Group Company;
 - ff) established or maintained any unlawful or unrecorded fund of Target Group Company moneys or other assets;
 - gg) made or received of any unlawful bribe, rebate, payoff, influence payment, kickback, or other payment.

21. Compliance with Sanctions

- 21.1 (i) None of the Target Group Companies or any of their Employees is or was a Sanctioned Person (as defined below), and (ii) each of the Target Group Companies and, to Sellers' Knowledge, their Employees, shareholders and agents acting on behalf of a Target Group Company was in compliance with, and has not violated, any of the Sanctions (as defined below) to the extent applicable to the Target Group Companies, and all other applicable economic sanctions laws and regulations of the United States or any other applicable jurisdiction.
- 21.2 (i) None of the Target Group Companies or their Employees, shareholders or agents is currently a Sanctioned Person (as defined below), and (ii) each of the Target Group Companies and, to Sellers' Knowledge, their Employees, shareholders and agents acting on behalf of a Target Group Company and the Seller Group Entities is currently in compliance with, and is currently not violating, any of the Sanctions (as defined below) to the extent applicable to the Target Group Companies and the Seller Group Entities, and all other applicable economic sanctions laws and regulations of the United States or any other applicable jurisdiction.
- 21.3 Each of the Target Group Companies and, to Sellers' Knowledge, their Employees, shareholders and agents acting on behalf of a Target Group Company is in compliance with and, to the extent any non-compliance is or may still become relevant as of today, has not previously violated, the U.S. Bank Secrecy Act, as amended by the USA Patriot Act of 2001, and the U.S. Money Laundering Control Act of 1986, as amended through the date hereof, to the extent applicable to the Target Group Companies and all other applicable anti-money laundering laws and regulations of the United States or any other applicable jurisdiction.
- None of (i) the obligations under this Agreement, (ii) the use of the payments made under this Agreement, (iii) the execution, delivery and performance of this Agreement or (iv) the consummation of any transaction contemplated hereby, or the fulfilment of the terms hereof or thereof, will result in a violation of the Target Group or the Sellers, Seller Group Entities or Seller

Guarantors, of any Sanctions or of any anti-money laundering laws or economic sanctions laws and regulations of the United States or any other applicable jurisdiction.

- 21.5 For the purposes of this Section: "Sanctions" means any of the laws, executive orders, regulations and rules related to sanctions programs administered by the Office of Foreign Assets Control of the U.S. Department of the Treasury ("OFAC"), the Bureau of Industry and Security of the U.S. Department of Commerce ("BIS") or the U.S. Department of State ("State Department") or the United Nations Security Council, the European Union or Her Majesty's Treasury's Office of Financial Sanctions Implementation or other relevant sanctions authority. For ease of reference, and not by way of limitation, OFAC's Sanctions programs are described on OFAC's website at www.treas.gov/ofac. "Sanctioned Person" means any government, country, corporation or other entity, group or individual with whom or which the Sanctions prohibit or restrict a U.S. Person from engaging in transactions, and includes without limitation any individual or corporation or other entity that appears on the OFAC list of Specially Designated Nationals and Blocked Persons (the "SDN List") and other lists maintained by OFAC, the BIS Entity List, Denied Persons List and Unverified List, and the sanctions lists maintained by the State Department, as each such list may be amended from time to time. For ease of reference, and not by way of limitation, OFAC's SDN List may be found on OFAC's website at www.treas.gov/offices/enforcement/ofac/sdn, and other relevant sanctions lists may be found at https://2016.export.gov/ecr/eg main 023148.asp. "U.S. Person" means any U.S. citizen, permanent resident alien, entity organized under the laws of the United States (including foreign branches), or any person (individual or entity) in the United States, and, with respect to OFAC's sanctions programs relating to Cuba and Iran, also includes any corporation or other entity that is owned or controlled by one of the foregoing, without regard to where it is organized or doing business.
- Sections 21.2 and 21.4 shall not apply if and to the extent that such representations, or compliance therewith, would violate, would be unenforceable pursuant to, or would otherwise result in a violation or breach of section 7 of the German Foreign Trade and Payments Ordinance (*Außenwirtschaftsverordnung*), the Blocking Regulation or any other applicable anti-boycott law or regulation implementing the Blocking Regulation in any member state of the European Union or any applicable equivalent blocking legislation adopted by the United Kingdom.

22. Product Liability and Warranty

- 22.1 Except as disclosed in Exhibit W.22.1, during the past five (5) years, no warranty claims (*vertragliche oder gesetzliche Garantieansprüche oder Mängelrechte*) or product liability claims with an amount of more than EUR 50,000 in each case, have been asserted against any Target Group Company, it being understood that all claims resulting from the same or similar circumstances constitute one case.
- Except as disclosed in Exhibit W.22.1, during the past five (5) years, there has been no public or silent recall nor any kind of field or service campaign or similar action by any Target Group Company that is based on any products designed, developed, manufactured or distributed or on

any services rendered by any Target Group Company, in each case, regardless of whether such action was requested by any public authority or customer or based upon own initiative.

22.3 To Sellers' Knowledge, no circumstances exist that do (i) give rise to any (current or future) warranty claims or product liability claims as specified in Section 22.1 or (ii) require (currently or in the future) any Target Group Company under applicable laws (including, but not limited to, product safety laws), industry standards or prudent business practices to perform any of the actions set forth in Section 22.2.

23. Environmental Matters

- The Business is, and, to the extent any non-compliance in the past is or may still become relevant as of today, has been, carried out in compliance with all Environmental Law.
- Except as set forth in Exhibit W.23.2, no Hazardous Materials are, or have been, used, stored, generated, otherwise processed or disposed on, from or under any real property that is owned, leased, or used for the purpose of the Business.
- 23.3 To Sellers' Knowledge, no Target Group Company has received any written order (*Bescheid*) from any public authority imposing investigation (*Untersuchungsmaßnahmen*), remediation (*Sanierungsmaßnahmen*), securing (*Sicherungsmaßnahmen*) or protective containment measures (*Schutz- und Beschränkungsmaßnahmen*) regarding an Environmental Contamination.
- To Sellers' Knowledge, no Target Group Company is responsible (wholly or in part) for any environmental clean-up, investigation (*Untersuchungsmaßnahmen*), remediation (*Sanierungsmaßnahmen*), securing (*Sicherungsmaßnahmen*) or protective containment measures (*Schutz- und Beschränkungsmaßnahmen*) or other corrective action in relation to Environmental Contaminations that exist on or have been caused prior to the Closing Date of the Owned Real Estate, Leased Real Estate or real estate formerly owned, leased or used or any part thereof or any other land or premises that exist or have been caused prior to the Closing Date
- 23.5 No legal or administrative proceeding is pending against any Target Group Company that alleges a violation of any Environmental Law. To Sellers' Knowledge, no such legal or administrative proceeding has been threatened or announced, in each case, in writing, and, to Sellers' Knowledge, no circumstances exist that can reasonably be expected to justify or give rise to any such proceeding.
- 23.6 No Target Group Company is, and has not been or alleged to be, a potentially responsible party pursuant to the Comprehensive Environmental Response, Compensation and Liability Act of 1980, 42 U.S.C. § 9601, et seq., as amended or any comparable state law.
- 23.7 To Sellers' knowledge, there are no, and there has been no release of, Hazardous Materials in material violation of, or that may give rise to material liability or obligation under, Environmental Laws at any property that is or formerly owned, leased, or used for the purpose of the Business.

24. Taxes

- 24.1 All Tax returns required to be filed with any Taxing Authority by or on behalf of the Target Group Companies have been prepared in accordance with applicable laws and regulations, were true, correct and complete when filed and have been filed when due. No Target Group Company is the beneficiary of any extension of time within which to file any Tax return (except for general extensions granted by the relevant authority, e.g. in connection with the COVID-19 pandemic).
- 24.2 The Target Group Companies have timely paid when due all Taxes owed by them, irrespective of whether or not such Taxes were presented on a Tax return.
- 24.3 Each Target Group Company has withheld and paid all Taxes required to have been withheld and paid in connection with any amounts paid or owing to any employee, independent contractor, creditor, stockholder, or other third party.
- 24.4 Each Target Group Company has properly (i) collected and remitted sales, value added and similar Taxes with respect to sales or leases made to or services provided to, its customers, and (ii) for all sales, leases or services that are exempt from sales, value added and similar Taxes and that were made without charging or remitting sales, value added or similar Taxes, received and retained any appropriate tax exemption certificates and other documentation qualifying such sale as exempt.
- 24.5 No Tax audit, investigation, dispute or other proceeding is pending in respect of the Target Group Companies. The Target Group Companies have not (a)been notified by any Taxing Authority in writing that such authority intends to commence any such proceeding or (b)received a written request for information related to Tax matters or a notice of deficiency or proposed adjustment for any amount of Tax proposed, asserted, or assessed by any taxing authority against a Target Group Company.
- 24.6 Each Target Group Company is Tax resident exclusively in the country of its statutory seat and is effectively managed and controlled exclusively from the country of its statutory seat.
- 24.7 No Target Group Company has any permanent establishment, and no claim has been made by any Taxing Authority against any Target Group Company in a jurisdiction where a Target Group Company does not file a Tax return that a Target Group Company is or may be subject to taxation by that jurisdiction.
- 24.8 No Target Group Company is registered for VAT purposes outside the country of its statutory seat.
- 24.9 There are no employment severance or termination agreements, other compensation arrangements or employee benefit plans in effect which provide for the payment of any amount by any Target Group Company to any employee due to the execution or consummation of this Agreement or the COSMIQ Option Agreement of which any employee of any Target Group Company is a party. No Tax needs to be withheld or disbursed by any Target Group Company in connection with any

payment by any third party to an employee of any Target Group Company due to the execution or consummation of this Agreement or the COSMIQ Option Agreement.

- 24.10 The Target Group Companies have not obtained a binding ruling and have not entered into any agreements, waivers or arrangements with any Taxing Authority, and no Target Group Company is a party to or bound by any Tax allocation, indemnification or sharing agreement. No Target Company has waived any statute of limitations in respect of Taxes or agreed to any extension of time with respect to a Tax assessment or deficiency.
- 24.11 The Target Group Companies keep books of accounts as required by applicable law, and the application thereof by the competent Taxing Authority, and have sufficient records relating to past events during all times prior to and including the Effective Date. In particular, all documentation duties as required by section 90 paragraph 3 of the German Fiscal Code (*Abgabenordnung*) or any other comparable foreign law are fulfilled by any Target Group Company in respect of the determination of transfer prices, the attention to the arm's length principle and the validity of any cost sharing agreements.
- 24.12 No Target Group Company receives passive income which may be subject to taxation in Germany according to the German Controlled Foreign Company Act (*Außensteuergesetz*) or any comparable foreign applicable law.
- 24.13The Target Group Companies have not written down any asset to its going concern value (keine Teilwertabschreibung).
- 24.14 The Target Group Companies have not been involved in any reorganisation that could lead to blocking periods or any similar restrictions.
- 24.15 There are no liens or other encumbrances for Taxes (other than Taxes not yet due and payable) upon any of the assets of a Target Group Company.
- 24.16 No Target Group Company has made an affirmative U.S. entity classification election pursuant to Treas. Reg. Secs. 301.7701-2 and 301.7701-3 for U.S. federal income tax purposes.
- 24.17 No Target Group Company is a party to any agreement, contract, arrangement or plan that has resulted or could result, separately or in the aggregate, in the payment of (a) any "excess parachute payment" within the meaning of Code §280G (or any corresponding provision of state, local, or non-U.S. Tax law) or (b) any amount that will not be fully deductible as a result of Code §162(m) (or any corresponding provision of state, local, or non-U.S. Tax law).
- 24.18 No Target Group Company (i) has been a member of an affiliated, consolidated, combined, unitary, or similar group for Tax purposes (other than a group the common parent of which was a Target Group Company) or (ii) has any liability for the Taxes of any person or entity under Reg. §1.1502-6 (or any similar provision of state, local, or non-U.S. law), as a transferee or successor, by contract, or otherwise.

- 24.19 No Target Group Company will be required to include any item of income in, or exclude any item of deduction from, taxable income for any taxable period (or portions thereof) ending after the Closing Date as a result of any: (i) change in method of accounting for a taxable period ending on or prior to the Closing Date; (ii) use of an improper method of accounting for a taxable period ending on or prior to the Closing Date; (iii) "closing agreement" as described in Code § 7121 (or any corresponding or similar provision of state, local, or non-U.S. income Tax law) executed on or prior to the Closing Date; (iv) intercompany transaction or excess loss account described in Treasury Regulations under Code § 1502 (or any corresponding or similar provision of state, local, or non-U.S. income Tax law); (v) instalment sale or open transaction disposition made on or prior to the Closing Date; (vi) prepaid amount or deferred revenue received or accrued on or prior to the Closing Date; or (vii) income inclusion pursuant to Sections 951 or 951A of the Code with respect to any interest held in a "controlled foreign corporation" (as that term is defined in Section 957 of the Code) on or before the Closing Date. No Target Group Company (i) is required to include any amount in income pursuant to Section 965 of the Code or pay any installment of the "net tax liability" described in Section 965(h)(1) of the Code or (ii) uses, or has used, the cash receipts and disbursements method of accounting for income Tax purposes.
- 24.20 No Target Group Company (i) has distributed stock of another Person, or has had its stock distributed by another Person, in a transaction that was purported or intended to be governed in whole or in part by Code §355 or Code §361, (ii) is or has been a party to any "reportable transaction," as defined in Code §6707A(c)(1) and Reg. §1.6011-4(b), (iii) has a permanent establishment (within the meaning of an applicable Tax treaty) or otherwise has an office or fixed place of business in a country other than the country in which it is organized, or (iv) has received any letter ruling from the Internal Revenue Service (or any comparable ruling from any other taxing authority).
- 24.21 No Target Group Company has (a)elected to defer the payment of any "applicable employment taxes" (as defined in Section 2302(d)(1) of the CARES Act) pursuant to Section 2302 of the CARES Act or (b)claimed any "employee retention credit" pursuant to Section 2301 of the CARES Act.
- 24.22 AF&R Mexico has complied with, all applicable Mexican Tax laws, and with all requirements in order to be eligible for maquiladora status, including compliance with the Mexican Income Tax Law (*Ley del Impuesto Sobre la Renta*) and any other applicable Tax regulations, and AF&R Mexico has appropriate and legally required supporting documentation to demonstrate proper compliance with such provisions.

25. Customers and Vendors

25.1 <u>Exhibit W.25.1</u> sets forth (i) a true, correct and complete list of the top 10 customers of the Business (the "**Material Vendors**") and the top 10 vendors of the Business (the "**Material Vendors**") as of 31 December 2021, based on, and together with a listing of, revenue of the Target

Group Companies from each such Material Customer and purchases by the Target Group Companies from each such Material Vendor for the fiscal year ended 31 December 2021.

- 25.2 Since 1 January 2022, no Material Vendor has (i) materially modified or terminated its relationship with any Target Group Company or has decreased materially (except for decreases occurring in the ordinary course of business) its services, supplies, materials or finished products to any Target Group Company, and to Sellers' Knowledge, no such person or entity has threatened in writing any such action in this clause (i), or (ii) given any Target Group Company written notice of modification or termination or intent to modify or terminate its relationship with any Target Group Company.
- 25.3 Since 1 January 2022, no Material Customer has (i) materially modified or terminated its relationship with any Target Group Company or has decreased materially (except for decreases or limits occurring in the Ordinary Course) its usage or purchase of the products of any Target Group Company, and to Sellers' Knowledge, no such person or entity has threatened any such action in this clause (i) in writing, or (ii) given any Target Group Company notice in writing of modification or termination or intent to modify or terminate its relationship with any Target Group Company.
- To Seller's Knowledge, no Target Group Company is, or at any time in the past three years has been, in material violation with any Material Customer's ethics policies applicable to its customers.
- 25.5 No Target Group Company has agreed in writing with any Material Customer to repurchase, or issue a credit or allow a return in respect of, any products or services of any Target Group Company following the Closing Date, except to the extent such products or services fail to meet specifications.

* * * *

This deed - including Parts 1-IV, including Exhibit 23.1.1-A to Part I - was read aloud to the persons appearing, approved by them and signed by them and the acting notary in their own hands as follows at 05:11 a.m. CEST:

/s/ Markus Gebhardt /s/ Andreas Gebhardt /s/ Julius Scheifele /s/ Malte Schwertmann /s/ Sebastian Herrler, Notary

CERTIFICATION

- I, Phillip Eyler, certify that:
- 1. I have reviewed this quarterly report on Form 10-Q of Gentherm Incorporated;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

/s/ Phillip Eyler

Phillip Eyler
President and Chief Executive Officer
May 4, 2022

CERTIFICATION

- I, Matteo Anversa, certify that:
- 1. I have reviewed this quarterly report on Form 10-Q of Gentherm Incorporated;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

/s/ Matteo Anversa

Matteo Anversa Executive Vice President, Chief Financial Officer and Treasurer May 4, 2022

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of Gentherm Incorporated (the "Company") on Form 10-Q for the period ended March 31, 2022 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Phillip Eyler, President and Chief Executive Officer of the Company, certify, pursuant to 18 U.S.C. § 1350, as adopted pursuant to § 906 of the Sarbanes-Oxley Act of 2002, that:

- (1). The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2). The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

/s/ Phillip Eyler
Phillip Eyler
President and Chief Executive Officer
May 4, 2022

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of Gentherm Incorporated (the "Company") on Form 10-Q for the period ended March 31, 2022 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Matteo Anversa, Chief Financial Officer of the Company, certify, pursuant to 18 U.S.C. § 1350, as adopted pursuant to § 906 of the Sarbanes-Oxley Act of 2002, that:

- (1). The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2). The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

/s/ Matteo Anversa

Matteo Anversa Executive Vice President, Chief Financial Officer and Treasurer May 4, 2022

Document and Entity 3 Months Ended Information - shares Mar. 31, 2022 Apr. 29, 2022

Cover [Abstract]

Document Type 10-Q
Amendment Flag false

Document Period End Date Mar. 31, 2022

Document Fiscal Year Focus2022Document Fiscal Period FocusQ1Trading SymbolTHRM

Entity Registrant Name GENTHERM INCORPORATED

Entity Central Index Key 0000903129

Entity Current Reporting Status Yes
Current Fiscal Year End Date --12-31

Entity Filer Category Large Accelerated Filer

Entity Small BusinessfalseEntity Emerging Growth Companyfalse

Entity Common Stock, Shares Outstanding 33,131,660

Entity Shell CompanyfalseEntity File Number0-21810Entity Tax Identification Number95-4318554

Entity Address, Address Line One 21680 Haggerty Road

Entity Address, City or Town Northville

Entity Address, State or Province MI
Entity Address, Postal Zip Code 48167
City Area Code 248

<u>Local Phone Number</u> 504-0500 Entity Interactive Data Current Yes

Entity Incorporation, State or Country Code MI

Document Quarterly Report true

Security Exchange Name NASDAQ

<u>Title of 12(b) Security</u> Common Stock, no par value

<u>Document Transition Report</u> false

Current Assets:	
Cash and cash equivalents \$ \$	_
177,917 190,606	
Accounts receivable, net 207,351 182,987	7
Inventory:	
<u>Raw materials</u> 111,621 96,426	
<u>Work in process</u> 10,991 9,495	
<u>Finished goods</u> 53,556 53,556	
<u>Inventory, net</u> 176,168 159,477	7
Other current assets 41,034 32,775	
<u>Total current assets</u> 602,470 565,845	5
Property and equipment, net 152,288 155,270)
Goodwill 64,979 66,033	
Other intangible assets, net 34,982 37,554	
Operating lease right-of-use assets 27,445 24,387	
<u>Deferred income tax assets</u> 69,420 69,630	
Other non-current assets 15,529 16,624	
<u>Total assets</u> 967,113 935,343	3
Current Liabilities:	
<u>Accounts payable</u> 156,236 122,727	7
<u>Current lease liabilities</u> 5,920 5,669	
<u>Current maturities of long-term debt</u> 2,500 2,500	
Other current liabilities 79,044 82,193	
Total current liabilities 243,700 213,089	9
Long-term debt, less current maturities 36,250 36,250	
Non-current lease liabilities 19,978 19,789	
Pension benefit obligation 6,362 6,832	
Other non-current liabilities 5,036 5,577	
<u>Total liabilities</u> 311,326 281,537	7
Common Stock:	
No par value; 55,000,000 shares authorized 33,127,531 and 33,008,185 issued and	
outstanding at March 31, 2022 and December 31, 2021, respectively)
<u>Paid-in capital</u> 5,720 5,866	
Accumulated other comprehensive loss (45,728) (36,922)	2)
Accumulated earnings 577,963 566,216	6
Total shareholders' equity 655,787 653,806	6
Total liabilities and shareholders' equity \$	
967,113 935,343	3

Consolidated Condensed Balance Sheets

(Parenthetical) - \$ / shares

Statement Of Financial Position [Abstract]

Common Stock, par value

Common Stock, shares authorized	55,000,000	55,000,000
Common Stock, shares issued	33,127,531	33,008,185
Common Stock, shares outstanding	33,127,531	33,008,185

Mar. 31, 2022 Dec. 31, 2021

Consolidated Condensed Statements of Income - USD

3 Months Ended

Consolidated Condensed	5 Months Ended				
Statements of Income - USD					
(\$)	Mar. 31, 2022	Mar. 31, 2021			
\$ in Thousands	ŕ	,			
Income Statement [Abstract]					
Product revenues	\$ 267,657	\$ 288,535			
Type of Revenue [Extensible List]	us-gaap:ProductMemb	er us-gaap:ProductMember			
Cost of sales	\$ 203,544	\$ 200,866			
Gross margin	64,113	87,669			
Operating expenses:					
Net research and development expenses	20,434	17,603			
Selling, general and administrative expenses	29,308	28,526			
Restructuring expenses	181	791			
Total operating expenses	49,923	46,920			
Operating income	14,190	40,749			
Interest expense, net	(569)	(1,039)			
Foreign currency gain	2,217	773			
Other income (loss)	204	(9)			
Earnings before income tax	16,042	40,474			
Income tax expense	4,295	7,565			
Net income	\$ 11,747	\$ 32,909			
Basic earnings per share	\$ 0.36	\$ 1.00			
Diluted earnings per share	\$ 0.35	\$ 0.99			
Weighted average number of shares – basic	33,034,872	32,945,970			
Weighted average number of shares – diluted	133,376,764	33,389,591			

Consolidated Condensed 3 Months Ended **Statements of Comprehensive Income -**Mar. 31, 2022 Mar. 31, 2021 **USD (\$)** \$ in Thousands **Statement Of Income And Comprehensive Income [Abstract]** \$ 11,747 \$ 32,909 Net income **Other comprehensive loss:** (29) Pension benefit obligations Foreign currency translation adjustments (9,294)(13,152)Unrealized gain (loss) on foreign currency derivative securities, net of tax 464 (522)Unrealized loss on commodity derivative securities, net of tax Other comprehensive loss, net of tax (8,806)(13,674)Comprehensive income \$ 2,941 \$ 19,235

Consolidated Condensed	3 Months Ended			
Statements of Cash Flows - USD (\$) \$ in Thousands	Mar. 31, 2022	Mar. 31, 2021		
Operating Activities:				
Net income	\$ 11,747	\$ 32,909		
Adjustments to reconcile net income to net cash provided by operating				
activities:				
Depreciation and amortization	9,577	9,854		
Deferred income taxes	(778)	105		
Non-cash stock based compensation	2,279	4,460		
Loss on disposition of property and equipment	107	242		
<u>Other</u>	256	(103)		
Changes in assets and liabilities:				
Accounts receivable, net	(25,788)	(13,931)		
Inventory	(18,116)	(11,546)		
Other assets	(10,716)	3,243		
Accounts payable	34,097	18,113		
Other liabilities	(3,349)	(3,679)		
Net cash (used in) provided by operating activities	(684)	39,667		
Investing Activities:				
Purchases of property and equipment	(5,659)	(9,913)		
Proceeds from the sale of property and equipment	52	10		
Cost of technology investments	(350)	(200)		
Net cash used in investing activities	(5,957)	(10,103)		
Financing Activities:				
Repayments of debt		(130,000)		
Proceeds from the exercise of Common Stock options	569	5,984		
Taxes withheld and paid on employees' share-based payment awards	(4,319)	(1,532)		
Acquisition contingent consideration payment		(68)		
Net cash used in financing activities	(3,750)	(125,616)		
Foreign currency effect	(2,298)	(1,338)		
Net decrease in cash and cash equivalents	(12,689)	(97,390)		
Cash and cash equivalents at beginning of period	190,606	268,345		
Cash and cash equivalents at end of period	177,917	170,955		
Supplemental disclosure of cash flow information:				
Cash paid for taxes	3,267	2,555		
<u>Cash paid for interest</u>	\$ 421	\$ 844		

Consolidated Condensed Statements of Changes In Shareholders' Equity - USD (\$)	Total	Common Stock	Paid-in Capital	Accumulated Other Comprehensive Loss	Accumulated Earnings
\$ in Thousands					
Beginning Balance at Dec. 31, 2020	\$ 586,331	\$ 121,073	\$ 7,458	\$ (14,982)	\$ 472,782
Beginning Balance (in shares) at Dec. 31, 2020		32,921,000)		
Net income	32,909				32,909
Other comprehensive loss	(13,674)			(13,674)	
Stock compensation, net	7,192	\$ 8,527	(1,335)		
Stock compensation, net, shares		190,000			
Ending Balance at Mar. 31, 2021	612,758	\$ 129,600	6,123	(28,656)	505,691
Ending Balance (in shares) at		33,111,000)		
Mar. 31, 2021		55,111,000	,		
Beginning Balance at Dec. 31, 2021	\$ 653,806	\$ 118,646	5,866	(36,922)	566,216
Beginning Balance (in shares) at Dec. 31, 2021	33,008,185	5 33,008,000)		
Net income	\$ 11,747				11,747
Other comprehensive loss	(8,806)			(8,806)	
Stock compensation, net	(960)	\$ (814)	(146)		
Stock compensation, net, shares		119,000			
Ending Balance at Mar. 31, 2022	\$ 655,787	\$ 117,832	\$ 5,720	\$ (45,728)	\$ 577,963
Ending Balance (in shares) at Mar. 31, 2022	33,127,531	1 33,127,000)		

Overview

3 Months Ended Mar. 31, 2022

Organization Consolidation
And Presentation Of
Financial Statements
[Abstract]
Overview

Note 1 – Overview

Gentherm Incorporated, a Michigan corporation, and its consolidated subsidiaries ("Gentherm", "we", "us", "our" or the "Company") is a global developer, manufacturer and marketer of innovative thermal management technologies for a broad range of heating, cooling and temperature control applications, primarily in the automotive and medical industries. Within the automotive industry, our products provide solutions for passenger climate comfort and convenience, battery thermal management and cell connecting systems. Within the medical industry our products provide patient temperature management solutions. Our automotive products can be found on vehicles manufactured by nearly all the major automotive original equipment manufacturers ("OEMs") operating in North America and Europe, and several major OEMs in Asia. We operate in locations aligned with our major customers' product strategies to provide locally enhanced design, integration and production capabilities. The Company is also developing a number of new technologies and products that are expected to enable improvements to existing products and to create new product applications for existing and new markets.

Basis of Presentation and Significant Accounting Policies

The unaudited consolidated condensed financial statements included herein have been prepared in accordance with accounting principles generally accepted in the United States of America ("U.S. GAAP") for interim financial information and pursuant to the rules and regulations of the U.S. Securities and Exchange Commission. Certain information and footnote disclosures normally included in financial statements prepared in accordance with U.S. GAAP have been condensed or omitted pursuant to those rules and regulations. The information furnished in the consolidated condensed financial statements include all adjustments (consisting of only normal, recurring adjustments), considered necessary to present fairly the results of operations, financial position and cash flows of the Company. These financial statements should be read in conjunction with the Company's Annual Report on Form 10-K for the year ended December 31, 2021. The operating results for interim periods are not necessarily indicative of results that may be expected for any other interim period or for the full year.

In preparing these financial statements, management was required to make estimates and assumptions that affect the reported amounts of assets, liabilities, revenues and expenses. These estimates and assumptions are based on our historical experience, the terms of existing contracts, our evaluation of trends in the industry, information provided by our customers and suppliers and information available from other outside sources, as appropriate. These estimates and assumptions are subject to an inherent degree of uncertainty. We are not presently aware of any events or circumstances that would require us to update such estimates and assumptions or revise the carrying value of our assets or liabilities. Our estimates may change, however, as new events occur and additional information is obtained. As a result, actual results may differ significantly from our estimates, and any such differences may be material to our financial statements.

Principles of Consolidation

The consolidated condensed financial statements include the accounts of the Company, its wholly owned subsidiaries and those entities in which it has a controlling financial interest. The Company evaluates its relationship with other entities for consolidation and to identify whether such entities are variable interest entities ("VIE") and to assess whether the Company is the primary beneficiary of such entities. Investments in affiliates in which Gentherm does not have control but does have the ability to exercise significant influence over operating and financial policies are accounted for under the equity method. When Gentherm does not have the ability to

exercise significant influence (generally when ownership interest is less than 20%), investments in affiliates are measured at cost, less impairments, adjusted for observable price changes in orderly transactions for identical or similar investments of the same issuer.

Variable Interest Entities

The Company maintains an ownership interest in a VIE, Carrar Ltd. ("Carrar"). Carrar is a technology developer of advanced thermal management systems for the electric mobility market. The Company determined that Carrar is a VIE; however, the Company does not have a controlling financial interest or have the power to direct the activities that most significantly affect the economic performance of the investment. Therefore, the Company has concluded that it is not the primary beneficiary. Gentherm's investment in Carrar is measured at cost, less impairments, adjusted for observable price changes in orderly transactions for identical or similar investments of the same issuer. The Carrar investment was \$5,200 as of March 31, 2022 and December 31, 2021, and is recorded in Other non-current assets in the consolidated condensed balance sheet.

Revenue Recognition

The Company has no material contract assets or contract liabilities as of March 31, 2022.

The Company recognizes an asset for the incremental costs of obtaining a contract with a customer if the benefits of those costs are expected to be realized for a period greater than one year. Total capitalized costs to obtain a contract were \$1,884 and \$1,946 as of March 31, 2022 and December 31, 2021, respectively. These amounts are recorded in Other non-current assets and are being amortized into Product revenues over the expected production life of the applicable program.

Segment Reporting

The Company has two reportable segments: Automotive and Medical.

The Automotive reporting segment is comprised of the results from our global automotive businesses, including the design, development, manufacturing and sales of automotive climate comfort systems, automotive cable systems, battery performance solutions, and automotive electronic and software systems.

The Medical reporting segment is comprised of the results from the patient temperature management business in the medical industry. Patient temperature management includes temperature management systems across multiple product categories addressing the needs of hyper-hypothermia therapy in intensive care, normothermia in surgical procedures and additional warming/cooling therapies utilized in acute care, ambulatory, clinics and home health.

New Accounting Pronouncements

New Accounting
Pronouncements And
Changes In Accounting
Principles [Abstract]
New Accounting
Pronouncements

3 Months Ended Mar. 31, 2022

Note 2 – New Accounting Pronouncements

Reference Rate Reform

In March 2020, the FASB issued ASU 2020-04, "Reference Rate Reform (Topic 848): Facilitation of Effects of Reference Rate Reform on Financial Reporting". ASU 2020-04 provides practical expedients and exceptions for applying GAAP to contracts, hedging relationships, and other transactions affected by reference rate reform if certain criteria are met. The expedients and exceptions provided by the amendments in this update apply only to contracts, hedging relationships, and other transactions that reference the London Interbank Offered Rate ("LIBOR") or another reference rate expected to be discontinued as a result of reference rate reform. These amendments are not applicable to contract modifications made and hedging relationships entered into or evaluated after December 31, 2022. In January 2021, the FASB subsequently issued ASU 2021-01, "Reference Rate Reform (Topic 848): Scope" to clarify that certain optional expedients and exceptions in Topic 848 for contract modifications and hedge accounting apply to derivatives that are affected by the discounting transition. ASU 2020-04 and ASU 2021-01 are effective as of March 12, 2020 through December 31, 2022 and may be applied retrospectively to contract modifications and hedging relationships from the beginning of an interim period that includes or is subsequent to March 12, 2020, or on a prospective basis to new modifications from any date within an interim period that includes or is subsequent to the date of the issuance of a final update, up to the date that financial statements are available to be issued. The adoption of this standard is not expected to have a material impact on the Company's financial statements.

Restructuring

3 Months Ended Mar. 31, 2022

Restructuring And Related Activities [Abstract] Restructuring

Note 3 – Restructuring

Manufacturing Footprint Rationalization

In September 2019, the Company committed to a restructuring plan ("Plan") to improve the Company's manufacturing productivity and rationalize its footprint. Under this Plan, the Company is relocating and consolidating certain automotive electronics manufacturing plants in North America and China. During 2021, the Company completed the closures and relocation of its automotive electronics manufacturing operations from Burlington, Canada to Celaya, Mexico and from Longgang, Shenzhen, China to Bantian, Shenzhen, China. As of March 31, 2022, the electronics manufacturing in Acuña, Mexico continues to transition to Celaya, Mexico.

During the three months ended March 31, 2022, the Company recognized restructuring expense of \$50 for employee separation costs and \$101 for other costs. During the three months ended March 31, 2021, the Company recognized restructuring expense of \$206 for employee separation costs and \$259 for other costs.

The Company has recorded approximately \$10,256 of restructuring expenses since the inception of this program and as of March 31, 2022, \$538 remains accrued. Actions under the Plan are expected to be substantially completed by the end of 2022 and future expenses are expected to be less than \$1,000.

Other Restructuring Activities

As part of the Company's continued efforts to optimize its cost structure, the Company has undertaken several discrete restructuring actions. During the three months ended March 31, 2022, the Company recognized \$30 for other costs. During the three months ended March 31, 2021, the Company recognized \$326 for employee separation costs. These restructuring expenses were primarily associated with restructuring actions focused on the rotation of our manufacturing footprint to best cost locations and the reduction of global overhead costs.

Restructuring Expenses By Reporting Segment

The following table summarizes restructuring expense for the three months ended March 31, 2022 and 2021 by reporting segment:

Thre			l March	
2	022	2	2021	
\$	181	\$	791	
	_		_	
	_		_	
\$	181	\$	791	
		\$\frac{2022}{\$\times 181}\$	\$ 181 \$ ————————————————————————————————————	

Restructuring Liability

Restructuring liabilities are classified as Other current liabilities in the consolidated condensed balance sheets. The following table summarizes restructuring liability for the three months ended March 31, 2022:

	Employee	Other		
	Separation	Related		
	Costs	Costs	Total	
Balance at December 31, 2021	\$ 1,494	\$ —	\$ 1,494	

Additions, charged to restructuring expenses	_	131	131
Cash payments	(544)	(131)	(675)
Non-cash utilization			_
Change in estimate	50	_	50
Currency translation	(18)		(18)
Balance at March 31, 2022	\$ 982	\$	\$ 982

Details of Certain Balance Sheet Components

3 Months Ended Mar. 31, 2022

Balance Sheet Components [Abstract]

Details of Certain Balance Sheet Components

Note 4 – Details of Certain Balance Sheet Components

	M	March 31, 2022		ecember 1, 2021
Other current assets:				
Notes receivable	\$	15,958	\$	13,033
Income tax and other tax receivable		12,501		10,681
Billable tooling		4,993		3,778
Prepaid expenses		4,751		3,407
Other		2,831		1,876
Total other current assets	\$	41,034	\$	32,775
Other current liabilities:				
Liabilities from discounts and rebates	\$	26,632	\$	27,343
Accrued employee liabilities		21,244		28,818
Income tax and other taxes payable		19,529		17,068
Restructuring		982		1,494
Accrued warranty		1,772		1,916
Other		8,885		5,554
Total other current liabilities	\$	79,044	\$	82,193

Goodwill and Other Intangibles

Goodwill And Intangible
Assets Disclosure [Abstract]

Goodwill and Other Intangibles

3 Months Ended Mar. 31, 2022

Note 5 - Goodwill and Other Intangibles

Goodwill

Changes in the carrying amount of goodwill, by reportable segment, for the three months ended March 31, 2022 was as follows:

	Automotive		 Medical	Total	
'Balance as of December 31, 2021	\$	37,329	\$ 28,704	\$	66,033
Exchange rate impact		(762)	(292)		(1,054)
Balance as of March 31, 2022	\$	36,567	\$ 28,412	\$	64,979

Other Intangible Assets

Other intangible assets and accumulated amortization balances as of March 31, 2022 and December 31, 2021 were as follows:

	March 31, 2022				De	ecen	iber 31, 2021	1
	Gross Carrying Value		cumulated nortization	Net Carrying Value	Gross Carrying Value		cumulated nortization	Net Carrying Value
Definite-lived:								
Customer relationships	\$ 87,854	\$	(63,445)	\$24,409	\$ 90,448	\$	(64,105)	\$26,343
Technology	28,706		(24,288)	4,418	29,464		(24,487)	4,977
Product development								
costs	19,661		(19,183)	478	20,329		(19,772)	557
Software development	1,007		_	1,007	1,007		_	1,007
Indefinite-lived:								
Tradenames	4,670		_	4,670	4,670		_	4,670
Total	\$141,898	\$	(106,916)	\$34,982	\$145,918	\$	(108,364)	\$37,554

In addition to annual impairment testing, which is performed in the fourth quarter of each fiscal year, the Company continuously monitors for events and circumstances that could negatively impact the key assumptions used in determining fair value and therefore require interim impairment testing, including long-term revenue growth projections, profitability, discount rates, recent market valuations from transactions by comparable companies, volatility in the Company's market capitalization, and general industry, market and macroeconomic conditions. We are not presently aware of any events or circumstances that would require us to revise the carrying value of our assets or liabilities as of March 31, 2022.

Debt

<u>Debt Disclosure [Abstract]</u> Debt

3 Months Ended Mar. 31, 2022

Note 6 - Debt

The following table summarizes the Company's debt as of March 31, 2022 and December 31, 2021:

	March 3	1, 2022	December 31, 2021		
	Interest Rate	Principal Balance	Interest Rate	Principal Balance	
Amended Credit Agreement:					
U.S. Revolving Note (U.S. Dollar denominations)	1.71%	\$ 35,000	1.35%	\$ 35,000	
U.S. Revolving Note (Euro denominations)	_	_	_	_	
DEG Vietnam Loan	5.21%	3,750	5.21%	3,750	
Total debt		38,750		38,750	
Current maturities		(2,500)		(2,500)	
Long-term debt, less current maturities		\$ 36,250		\$ 36,250	

Credit Agreement

On June 27, 2019, the Company entered into an Amended and Restated Credit Agreement (the "Credit Agreement") with a consortium of lenders and Bank of America, N.A. as administrative agent, which includes a revolving credit note ("U.S. Revolving Note"). The Credit Agreement has a maximum borrowing capacity of \$475,000 and matures on June 27, 2024. The Credit Agreement also provides \$15,000 availability for the issuance of letters of credit and a maximum of \$40,000 for swing line borrowing. Any amount of the facility utilized for letters of credit or swing line loans outstanding will reduce the amount available under the Credit Agreement. The Company had no outstanding letters of credit issued under the Credit Agreement as of March 31, 2022 and December 31, 2021.

The U.S. borrowers and guarantors participating in the Credit Agreement also entered into a related amended and restated pledge and security agreement. The amended and restated pledge and security agreement grants a security interest to the lenders in substantially all of the personal property of the Company and its U.S. subsidiaries designated as borrowers to secure their respective obligations under the Credit Agreement, including the stock and membership interests of specified subsidiaries (limited to 66% of the stock in the case of certain non-U.S. subsidiaries). In addition to the security obligations, all obligations under the Credit Agreement are unconditionally guaranteed by certain of the Company's subsidiaries. The Credit Agreement restricts, among other things, the amount of dividend payments the Company can make to shareholders.

The Credit Agreement contains covenants, that, among other things, (i) prohibit or limit the ability of the borrowers and any material subsidiary to incur additional indebtedness, create liens, pay dividends, make certain types of investments (including acquisitions), enter into certain types of transactions with affiliates, prepay other indebtedness, sell assets, merge with other companies or enter into certain other transactions outside the ordinary course of business, and (ii) require that Gentherm maintain a minimum Consolidated Interest Coverage Ratio and Consolidated Leverage Ratio (based on consolidated EBITDA for the applicable trailing 12-month period as defined in the Credit Agreement) as of the end of any fiscal quarter. The Credit Agreement also contains customary events of default. As of March 31, 2022, the Company was in compliance with the terms of the Credit Agreement.

Under the Credit Agreement, U.S. Dollar denominated loans bear interest at either a base rate ("Base Rate Loans") or Eurocurrency rate ("Eurocurrency Rate Loans"), plus a margin ("Applicable Rate"). The rate for Base Rate Loans is equal to the highest of the Federal Funds

Rate (0.33% at March 31, 2022) plus 0.50%, Bank of America's prime rate (3.50% at March 31, 2022), or the Eurocurrency rate plus 1.00%. The rate for Eurocurrency Rate Loans denominated in U.S. Dollars is equal to the London Interbank Offered Rate (0.45% at March 31, 2022). All loans denominated in a currency other than the U.S. Dollar must be Eurocurrency Rate Loans. Interest is payable at least quarterly.

The Applicable Rate varies based on the Consolidated Leverage Ratio reported by the Company. As long as the Company is not in default of the terms and conditions of the Credit Agreement, the lowest and highest possible Applicable Rate is 1.25% and 2.25%, respectively, for Eurocurrency Rate Loans and 0.25% and 1.25%, respectively, for Base Rate Loans.

As of March 31, 2022, \$35,000 was outstanding under the Credit Agreement. Borrowing availability is subject to, among other things, the Company's compliance with the minimum Consolidated Interest Coverage Ratio and Consolidated Leverage Ratio as of the end of any fiscal quarter. Based upon consolidated EBITDA for the trailing twelve months calculated for purposes of the Consolidated Leverage Ratio, \$424,521 remained available as of March 31, 2022 for additional borrowings under the Credit Agreement subject to specified conditions that Gentherm currently satisfies.

DEG Vietnam Loan

The Company also has a fixed interest rate loan with the German Investment Corporation ("DEG"), a subsidiary of KfW Banking Group, a Germany government-owned development bank. The fixed interest rate senior loan agreement with DEG was used to finance the construction and set up of the Vietnam production facility ("DEG Vietnam Loan"). The DEG Vietnam Loan is subject to semi-annual principal payments that began November 2017 and will end May 2023. Under the terms of the DEG Vietnam Loan, the Company must maintain a minimum Enhanced Equity Ratio, as defined by the DEG Vietnam Loan agreement, based on the financial statements of Gentherm's wholly owned subsidiary, Gentherm Vietnam Co. Ltd. As of March 31, 2022, the Company was in compliance with the terms of the DEG Vietnam Loan.

The scheduled principal maturities of our debt as of March 31, 2022 were as follows:

	DEG Vietnam Note	F	U.S. Revolving Note	Total
2022	\$ 2,500	\$	_	\$ 2,500
2023	1,250		_	1,250
2024	 		35,000	35,000
Total	\$ 3,750	\$	35,000	\$ 38,750

Commitments and Contingencies

Commitments And
Contingencies Disclosure
[Abstract]
Commitments and
Contingencies

3 Months Ended Mar. 31, 2022

Note 7 – Commitments and Contingencies

Legal and other contingencies

The Company may be subject to various legal actions and claims in the ordinary course of its business, including those arising out of breach of contracts, intellectual property rights, environmental matters, regulatory matters and employment-related matters. The Company establishes accruals for matters which it believes that losses are probable and can be reasonably estimated. Although it is not possible to predict with certainty the outcome of these matters, the Company is of the opinion that the ultimate resolution of these matters will not have a material adverse effect on its consolidated results of operations or financial position.

Product Liability and Warranty Matters

In the event that the Company's products fail to perform as expected or result in alleged bodily injury or property damage, our products may subject us to warranty claims and product liability. If any of our products are or are alleged to be defective, we may be required to participate in a recall or other corrective action involving such products. The Company maintains liability insurance coverage at levels based on commercial norms and historical claims experience. The Company can provide no assurances that it will not experience material claims in the future or that it will not incur significant costs to defend such claims.

The Company accrues warranty obligations for products sold based on management estimates of future failure rates and current claim cost experience, with support from the sales, engineering, quality and legal functions. Using historical information available to the Company, including claims already filed by customers, the warranty accrual is adjusted quarterly to reflect management's best estimate of future claims.

The following is a reconciliation of the changes in accrued warranty costs:

	Three Months Ended March 31,				
	2022		2021		
Balance at the beginning of the period	\$	1,916	\$	2,391	
Warranty claims paid		(334)		(352)	
Warranty expense for products shipped during the current					
period		354		519	
Adjustments to warranty estimates from prior periods		(152)		271	
Adjustments due to currency translation		(12)		(31)	
Balance at the end of the period	\$	1,772	\$	2,798	

Other matters

Purchase commitments for materials, supplies, services and capital expenditures, as part of the normal course of business, are generally consistent from year to year. In addition, due to supply shortages of semiconductors, the Company has entered into agreements with various suppliers to reserve the right to purchase certain semiconductor chips over periods of 12-24 months, with volume commitments determined based on our anticipated production requirements. As of March 31, 2022, the Company's total commitments for these semiconductor chip agreements was \$38,919. Such agreements provide the Company with priority access to semiconductor chips as they become available, however, these agreements do not guarantee that our suppliers will meet the timing and quantities requested by Gentherm. All other purchase commitments as of March 31, 2022 were immaterial.

Earnings Per Share

Earnings Per Share
[Abstract]
Earnings per Share

3 Months Ended Mar. 31, 2022

Note 8 - Earnings Per Share

Basic earnings per share are computed by dividing net income by the weighted average number of shares of Common Stock outstanding during the period. The Company's diluted earnings per share give effect to all potential shares of Common Stock outstanding during a period that do not have an anti-dilutive impact to the calculation. In computing the diluted earnings per share, the treasury stock method is used in determining the number of shares assumed to be issued from the exercise of Common Stock equivalents.

The following table illustrates earnings per share and the weighted average shares outstanding used in calculating basic and diluted earnings per share:

	Three Months Ended March 31			
		2022		2021
Net income	\$	11,747	\$	32,909
Basic weighted average shares of Common Stock outstanding	3.	3,034,872		32,945,970
Dilutive effect of stock options, restricted stock awards and				
restricted stock units		341,892		443,621
Diluted weighted average shares of Common Stock outstanding	3.	3,376,764	33,389,591	
Basic earnings per share	\$	0.36	\$	1.00
Diluted earnings per share	\$	0.35	\$	0.99

There were no shares excluded from the Company's diluted earnings per share for the three months ended March 31, 2022 and 2021 on the basis that their inclusion would have an anti-dilutive impact on the calculation.

Financial Instruments

3 Months Ended Mar. 31, 2022

Derivative Instruments And Hedging Activities
Disclosure [Abstract]
Financial Instruments

Note 9 – Financial Instruments

Derivative Financial Instruments

The Company is exposed to various market risks including, but not limited to, changes in foreign currency exchange rates, changes in interest rates and price fluctuations of certain material commodities such as copper. Market risks for changes in interest rates relate primarily to its debt obligations under the Credit Agreement. Foreign currency exchange risks are attributable to sales to foreign customers and purchases from foreign suppliers not denominated in a location's functional currency, foreign plant operations, intercompany indebtedness, intercompany investments and include exposures to the Euro, Mexican Peso, Canadian Dollar, Hungarian Forint, North Macedonian Denar, Ukrainian Hryvnia, Japanese Yen, Chinese Renminbi, Korean Won and Vietnamese Dong.

The Company regularly enters into derivative contracts with the objective of managing its financial and operational exposure arising from these risks by offsetting gains and losses on the underlying exposures with gains and losses on the financial instruments used to hedge them. The maximum length of time over which the Company hedges its exposure to foreign currency exchange risks and price fluctuations in material commodities is fifteen months. The Company had foreign currency derivative contracts with a notional value of \$21,180 and \$13,974 and copper commodity swap contracts with a notional value of \$0 and \$309 outstanding as of March 31, 2022 and December 31, 2021, respectively. The principal currency hedged by the Company was the Mexican Peso.

The Company does not enter into derivative financial instruments for speculative or trading purposes. The Company's hedging relationships are formally documented at the inception of the hedge, and hedges must be highly effective in offsetting changes to future cash flows on hedged transactions both at the inception of a hedge and on an ongoing basis to be designated for hedge accounting treatment. For derivative contracts which can be classified as a cash flow hedge, the effective portion of the change in the fair value of the derivative is recorded to Accumulated other comprehensive loss in the consolidated condensed balance sheets. When the underlying hedge transaction is realized, the gain or loss included in Accumulated other comprehensive loss is recorded in earnings in the consolidated condensed statements of income on the same line as the gain or loss on the hedged item attributable to the hedged risk. The Company records the ineffective portion of foreign currency and copper commodity hedging instruments, if any, to cost of sales, in the consolidated condensed statements of income. Cash flows associated with derivatives are reported in net cash (used in) provided by operating activities in the Company's consolidated condensed statements of cash flows.

The Company uses an income approach to value derivative instruments, analyzing quoted market prices to calculate the forward values and then discounting such forward values to the present value using benchmark rates at commonly quoted intervals for the instrument's full term.

Information related to the recurring fair value measurement of derivative instruments in our consolidated condensed balance sheet as of March 31, 2022 is as follows:

			Asset Derivatives Liability Derivativ		rivatives			
	Hedge Designation	Fair Value Hierarchy	Balance Sheet Location	Fair Value	Balance Sheet Location	Fair Value	Net Ass (Liabilit	
	Cash		Other		Other			
Foreign currency	flow		current		current			
derivatives	hedge	Level 2	assets	\$ 904	liabilities	\$ —	\$ 9	904

	Cash		Other		Other		
	flow		current		current		
Commodity hedges	hedge	Level 2	assets	\$ —	liabilities	\$ — \$	

Information related to the recurring fair value measurement of derivative instruments in our consolidated condensed balance sheet as of December 31, 2021 is as follows:

			Asset Deri	rivatives Liability Der		rivat	ives		
	Hedge Designation	Fair Value Hierarchy	Balance Sheet Location		air alue	Balance Sheet Location	Fair Value		t Asset/ abilities)
	Cash		Other			Other			
Foreign currency	flow		current			current			
derivatives	hedge	Level 2	assets	\$	294	liabilities	\$	_	\$ 294
	Cash		Other			Other			
	flow		current			current			
Commodity hedges	hedge	Level 2	assets	\$	6	liabilities	\$	_	\$ 6

Information relating to the effect of derivative instruments on our consolidated condensed statements of income and the consolidated condensed statements of comprehensive income is as follows:

		Three Months Ended March 31,				
	Location		2022		2021	
Foreign currency derivatives	Cost of sales – income	\$	153	\$	478	
	Other comprehensive income					
	(loss)		610		(667)	
Total foreign currency						
derivatives		\$	763	\$	(189)	
				-		
Commodity derivatives	Cost of sales – income	\$	19	\$	_	
	Other comprehensive loss		(6)		_	
Total commodity derivatives		\$	13	\$		

The Company did not incur any hedge ineffectiveness during the three months ended March 31, 2022 and 2021.

Accounts Receivable Factoring

The Company is party to a receivable factoring agreement with HSBC Bank USA, National Association. Under the receivable factoring agreement, we can sell receivables for certain of our North America account debtors up to \$41,300, on a revolving basis, subject to outstanding balances and concentration limits. As of March 31, 2022, there were no outstanding receivables transferred under the receivable factoring agreement and our availability under the receivables factoring agreement was \$19,385.

Fair Value Measurements

3 Months Ended Mar. 31, 2022

Fair Value Disclosures
[Abstract]
Fair Value Measurements

Note 10 – Fair Value Measurements

Fair value is defined as the exchange price that would be received to sell an asset or paid to transfer a liability (an exit price) in the principal or most advantageous market for the asset or liability in an orderly transaction between market participants on the measurement date. Fair value measurements are based on one or more of the following three valuation techniques:

Market: This approach uses prices and other relevant information generated by market transactions involving identical or comparable assets or liabilities.

Income: This approach uses valuation techniques to convert future amounts to a single present value amount based on current market expectations.

Cost: This approach is based on the amount that would be required to replace the service capacity of an asset (replacement cost).

The Company uses the following fair value hierarchy to measure fair value into three broad levels, which are described below:

Level 1: Quoted prices (unadjusted) in active markets that are accessible at the measurement date for assets or liabilities. The fair value hierarchy gives the highest priority to Level 1 inputs.

Level 2: Inputs, other than quoted market prices included in Level 1, that are observable either directly or indirectly for the asset or liability.

Level 3: Unobservable inputs that are used when little or no market data is available. The fair value hierarchy gives the lowest priority to Level 3 inputs.

Items Measured at Fair Value on a Recurring Basis

Except for derivative instruments (see Note 9) and pension plan assets, the Company had no material financial assets and liabilities that were carried at fair value at March 31, 2022 and December 31, 2021. In determining fair value, the Company utilizes valuation techniques that maximize the use of observable inputs and minimize the use of unobservable inputs to the extent possible and also considers counterparty credit risk in its assessment of fair value.

Items Measured at Fair Value on a Nonrecurring Basis

The Company measures certain assets and liabilities at fair value on a non-recurring basis. As these nonrecurring fair value measurements are generally determined using unobservable inputs, these fair value measurements are classified within Level 3 of the fair value hierarchy. As of March 31, 2022 and December 31, 2021, there were no significant assets or liabilities measured at fair value on a non-recurring basis.

Items Not Carried at Fair Value

The Company uses an income valuation technique to measure the fair values of its debt instruments by converting amounts of future cash flows to a single present value amount using rates based on current market expectations (Level 2 inputs). As of March 31, 2022, and December 31, 2021, the carrying values of the indebtedness under the Company's Credit Agreement were not materially different than the estimated fair values because the interest rates on variable rate debt approximated rates currently available to the Company (see Note 6). Discount rates used to measure the fair value of Gentherm's DEG Vietnam Loan are based on quoted swap rates. As of March 31, 2022, the carrying value of the DEG Vietnam Loan was \$3,750 as compared to an estimated fair value of \$3,796. As of December 31, 2021, the carrying value of the DEG Vietnam Loan was \$3,750 as compared to an estimated fair value of \$3,778.

Equity

Stockholders Equity Note
[Abstract]
Equity

3 Months Ended Mar. 31, 2022

Note 11 - Equity

In December 2020, the Board of Directors of Gentherm Incorporated ("Board of Directors") authorized a stock repurchase program (the "2020 Stock Repurchase Program") to commence upon expiration of the prior stock repurchase program on December 15, 2020. Under the 2020 Stock Repurchase Program, the Company is authorized to repurchase up to \$150,000 of its issued and outstanding common stock over a three-year period, expiring December 15, 2023.

Repurchases may be made, from time to time, in amounts and at prices the Company deems appropriate, subject to market conditions, applicable legal requirements, debt covenants and other considerations. Any such repurchases may be executed using open market purchases, privately negotiated agreements or other transactions. Repurchases may be funded from cash on hand, available borrowings or proceeds from potential debt or other capital markets sources. The Company did not make any repurchases under the 2020 Stock Repurchase Program during the three months ended March 31, 2022, or March 31, 2021. The 2020 Stock Repurchase Program had \$130,000 repurchase authorization remaining as of March 31, 2022.

Reclassifications Out of Accumulated Other Comprehensive Loss

Equity [Abstract]

Reclassifications Out of Accumulated Other Comprehensive Loss

3 Months Ended Mar. 31, 2022

Note 12 - Reclassifications Out of Accumulated Other Comprehensive Loss

Reclassification adjustments and other activities impacting Accumulated other comprehensive loss during the three months ended March 31, 2022 and 2021 were as follows:

	Defined Benefit Pension Plans	Foreign Currency Translation Adjustments	Foreign Currency Hedge Derivatives	Commodity Hedge Derivatives	Total
Balance at December 31, 2021	\$(2,893)	\$ (34,188)	\$ 154	\$ 5	\$(36,922)
Other comprehensive (loss) income before reclassifications	_	(9,154)	763	13	(8,378)
Income tax effect of other comprehensive loss before reclassifications	_	(140)	(179)	(3)	(322)
Amounts reclassified from accumulated other comprehensive loss into net income	35	_	(153)a	(19)	a (137)
Income taxes reclassified into net income	(6)		33	4	31
Net current period other comprehensive income (loss)	29	(9,294)	464	(5)	(8,806)
Balance at March 31, 2022	\$(2,864)	\$ (43,482)	\$ 618	<u> </u>	\$(45,728)

(a) The amounts reclassified from accumulated other comprehensive loss were included in cost of sales.

]	Defined Benefit Pension Plans	T	Foreign Currency ranslation djustments	C	Toreign urrency Hedge rivatives	Total
Balance at December 31, 2020	\$	(3,451)	\$	(12,637)	\$	1,106	\$ (14,982)
Other comprehensive loss before reclassifications		_		(12,872)		(189)	(13,061)
Income tax effect of other comprehensive loss before reclassifications		_		(280)		41	(239)
Amounts reclassified from accumulated other comprehensive loss into net income		_		_		(478)	a (478)
Income taxes reclassified into net income		_		_		104	104
Net current period other comprehensive loss		_		(13,152)		(522)	(13,674)
Balance at March 31, 2021	\$	(3,451)	\$	(25,789)	\$	584	\$ (28,656)

⁽a) The amounts reclassified from accumulated other comprehensive loss were included in cost of sales.

The Company expects all of the existing gains and losses related to foreign currency and commodity hedge derivatives reported in Accumulated other comprehensive loss as of March 31, 2022 to be reclassified into earnings during the next twelve months. See Note 9 for additional information about derivative financial instruments and the effects from reclassification to net income.

Income Taxes

Income Tax Disclosure
[Abstract]
Income Taxes

3 Months Ended Mar. 31, 2022

Note 13 – Income Taxes

At the end of each interim period, the Company makes its best estimate of the annual expected effective income tax rate and applies that rate to its ordinary year-to-date earnings or loss. The income tax provision or benefit related to unusual or infrequent items, if applicable, that will be separately reported or reported net of their related tax effects are individually computed and recognized in the interim period in which those items occur. In addition, the effect of changes in enacted tax laws or rates, tax status, judgment on the realizability of a beginning-of-the-year deferred tax asset in future years or income tax contingencies is recognized in the interim period in which the change occurs.

The computation of the annual expected effective income tax rate at each interim period requires certain estimates and assumptions including, but not limited to, the expected pre-tax income (or loss) for the year, projections of the proportion of income (and/or loss) earned and taxed in respective jurisdictions, permanent and temporary differences, and the likelihood of the realizability of deferred tax assets generated in the current year. Jurisdictions with a projected loss for the year for which no tax benefit can be recognized due to a valuation allowance are excluded from the estimated annual effective tax rate. The impact of such an exclusion could result in a higher or lower effective tax rate during a particular quarter, based upon the composition and timing of actual earnings compared to annual projections. The estimates used to compute the provision or benefit for income taxes may change as new events occur, additional information is obtained or as our tax environment changes. To the extent that the expected annual effective income tax rate changes, the effect of the change on prior interim periods is included in the income tax provision in the period in which the change in estimate occurs.

A summary of the provision for income taxes and the corresponding effective tax rate for the three months ended March 31, 2022 and 2021, is shown below:

	Th	Three Months Ended March 31,					
		2022		2021			
Income tax expense	\$	4,295	\$	7,565			
Earnings before income tax	\$	16,042	\$	40,474			
Effective tax rate		26.8%	, —	18.7%			

Income tax expense was \$4,295 for the three months ended March 31, 2022, on earnings before income tax of \$16,042, representing an effective tax rate of 26.8%. The effective tax rate differed from the U.S. Federal statutory rate of 21.0% primarily due to the impact of income taxes on foreign earnings taxed at rates varying from the U.S. statutory rate, the unfavorable impact of the global intangible low-tax income ("GILTI"), and the quarterly accrual for uncertain tax positions, partially offset by the impact of certain favorable tax effects on equity vesting.

Income tax expense was \$7,565 for the three months ended March 31, 2021, on earnings before income tax of \$40,474, representing an effective tax rate of 18.7%. The effective tax rate differed from the U.S. Federal statutory rate of 21.0% primarily due to certain intercompany transactions, partially offset by unfavorable impact of the global intangible low-tax income ("GILTI").

Segment Reporting

Segment Reporting [Abstract]
Segment Reporting

3 Months Ended Mar. 31, 2022

Note 14 - Segment Reporting

Segment information is used by management for making strategic operating decisions for the Company. Management evaluates the performance of the Company's segments based primarily on operating income or loss.

The Company's reportable segments are as follows:

- *Automotive* this segment represents the design, development, manufacturing and sales of automotive climate comfort systems, automotive cable systems, battery performance solutions, and automotive electronic and software systems.
- *Medical* this segment represents the results from our patient temperature management business within the medical industry.

The *Corporate* category includes unallocated costs related to our corporate headquarter activities, including selling, general and administrative costs and acquisition transaction costs, which do not meet the requirements for being classified as an operating segment.

The tables below present segment information about the reported product revenues, depreciation and amortization and operating income (loss) of the Company for three months ended March 31, 2022 and 2021.

Three Months Ended March 31,	Automotive	Medical Corporate		Total
2022:				
Product revenues	\$ 257,864	\$ 9,793	\$ —	\$ 267,657
Depreciation and amortization	8,667	604	306	\$ 9,577
Operating income (loss)	31,275	(851)	(16,234)	\$ 14,190
2021:				
Product revenues	\$ 279,370	\$ 9,165	\$ —	\$ 288,535
Depreciation and amortization	9,053	584	217	\$ 9,854
Operating income (loss)	53,116	(531)	(11,836)	\$ 40,749

Automotive and Medical segment product revenues by product category for the three months ended March 31, 2022 and 2021 were as follows:

	T	Three Months Ended March 31,			
		2022		2021	
Climate Control Seat	\$	102,734	\$	109,173	
Seat Heaters		68,896		76,721	
Steering Wheel Heaters		28,736		28,864	
Automotive Cables		22,045		24,281	
Battery Performance Solutions		17,613		17,760	
Electronics		10,828		15,105	
Other Automotive		7,012		7,466	
Subtotal Automotive segment		257,864		279,370	
Medical segment		9,793		9,165	
Total Company	\$	267,657	\$	288,535	

Total product revenues information by geographic area for the three months ended March 31, 2022 and 2021 is as follows (based on shipment destination):

Three Months Ended March 31,

	 2022	2021
United States	\$ 104,122	\$ 115,287
China	38,353	33,515
South Korea	21,175	24,640
Germany	19,786	19,180
Romania	12,755	15,085
Japan	11,816	17,920
Other	59,650	62,908
Total Non-U.S.	163,535	173,248
Total Company	\$ 267,657	\$ 288,535

Subsequent Events

Subsequent Events
[Abstract]
Subsequent Events

3 Months Ended Mar. 31, 2022

Note 15 – Subsequent Events

On May 4, 2022, the Company entered into a definitive purchase agreement to acquire the automotive business of Alfmeier Präzision SE ("Alfmeier"), the global leader in automotive lumbar and massage comfort solutions and a leading provider of advanced valve systems technology, integrated electronics and software.

Under the terms of the agreement, the Company will acquire all shares in Alfmeier for €177,500, net of cash and debt, subject to customary adjustments related to Alfmeier's net working capital as of the closing and may be increased by up to €4,000 at or post-closing upon resolution of certain Alfmeier tax matters. The transaction will be funded through a combination of the Company's existing cash balances and revolving credit facility. The transaction is expected to close during the third quarter of 2022 subject to the receipt of regulatory approvals and the satisfaction or waiver of other customary closing conditions.

Also on May 4, 2022, the Company entered into a put and call option agreement for the real property in Germany on which Alfmeier's headquarters is located. Pursuant to this agreement, the seller will have a put to sell the headquarters and the Company will have an option to purchase the headquarters post-closing of the acquisition of Alfmeier.

New Accounting Pronouncements (Policies)

Accounting Policies
[Abstract]

Basis of Presentation and Significant Accounting Policies

3 Months Ended Mar. 31, 2022

Basis of Presentation and Significant Accounting Policies

The unaudited consolidated condensed financial statements included herein have been prepared in accordance with accounting principles generally accepted in the United States of America ("U.S. GAAP") for interim financial information and pursuant to the rules and regulations of the U.S. Securities and Exchange Commission. Certain information and footnote disclosures normally included in financial statements prepared in accordance with U.S. GAAP have been condensed or omitted pursuant to those rules and regulations. The information furnished in the consolidated condensed financial statements include all adjustments (consisting of only normal, recurring adjustments), considered necessary to present fairly the results of operations, financial position and cash flows of the Company. These financial statements should be read in conjunction with the Company's Annual Report on Form 10-K for the year ended December 31, 2021. The operating results for interim periods are not necessarily indicative of results that may be expected for any other interim period or for the full year.

In preparing these financial statements, management was required to make estimates and assumptions that affect the reported amounts of assets, liabilities, revenues and expenses. These estimates and assumptions are based on our historical experience, the terms of existing contracts, our evaluation of trends in the industry, information provided by our customers and suppliers and information available from other outside sources, as appropriate. These estimates and assumptions are subject to an inherent degree of uncertainty. We are not presently aware of any events or circumstances that would require us to update such estimates and assumptions or revise the carrying value of our assets or liabilities. Our estimates may change, however, as new events occur and additional information is obtained. As a result, actual results may differ significantly from our estimates, and any such differences may be material to our financial statements.

Principles of Consolidation

Principles of Consolidation

The consolidated condensed financial statements include the accounts of the Company, its wholly owned subsidiaries and those entities in which it has a controlling financial interest. The Company evaluates its relationship with other entities for consolidation and to identify whether such entities are variable interest entities ("VIE") and to assess whether the Company is the primary beneficiary of such entities. Investments in affiliates in which Gentherm does not have control but does have the ability to exercise significant influence over operating and financial policies are accounted for under the equity method. When Gentherm does not have the ability to exercise significant influence (generally when ownership interest is less than 20%), investments in affiliates are measured at cost, less impairments, adjusted for observable price changes in orderly transactions for identical or similar investments of the same issuer.

Variable Interest Entities

Variable Interest Entities

The Company maintains an ownership interest in a VIE, Carrar Ltd. ("Carrar"). Carrar is a technology developer of advanced thermal management systems for the electric mobility market. The Company determined that Carrar is a VIE; however, the Company does not have a controlling financial interest or have the power to direct the activities that most significantly affect the economic performance of the investment. Therefore, the Company has concluded that it is not the primary beneficiary. Gentherm's investment in Carrar is measured at cost, less impairments, adjusted for observable price changes in orderly transactions for identical or similar investments of the same issuer. The Carrar investment was \$5,200 as of March 31, 2022 and December 31, 2021, and is recorded in Other non-current assets in the consolidated condensed balance sheet.

Revenue Recognition

Revenue Recognition

The Company has no material contract assets or contract liabilities as of March 31, 2022.

The Company recognizes an asset for the incremental costs of obtaining a contract with a customer if the benefits of those costs are expected to be realized for a period greater than one year. Total capitalized costs to obtain a contract were \$1,884 and \$1,946 as of March 31, 2022 and December 31, 2021, respectively. These amounts are recorded in Other non-current assets and are being amortized into Product revenues over the expected production life of the applicable program.

Segment Reporting

Segment Reporting

The Company has two reportable segments: Automotive and Medical.

The Automotive reporting segment is comprised of the results from our global automotive businesses, including the design, development, manufacturing and sales of automotive climate comfort systems, automotive cable systems, battery performance solutions, and automotive electronic and software systems.

The Medical reporting segment is comprised of the results from the patient temperature management business in the medical industry. Patient temperature management includes temperature management systems across multiple product categories addressing the needs of hyper-hypothermia therapy in intensive care, normothermia in surgical procedures and additional warming/cooling therapies utilized in acute care, ambulatory, clinics and home health.

Reference Rate Reform

Reference Rate Reform

In March 2020, the FASB issued ASU 2020-04, "Reference Rate Reform (Topic 848): Facilitation of Effects of Reference Rate Reform on Financial Reporting". ASU 2020-04 provides practical expedients and exceptions for applying GAAP to contracts, hedging relationships, and other transactions affected by reference rate reform if certain criteria are met. The expedients and exceptions provided by the amendments in this update apply only to contracts, hedging relationships, and other transactions that reference the London Interbank Offered Rate ("LIBOR") or another reference rate expected to be discontinued as a result of reference rate reform. These amendments are not applicable to contract modifications made and hedging relationships entered into or evaluated after December 31, 2022. In January 2021, the FASB subsequently issued ASU 2021-01, "Reference Rate Reform (Topic 848): Scope" to clarify that certain optional expedients and exceptions in Topic 848 for contract modifications and hedge accounting apply to derivatives that are affected by the discounting transition. ASU 2020-04 and ASU 2021-01 are effective as of March 12, 2020 through December 31, 2022 and may be applied retrospectively to contract modifications and hedging relationships from the beginning of an interim period that includes or is subsequent to March 12, 2020, or on a prospective basis to new modifications from any date within an interim period that includes or is subsequent to the date of the issuance of a final update, up to the date that financial statements are available to be issued. The adoption of this standard is not expected to have a material impact on the Company's financial statements.

Restructuring (Tables)

3 Months Ended Mar. 31, 2022

Restructuring And Related Activities [Abstract]

Summary of Restructuring Expense by Reportable Segment

The following table summarizes restructuring expense for the three months ended March 31, 2022 and 2021 by reporting segment:

	Thre	e Months	ed March
	2	2022	2021
Automotive	\$	181	\$ 791
Medical		_	_
Corporate		_	_
Total	\$	181	\$ 791

Summary of Restructuring Liability

The following table summarizes restructuring liability for the three months ended March 31, 2022:

	Employee Separation Costs		F	Other Related Costs		Total
Balance at December 31, 2021	\$	1,494	\$	_	\$	1,494
Additions, charged to restructuring expenses		_		131		131
Cash payments		(544)		(131)		(675)
Non-cash utilization		_		_		
Change in estimate		50		_		50
Currency translation		(18)				(18)
Balance at March 31, 2022	\$	982	\$		\$	982

Details of Certain Balance Sheet Components (Tables)

Balance Sheet Components [Abstract]
Summary of Certain Balance Sheet
Components

3 Months Ended Mar. 31, 2022

	March 31, 2022		December 31, 2021	
Other current assets:				
Notes receivable	\$	15,958	\$	13,033
Income tax and other tax receivable		12,501		10,681
Billable tooling		4,993		3,778
Prepaid expenses		4,751		3,407
Other		2,831		1,876
Total other current assets	\$	41,034	\$	32,775
Other current liabilities:				
Liabilities from discounts and rebates	\$	26,632	\$	27,343
Accrued employee liabilities		21,244		28,818
Income tax and other taxes payable		19,529		17,068
Restructuring		982		1,494
Accrued warranty		1,772		1,916
Other		8,885		5,554
Total other current liabilities	\$	79,044	\$	82,193

Goodwill and Other Intangibles (Tables)

Goodwill And Intangible Assets Disclosure [Abstract]

Summary of Changes in the Carrying Amount of Goodwill, By Reportable Segment

3 Months Ended Mar. 31, 2022

Changes in the carrying amount of goodwill, by reportable segment, for the three months ended March 31, 2022 was as follows:

	Automotive		 Medical		Total	
'Balance as of December 31, 2021	\$	37,329	\$ 28,704	\$	66,033	
Exchange rate impact		(762)	(292)		(1,054)	
Balance as of March 31, 2022	\$	36,567	\$ 28,412	\$	64,979	

<u>Summary of Other Intangible Assets and</u> Accumulated Amortization Balances

Other intangible assets and accumulated amortization balances as of March 31, 2022 and December 31, 2021 were as follows:

	I	March 31, 2022				December 31, 2021			
	Gross Carrying Value		ccumulated mortization	Net Carrying Value	(Gross Carrying Value		ccumulated mortization	Net Carrying Value
Definite-lived:									
Customer relationships	\$ 87,854	\$	(63,445)	\$24,409	\$	90,448	\$	(64,105)	\$26,343
Technology	28,706		(24,288)	4,418		29,464		(24,487)	4,977
Product development costs	19,661		(19,183)	478		20,329		(19,772)	557
Software development	1,007		_	1,007		1,007		_	1,007
Indefinite- lived:									
Tradenames	4,670		_	4,670		4,670		_	4,670
Total	\$141,898	\$	(106,916)	\$34,982	\$	145,918	\$	(108,364)	\$37,554

Debt (Tables)

3 Months Ended Mar. 31, 2022

Debt Disclosure [Abstract]

Summary of Company's Debt

The following table summarizes the Company's debt as of March 31, 2022 and December 31, 2021:

	March 3	1, 2022	December 31, 2021		
	Interest Rate	Principal Balance	Interest Rate	Principal Balance	
Amended Credit Agreement:					
U.S. Revolving Note (U.S. Dollar denominations)	1.71%	\$ 35,000	1.35%	\$ 35,000	
U.S. Revolving Note (Euro denominations)	_	_	_	_	
DEG Vietnam Loan	5.21%	3,750	5.21%	3,750	
Total debt		38,750		38,750	
Current maturities		(2,500)		(2,500)	
Long-term debt, less current maturities		\$ 36,250		\$ 36,250	

Principal Maturities of Debt

The scheduled principal maturities of our debt as of March 31, 2022 were as follows:

	DEG Vietnam Note	R	U.S. evolving Note	Total
2022	\$ 2,500	\$		\$ 2,500
2023	1,250		_	1,250
2024	 		35,000	 35,000
Total	\$ 3,750	\$	35,000	\$ 38,750

Commitments and Contingencies (Tables)

Commitments And Contingencies

Disclosure [Abstract]

Reconciliation of Changes in Accrued Warranty Costs

3 Months Ended Mar. 31, 2022

The following is a reconciliation of the changes in accrued warranty costs:

	Three Months Ended Marc 31,			
		2022		2021
Balance at the beginning of the period	\$	1,916	\$	2,391
Warranty claims paid		(334)		(352)
Warranty expense for products shipped during				
the current period		354		519
Adjustments to warranty estimates from prior				
periods		(152)		271
Adjustments due to currency translation		(12)		(31)
Balance at the end of the period	\$	1,772	\$	2,798

Earnings Per Share (Tables)

Earnings Per Share [Abstract]

Earnings per Share and Weighted Average Shares
Outstanding Used in Calculating Basic and Diluted
Earnings per Share

3 Months Ended Mar. 31, 2022

The following table illustrates earnings per share and the weighted average shares outstanding used in calculating basic and diluted earnings per share:

	Three Months Ended March 31,					
		2022	2021			
Net income	\$	11,747	\$ 32,90			
			_			
Basic weighted average shares of						
Common Stock outstanding	33	,034,872	32,945,970			
Dilutive effect of stock options,						
restricted stock awards and restricted						
stock units		341,892		443,621		
Diluted weighted average shares of						
Common Stock outstanding	33	,376,764	33,389,591			
Basic earnings per share	\$	0.36	\$	1.00		
Diluted earnings per share	\$	0.35	\$	0.99		

Financial Instruments (Tables)

Derivative Instruments And Hedging Activities Disclosure [Abstract]

Information Related to Recurring Fair Value Consolidated Condensed Balance Sheet

3 Months Ended Mar. 31, 2022

Information related to the recurring fair value measurement of derivative Measurement of Derivative Instruments in Our instruments in our consolidated condensed balance sheet as of March 31, 2022 is as follows:

			Asset Derivatives		Liabili Derivat		
	Hedge Designation	Fair Value Hierarchy	Balance Sheet Location	Fair Value	Balance Sheet Location	Fair Value	Net Asset/ (Liabilities)
Foreign	Cash		Other		Other		
currency	flow		current		current		
derivatives	hedge	Level 2	assets	\$ 904	liabilities	\$ —	\$ 904
	Cash		Other		Other		
Commodity	flow		current		current		
hedges	hedge	Level 2	assets	\$ —	liabilities	\$ —	\$ —

Information related to the recurring fair value measurement of derivative instruments in our consolidated condensed balance sheet as of December 31, 2021 is as follows:

			Asset Derivatives		Liability Derivatives		
	Hedge Designation	Fair Value Hierarchy	Balance Sheet Location	Fair Value	Balance Sheet Location	Fair Value	 t Asset/
Foreign currency	Cash flow	- 14	Other current		Other		•••
derivatives	hedge Cash	Level 2	Other	\$ 294	liabilities Other	\$ —	\$ 294
Commodity hedges	flow hedge	Level 2	current assets	\$ 6	current liabilities	\$ —	\$ 6

Information Related to Effect of Derivative Instruments on Our Consolidated Condensed Statements of Income and Consolidated Condensed Statements of Comprehensive Income (Loss)

Information relating to the effect of derivative instruments on our consolidated condensed statements of income and the consolidated condensed statements of comprehensive income is as follows:

		Th	ree Months		led March
	Location		2022	2021	
Foreign currency					
derivatives	Cost of sales – income	\$	153	\$	478
	Other comprehensive income (loss)		610		(667)
Total foreign currency					
derivatives		\$	763	\$	(189)
Commodity					
derivatives	Cost of sales – income	\$	19	\$	_
	Other comprehensive loss		(6)		_
Total commodity					
derivatives		\$	13	\$	

Reclassifications Out of Accumulated Other Comprehensive Loss (Tables)

3 Months Ended

Mar. 31, 2022

Equity [Abstract]

Schedule of Reclassification Adjustments and Comprehensive Loss

Reclassification adjustments and other activities impacting Accumulated Other Activities Impacting Accumulated Other other comprehensive loss during the three months ended March 31, 2022 and 2021 were as follows:

	Defined Benefit Pension Plans	Foreign Currency Translation Adjustments	Foreign Currency Hedge Derivatives	Commodity Hedge Derivatives	Total
Balance at					
December 31,	# (2 002)	Φ (2.4.1 00)	Φ 154	Φ	# (2 < 022)
2021	\$(2,893)	\$ (34,188)	\$ 154	\$ 5	\$(36,922)
Other comprehensive (loss) income before					
reclassifications	_	(9,154)	763	13	(8,378)
Income tax effect of other comprehensive loss before reclassifications		(140)	(179)	(3)	(322)
Amounts reclassified from accumulated other comprehensive loss into net	_	(140)	(179)	(3)	(322)
income	35	_	(153)	a (19)	a (137)
Income taxes reclassified into net income	(6)		33	4	31
Net current period other comprehensive income (loss)	29	(9,294)	464	(5)	(8,806)
Balance at March 31, 2022	\$(2,864)			<u> </u>	\$(45,728)

(a) The amounts reclassified from accumulated other comprehensive loss were included in cost of sales.

	Defined Benefit Pension Plans	Foreign Currency Translation Adjustments	Foreign Currency Hedge Derivatives	Total
Balance at December 31, 2020	\$(3,451)	\$ (12,637)	\$ 1,106	\$(14,982)
Other comprehensive loss before reclassifications	_	(12,872)	(189)	(13,061)
Income tax effect of other comprehensive loss before				
reclassifications	_	(280)	41	(239)

Amounts reclassified from accumulated other comprehensive loss into net (478)a income (478)Income taxes reclassified into net income 104 104 Net current period other comprehensive loss (13,152)(522) (13,674) Balance at March 31, 2021 \$(3,451) \$ (25,789) \$ 584 \$(28,656)

⁽a) The amounts reclassified from accumulated other comprehensive loss were included in cost of sales.

Income Taxes (Tables)

3 Months Ended Mar. 31, 2022

Income Tax Disclosure [Abstract]

Summary of Provision for Income Taxes and Corresponding Effective Tax Rate

A summary of the provision for income taxes and the corresponding effective tax rate for the three months ended March 31, 2022 and 2021, is shown below:

	Three Months Ended March 31,			
		2022		2021
Income tax expense	\$	4,295	\$	7,565
Earnings before income tax	\$	16,042	\$	40,474
Effective tax rate		26.8%	, 	18.7%

Segment Reporting (Tables)

Segment Reporting [Abstract]

Segment Information about Reported

Product Revenues, Depreciation and

Amortization and Operating Income (Loss)

3 Months Ended Mar. 31, 2022

The tables below present segment information about the reported product revenues, depreciation and amortization and operating income (loss) of the Company for three months ended March 31, 2022 and 2021.

Three Months Ended March 31,	Αι	utomotive	M	Iedical	C	orporate		Total
2022:				_				
Product revenues	\$	257,864	\$	9,793	\$	_	\$2	267,657
Depreciation and amortization		8,667		604		306	\$	9,577
Operating income (loss)		31,275		(851)		(16,234)	\$	14,190
2021:								
Product revenues	\$	279,370	\$	9,165	\$	_	\$2	288,535
Depreciation and amortization		9,053		584		217	\$	9,854
Operating income (loss)		53,116		(531)		(11,836)	\$	40,749

Automotive and Medical segment product revenues by product category for the three months ended March 31, 2022 and 2021 were as follows:

Segment Information About Reported
Segment Product Revenues by Product
Category

	Three Months Ended March 31,			
		2022		2021
Climate Control Seat	\$	102,734	\$	109,173
Seat Heaters		68,896		76,721
Steering Wheel Heaters		28,736		28,864
Automotive Cables		22,045		24,281
Battery Performance Solutions		17,613		17,760
Electronics		10,828		15,105
Other Automotive		7,012		7,466
Subtotal Automotive segment		257,864		279,370
Medical segment		9,793		9,165
Total Company	\$	267,657	\$	288,535

Product Revenues Information by Geographic Area

Total product revenues information by geographic area for the three months ended March 31, 2022 and 2021 is as follows (based on shipment destination):

	Ti	Three Months Ended March 31,			
		2022 20			
United States	\$	104,122	\$	115,287	
China		38,353		33,515	
South Korea		21,175		24,640	
Germany		19,786		19,180	
Romania		12,755		15,085	
Japan		11,816		17,920	
Other	<u> </u>	59,650		62,908	
Total Non-U.S.		163,535		173,248	
Total Company	\$	267,657	\$	288,535	

Overview - Additional Information (Detail) \$ in Thousands	3 Months Ended Mar. 31, 2022 USD (\$) Segment	Dec. 31, 2021 USD (\$)
Organization Consolidation And Presentation Of Financial Statements [Line		
<u>Items</u>]		
<u>Investment amount</u>	\$ 5,200	\$ 5,200
Capitalized costs to obtain contract	\$ 1,884	\$ 1,946
Number of reportable segments Segment	2	
<u>Maximum</u>		
Organization Consolidation And Presentation Of Financial Statements [Line		
<u>Items</u>]		
Ownership interest	20.00%	
<u>Minimum</u>		
Organization Consolidation And Presentation Of Financial Statements [Line		
<u>Items</u>]		
Expected period of costs to be realized to recognize assets	1 year	

Restructuring - Additional Information (Detail) - USD (\$) Restructuring Cost And Reserve [Line	3 Months Ended Mar. 31, 2022	Mar. 31, 2021
Items]		
Restructuring expenses	\$ 181,000	\$ 791,000
Restructuring and related activities, description	Actions under the Plan are expected to be substantially completed by the end of 2022	
<u>Maximum</u>		
Restructuring Cost And Reserve [Line Items]		
Future restructuring expenses expected Manufacturing Footprint Rationalization	\$ 1,000,000	
Restructuring Cost And Reserve [Line		
<u>Items</u>]		
Restructuring expenses	10,256,000	
Other restructuring costs	101,000	259,000
Restructuring expenses remains accrued	538,000	
Other Restructuring Activities		
Restructuring Cost And Reserve [Line		
<u>Items</u>]		
Other restructuring costs	30,000	
Employee Separation Costs Manufacturing		
Footprint Rationalization		
Restructuring Cost And Reserve [Line		
<u>Items</u>]	4.70.000	• • • • • • •
Restructuring expenses	\$ 50,000	206,000
Employee Separation Costs Other		
Restructuring Activities		
Restructuring Cost And Reserve [Line		
Items]		¢
Restructuring expenses		\$ 326,000

Restructuring - Summary of
Restructuring Expense by
Reportable Segment
(Details) - USD (\$)
\$ in Thousands

3 Months Ended

Mar. 31, 2022 Mar. 31, 2021

Restructuring Cost And Reserve [Line Items]

Restructuring expenses \$ 181 \$ 791

Operating Segments | Automotive Segments

Restructuring Cost And Reserve [Line Items]

Restructuring expenses \$ 181 \$ 791

Restructuring - Summary of Restructuring Liability (Detail) \$\text{ in Thousands}\$

Mar. 31, 2022 USD (\$)

3 Months Ended

Restructuring Cost And Reserve [Line Items]

Balance, beginning of period	\$ 1,494
Additions, charged to restructuring expenses	131
<u>Cash payments</u>	(675)
Change in estimate	50
<u>Currency translation</u>	(18)
Balance, end of period	982

Employee Separation Costs

Restructuring Cost And Reserve [Line Items]

Balance, beginning of period	1,494
<u>Cash payments</u>	(544)
Change in estimate	50
Currency translation	(18)
Balance, end of period	982

Other Related Costs

Restructuring Cost And Reserve [Line Items]

Additions, charged to restructuring expenses	131
<u>Cash payments</u>	\$ (131)

Details of Certain Balance Sheet Components (Detail) -USD (\$)

\$ in Thousands

Mar. 31, 2022 Dec. 31, 2021 Mar. 31, 2021 Dec. 31, 2020

Current Assets:

Notes receivable	\$ 15,958	\$ 13,033		
Income tax and other tax receivable	12,501	10,681		
Billable tooling	4,993	3,778		
Prepaid expenses	4,751	3,407		
<u>Other</u>	2,831	1,876		
Total other current assets	41,034	32,775		
Other current liabilities:				
Liabilities from discounts and rebate	<u>s</u> 26,632	27,343		
Accrued employee liabilities	21,244	28,818		
Income tax and other taxes payable	19,529	17,068		
Restructuring	982	1,494		
Accrued warranty	1,772	1,916	\$ 2,798	\$ 2,391
<u>Other</u>	8,885	5,554		
Total other current liabilities	\$ 79,044	\$ 82,193		

Goodwill and Other Intangibles - Summary of	3 Months Ended
Changes in the Carrying Amount of Goodwill, By Reportable Segment (Detail) \$ in Thousands	Mar. 31, 2022 USD (\$)
Goodwill [Line Items]	
Balance, beginning of period	\$ 66,033
Balance, end of period	64,979
Operating Segments	
Goodwill [Line Items]	
Balance, beginning of period	66,033
Exchange rate impact	(1,054)
Balance, end of period	64,979
Automotive Segments Operating Segment	<u>ts</u>
Goodwill [Line Items]	
Balance, beginning of period	37,329
Exchange rate impact	(762)
Balance, end of period	36,567
Medical Segments Operating Segments	
Goodwill [Line Items]	
Balance, beginning of period	28,704
Exchange rate impact	(292)
Balance, end of period	\$ 28,412

Goodwill and Other

Intangibles - Summary of

Other Intangible Assets and **Accumulated Amortization**

Balances (Detail) - USD (\$)

\$ in Thousands

Intangible Assets [Line Items]

Net Carrying Value

	4	
Gross Carrying Value	\$ 141,898	\$ 145,918
Accumulated Amortization	(106,916)	(108,364)
Net Carrying Value	34,982	37,554
Net Carrying Value	34,982	37,554
Customer Relationships		
Intangible Assets [Line Items	1	
Gross Carrying Value	87,854	90,448
Accumulated Amortization	(63,445)	(64,105)
Net Carrying Value	24,409	26,343
Technology		
Intangible Assets [Line Items	1	
Gross Carrying Value	28,706	29,464
Accumulated Amortization	(24,288)	(24,487)
Net Carrying Value	4,418	4,977
Product development costs		
Intangible Assets [Line Items	1	
Gross Carrying Value	19,661	20,329
Accumulated Amortization	(19,183)	(19,772)
Net Carrying Value	478	557
Software Development		
Intangible Assets [Line Items]		
Gross Carrying Value	1,007	1,007
Net Carrying Value	1,007	1,007
<u>Tradenames</u>		
Intangible Assets [Line Items]		

\$ 4,670

Mar. 31, 2022 Dec. 31, 2021

\$4,670

Summary of Company's		
Debt (Detail) - USD (\$)	Mar. 31, 20	22 Dec. 31, 2021
\$ in Thousands		
Debt Instrument [Line Items]		
<u>Total debt</u>	\$ 38,750	\$ 38,750
<u>Current maturities</u>	(2,500)	(2,500)
Long-term debt, less current maturities	\$ 36,250	\$ 36,250
U.S. Revolving Note (U.S. Dollar Denomina	tions)	
Debt Instrument [Line Items]		
Interest Rate	1.71%	1.35%
<u>Total debt</u>	\$ 35,000	\$ 35,000
DEG Vietnam Loan		
Debt Instrument [Line Items]		
Interest Rate	5.21%	5.21%
<u>Total debt</u>	\$ 3,750	\$ 3,750

Debt - Additional Information (Detail) - USD		3 Months Ended	
(\$)	Jun. 27, 2019	Mar. 31, 2022	Dec. 31, 2021
Debt Instrument [Line Items]			
Maximum percentage of stock of non US subsidiaries pledge to secure		66.00%	
<u>obligation</u>		00.00%	
Federal Funds Rate Minimum			
Debt Instrument [Line Items]			
<u>Interest rate</u>		0.50%	
Federal Funds Rate Maximum			
Debt Instrument [Line Items]			
<u>Interest rate</u>		0.33%	
London Interbank Offered Rate			
Debt Instrument [Line Items]			
<u>Interest rate</u>		0.45%	
United State Bank Of America Credit Facility			
Debt Instrument [Line Items]			
<u>Interest rate</u>		3.50%	
Euro Currency Rate Loans			
Debt Instrument [Line Items]			
<u>Interest rate</u>		1.00%	
Euro Currency Rate Loans Credit Agreement Minimum			
Debt Instrument [Line Items]			
<u>Interest rate</u>		1.25%	
Euro Currency Rate Loans Credit Agreement Maximum			
Debt Instrument [Line Items]			
<u>Interest rate</u>		2.25%	
Base Rate Loans Credit Agreement Minimum			
Debt Instrument [Line Items]			
<u>Interest rate</u>		0.25%	
Base Rate Loans Credit Agreement Maximum			
Debt Instrument [Line Items]			
<u>Interest rate</u>		1.25%	
DEG Vietnam Loan			
Debt Instrument [Line Items]			
Semi-annual principal payments earliest date		2017-11	
Semi-annual principal payments latest date		2023-05	
Revolving Credit Facility Revolving Note (U.S. Dollar)			
Debt Instrument [Line Items]			
Maximum borrowing capacity	\$ 475,000,000)	
Line of credit, outstanding amount	¬12,000,000	\$ 35,000,000	
Enic of credit, outstanding amount		ψ 33,000,000	

Debt maturity date Jun. 27, 2024 Remaining borrowing capacity 424,521,000 Revolving Credit Facility | Revolving Note (U.S. Dollar) | Letters of Credit **Debt Instrument [Line Items]** 15,000,000 Maximum borrowing capacity Line of credit, outstanding amount 0 \$0 Revolving Credit Facility | Revolving Note (U.S. Dollar) | Swing Line Loans **Debt Instrument [Line Items]** \$ 40,000,000 Maximum borrowing capacity

Principal Maturities of Debt (Detail) - USD (\$) \$ in Thousands	Mar. 31, 202	2 Dec. 31, 2021
Debt Instrument [Line Items]		
Debt maturing in 2022	\$ 2,500	
Debt maturing in 2023	1,250	
Debt maturing in 2024	35,000	
<u>Total debt</u>	38,750	\$ 38,750
DEG Vietnam Loan		
Debt Instrument [Line Items]		
Debt maturing in 2022	2,500	
Debt maturing in 2023	1,250	
<u>Total debt</u>	3,750	\$ 3,750
US and Euro Denominated Revolving Not	<u>e</u>	
Debt Instrument [Line Items]		
Debt maturing in 2024	35,000	

Total debt

\$ 35,000

Commitments and Contingencies - Reconciliation of Changes in	3 Months Ended	
Accrued Warranty Costs (Detail) - USD (\$)	Mar. 31, 202	2 Mar. 31, 2021
\$ in Thousands		
Loss Contingency [Abstract]		
Balance at the beginning of the period	\$ 1,916	\$ 2,391
Warranty claims paid	(334)	(352)
Warranty expense for products shipped during the current period	1354	519
Adjustments to warranty estimates from prior periods	(152)	271
Adjustments due to currency translation	(12)	(31)
Balance at the end of the period	\$ 1,772	\$ 2,798

Commitments and 3 Months Ended

Contingencies - Additional

Information (Detail) - Mar. 31, 2022 Semiconductor Chips \$ in Thousands USD (\$)

Loss Contingencies [Line Items]

Commitments \$ 38,919

Minimum

Loss Contingencies [Line Items]

Commitment period 12 months

Maximum

Loss Contingencies [Line Items]

Commitment period 24 months

Earnings Per Share (Detail) -3 Months Ended **USD (\$)** \$ / shares in Units, \$ in Mar. 31, 2022 Mar. 31, 2021 **Thousands Earnings Per Share [Abstract]** Net income \$ 11,747 \$ 32,909 Basic weighted average shares of Common Stock outstanding 33,034,872 32,945,970 Dilutive effect of stock options, restricted stock awards and restricted stock units 341,892 443,621 Diluted weighted average shares of Common Stock outstanding 33,376,764 33,389,591 Basic earnings per share \$ 0.36 \$ 1.00 \$ 0.35 Diluted earnings per share \$ 0.99

Earnings Per Share -Additional Information (Detail) - shares

3 Months Ended
Mar. 31, 2022 Mar. 31, 2021

Earnings Per Share [Abstract]

Shares excluded from the Company's diluted earnings 0

0

Financial Instruments -	3 Months Ended		
Additional Information (Detail) - USD (\$)	Mar. 31, 2022	Mar. 31, 2021	Dec. 31, 2021
Derivative [Line Items]			
Maximum length of time to hedges exposure to foreign currency exchange risk	15 months		
Sale of receivable	\$ 41,300,000		
Receivables transferred	0		
Availability under receivables factoring agreement	19,385,000		
Foreign Currency Derivatives			
Derivative [Line Items]			
Notional Value	21,180,000		\$ 13,974,000
Copper Commodity Swap Contracts			
Derivative [Line Items]			
Notional Value	0		\$ 309,000
Interest Rate Swap			
Derivative [Line Items]			
Hedge Ineffectiveness Incurred	\$ 0	\$ 0	

Information Related to Recurring Fair Value Measurement of Derivative Instruments in Our Consolidated Condensed

Balance Sheet (Detail) - Fair

Value, Inputs, Level 2 -Designated as Hedging Instrument - USD (\$)

\$ in Thousands

Foreign	Currency	Derivatives
TOTOTSII	Controlle	Delliatives

Derivatives Fair Value [Line Items]

Net Asset/ (Liabilities) \$ 904 \$ 294

Mar. 31, 2022 Dec. 31, 2021

Foreign Currency Derivatives | Other Current Assets

Derivatives Fair Value [Line Items]

Asset Derivatives, Fair Value \$ 904 294

Commodity Hedges

Derivatives Fair Value [Line Items]

Net Asset/ (Liabilities) 6

Commodity Hedges | Other Current Assets

Derivatives Fair Value [Line Items]

Asset Derivatives, Fair Value \$ 6

Information Related to Effect of Derivative Instruments on Our Consolidated Condensed Statements of Income and Consolidated Condensed Statements of Comprehensive Income (Loss) (Detail) - USD (\$) \$ in Thousands		ths Ended 2 Mar. 31, 2021
Foreign Currency Derivatives		
Derivative Instruments Gain Loss [Line Items]		
Gain (loss) on derivatives	\$ 763	\$ (189)
Commodity Hedges		
Derivative Instruments Gain Loss [Line Items]		
Gain (loss) on derivatives	13	
Other comprehensive income (loss) Foreign Currency Derivative	<u>es</u>	
Derivative Instruments Gain Loss [Line Items]		
Gain (loss) on derivatives	610	(667)
Other comprehensive income (loss) Commodity Hedges		
Derivative Instruments Gain Loss [Line Items]		
Gain (loss) on derivatives	(6)	
Cost of sales – income Foreign Currency Derivatives		
Derivative Instruments Gain Loss [Line Items]		
Gain (loss) on derivatives	153	\$ 478

\$ 19

Cost of sales – income | Commodity Hedges

Gain (loss) on derivatives

Derivative Instruments Gain Loss [Line Items]

Fair Value Measurements -Additional Information (Detail) - USD (\$)

Mar. 31, 2022 Dec. 31, 2021

Fair Value Balance Sheet Grouping Financial Statement Captions [Line Items	1	
Carrying value	\$ 38,750,000	\$ 38,750,000
DEG Vietnam Loan		
Fair Value Balance Sheet Grouping Financial Statement Captions [Line Items	1	
Carrying value	3,750,000	3,750,000
Fair value	3,796,000	3,778,000
Fair Value, Recurring Basis		
Fair Value Balance Sheet Grouping Financial Statement Captions [Line Items	1	
Financial assets, fair value	0	0
Financial liabilities, fair value	0	0
Fair Value, Nonrecurring Basis		
Fair Value Balance Sheet Grouping Financial Statement Captions [Line Items	1	
Financial assets, fair value	0	0
Financial liabilities, fair value	\$ 0	\$ 0

Equity - Additional Information (Detail) - USD

3 Months Ended

12 Months Ended

(\$)

Mar. 31, 2022 Mar. 31, 2021 Dec. 31, 2020

\$ in Thousands

Class Of Stock [Line Items]

Shares repurchased 0 0

Remaining authorized repurchase amount \$130,000

2020 Stock Repurchase Program

Class Of Stock [Line Items]

Stock repurchase program period 3 years

Stock repurchase program expiration date

Dec. 15, 2023

Maximum | 2020 Stock Repurchase Program

Class Of Stock [Line Items]

Stock repurchase program, authorized to repurchase amount

150,000

Schedule of Reclassification Adjustments and Other 3 Months Ended

Mar. 31, 2022 Mar. 31, 2021

Activities Impacting Accumulated Other

Comprehensive Loss (Detail)

- USD (\$)

\$ in Thousands		
Accumulated Other Comprehensive Income Loss [Line Items]		
Beginning Balance	\$ (36,922)	\$ (14,982)
Other comprehensive (loss) income before reclassifications	(8,378)	(13,061)
Income tax effect of other comprehensive loss before reclassifications	(322)	(239)
Amounts reclassified from accumulated other comprehensive loss into net income	(137)	(478)
Income taxes reclassified into net income	31	104
Other comprehensive loss, net of tax	(8,806)	(13,674)
Ending Balance	(45,728)	(28,656)
<u>Defined Benefit Pension Plans</u>		
Accumulated Other Comprehensive Income Loss [Line Items]		
Beginning Balance	(2,893)	(3,451)
Amounts reclassified from accumulated other comprehensive loss into net income	<u>2</u> 35	
Income taxes reclassified into net income	(6)	
Other comprehensive loss, net of tax	29	
Ending Balance	(2,864)	(3,451)
Foreign Currency Translation Adjustments		
Accumulated Other Comprehensive Income Loss [Line Items]		
Beginning Balance	(34,188)	(12,637)
Other comprehensive (loss) income before reclassifications	(9,154)	(12,872)
Income tax effect of other comprehensive loss before reclassifications	(140)	(280)
Other comprehensive loss, net of tax	(9,294)	(13,152)
Ending Balance	(43,482)	(25,789)
Foreign Currency Hedge Derivatives		
Accumulated Other Comprehensive Income Loss [Line Items]		
Beginning Balance	154	1,106
Other comprehensive (loss) income before reclassifications	763	(189)
Income tax effect of other comprehensive loss before reclassifications	(179)	41
Amounts reclassified from accumulated other comprehensive loss into net income	2(153)	[1] (478)
Income taxes reclassified into net income	33	104
Other comprehensive loss, net of tax	464	(522)
Ending Balance	618	\$ 584
Commodity Hedge Derivatives		
Accumulated Other Comprehensive Income Loss [Line Items]		
Beginning Balance	5	
Other comprehensive (loss) income before reclassifications	13	
Income tax effect of other comprehensive loss before reclassifications	(3)	
Amounts reclassified from accumulated other comprehensive loss into net income	2(19)	

Income taxes reclassified into net income	4
Other comprehensive loss, net of tax	\$ (5)

[1] The amounts reclassified from accumulated other comprehensive loss were included in cost of sales.

Income Taxes - Summary of Provision for Income Taxes and Corresponding Effective Tax Rate (Detail) - USD (\$) \$ in Thousands 3 Months Ended

Mar. 31, 2022 Mar. 31, 2021

Income Tax Disclosure [Abstract]

Income tax expense	\$ 4,295	\$ 7,565
Earnings before income tax	\$ 16,042	\$ 40,474
Effective tax rate	26.80%	18.70%

Income Taxes - Additional 3 Months Ended

Information (Detail) - USD

(\$) Mar. 31, 2022 Mar. 31, 2021

\$ in Thousands

Income Tax Disclosure [Abstract]

<u>Income tax expense</u>	\$ 4,295	\$ 7,565
Earnings before income tax	\$ 16,042	\$ 40,474
Effective tax rate	26.80%	18.70%
U.S. federal statutory rate	21.00%	21.00%

3	Mor	the	Ended	
J	TATOL	ıuıs	Liiutu	

Segment Reporting -Segment Information about Reported Product Revenues,

Depreciation and

Amortization and Operating Mar. 31, 2022 Mar. 31, 2021

Income (Loss) (Detail) - USD

(\$)

\$ in Thousands

Segment Reporting Information [Line Items]

<u>Product revenues</u> \$ 267,657 \$ 288,535

<u>Type of Revenue [Extensible List]</u> us-gaap:ProductMember us-gaap:ProductMember

Depreciation and amortization \$ 9,577 \$ 9,854 Operating income (loss) 14,190 40,749

Automotive Segments

Segment Reporting Information [Line Items]

<u>Product revenues</u> \$ 257,864 \$ 279,370

Type of Revenue [Extensible List] us-gaap:ProductMember us-gaap:ProductMember

Depreciation and amortization \$8,667 \$9,053 Operating income (loss) 31,275 53,116

Medical Segments

Segment Reporting Information [Line Items]

Product revenues \$ 9,793 \$ 9,165

Type of Revenue [Extensible List] us-gaap:ProductMember us-gaap:ProductMember

Depreciation and amortization \$ 604 \$ 584
Operating income (loss) \$ (851) \$ (531)

Corporate Segments

Segment Reporting Information [Line Items]

Type of Revenue [Extensible List] us-gaap:ProductMember us-gaap:ProductMember

Depreciation and amortization \$ 306 \$ 217 Operating income (loss) \$ (16,234) \$ (11,836)

Segment Reporting - Segment Information About	3 Mont	ths Ended
Reported Segment Product Revenues by Product Category (Detail) - USD (\$) \$ in Thousands	Mar. 31, 202	2 Mar. 31, 2021
Segment Reporting Information [Line Items]		
Product revenues	\$ 267,657	\$ 288,535
Automotive Segments		
Segment Reporting Information [Line Items]		
Product revenues	257,864	279,370
Automotive Segments Climate Control Seat		
Segment Reporting Information [Line Items]		
Product revenues	102,734	109,173
Automotive Segments Seat Heaters		
Segment Reporting Information [Line Items]		
Product revenues	68,896	76,721
Automotive Segments Steering Wheel Heaters		
Segment Reporting Information [Line Items]		
Product revenues	28,736	28,864
Automotive Segments Automotive Cables		
Segment Reporting Information [Line Items]		
Product revenues	22,045	24,281
Automotive Segments Battery Performance Solution	<u>1S</u>	
Segment Reporting Information [Line Items]		
Product revenues	17,613	17,760
Automotive Segments Electronics		
Segment Reporting Information [Line Items]		
Product revenues	10,828	15,105
Automotive Segments Other Automotive		
Segment Reporting Information [Line Items]		
Product revenues	7,012	7,466
Industrial Segments Medical Segments		
Segment Reporting Information [Line Items]		
<u>Product revenues</u>	\$ 9,793	\$ 9,165

Segment Reporting (Detail) -3 Months Ended **USD (\$)** Mar. 31, 2022 Mar. 31, 2021 \$ in Thousands **Segment Reporting Information [Line Items]** \$ 267,657 \$ 288,535 Product revenues Type of Revenue [Extensible List] us-gaap:ProductMember us-gaap:ProductMember **United States Segment Reporting Information [Line Items]** Product revenues \$ 104,122 \$ 115,287 China **Segment Reporting Information [Line Items]** 33,515 Product revenues 38,353 South Korea **Segment Reporting Information [Line Items]** Product revenues 24,640 21,175 Germany **Segment Reporting Information [Line Items]** Product revenues 19,786 19,180 Romania **Segment Reporting Information [Line Items]** Product revenues 12,755 15,085 Japan **Segment Reporting Information [Line Items]** 17,920 Product revenues 11,816 Other **Segment Reporting Information [Line Items]** Product revenues 59,650 62,908

Non U.S.

Product revenues

Segment Reporting Information [Line Items]

\$ 163,535

\$ 173,248

Subsequent Events -Additional Information (Detail) - Subsequent Events - Alfmeier

May 04, 2022 EUR (€)

€ in Thousands

Subsequent Event [Line Items]

Acquisition of shares, net of cash and debt € 177,500 Working capital adjustment € 4,000 __

Ē	
Ē	
¥	
Ē.	
Tentings by the second	
	nome in a particul desire passes a desire assesso).
The second secon	
Ī	
Esca	
Control of September 1 Control of September 2 Control of September	
E.	
	er el committation describente parties (année el mandre el confidenciamente est année describente (parties describente parties (année) ; l'année describentes (année el mandre el mandre el confidenciamente el confidenciamente (année el mandre el
The state of the s	
i	
E	(Marie America
	Marie Marie Marie Marie (
Times	
<u> </u>	
E	enc.)
	mar ^a
====	
The second secon	
Total Control of the	
Services of the control of the contr	
The second secon	
The second secon	
STATE OF THE STATE	
STATE OF THE STATE	

_

```
According to the content of the cont
```

```
Control (Control (Con
```

```
Again to a ...

The control of the c
               ELECTRONICATION OF THE PROPERTY OF THE PROPERT
```

__

_

__

__

James III.

Marie Marie

Employee

The second second

March State Common Comm

The second secon

The control of the co

Marie III.

Control Contro

Service Control of the Control of th

The second secon

March 10) March 100 March

The second secon

English To.